# A Dynamical System Approach to the Inverse Spectral Problem for Hankel Operators

by

Zhehui Liang B.Sc., Peking University, Beijing, China, 2016 M.Sc., Brown University, Providence RI, USA, 2022

A dissertation submitted in partial fulfillment of the requirements for the degree of Doctor of Philosophy in the Department of Mathematics at Brown University

> Providence, Rhode Island May 2022

@ Copyright 2022 by Zhehui Liang

This dissertation by Zhehui Liang is accepted in its present form by the Department of Mathematics as satisfying the dissertation requirement for the degree of Doctor of Philosophy

Date \_\_\_\_\_

Professor Sergei Treil, advisor

Recommended to the Graduate Council

Date \_\_\_\_\_

Professor Jill Pipher, Reader

Date \_\_\_\_\_

Professor Benoit Pausader, Reader

Approved by the Graduate Council

Date \_\_\_\_\_

Andrew G.Campbell, Dean of the Graduate School

## Curriculum Vitae

## Education

• Ph.D. in mathematics, Brown University, Providence, RI

• BSc. in Mathematics, Peking University, Beijing, China

September 2012 - May 2016

• BSc. in Economics, Peking University, Beijing, China

September 2013 - May 2016

### **Teaching Experience**

•	Teaching assistant for Math 0180 Intermediate Calculus	Spring 2021
•	Teaching assistant for Math 0520 Linear Algebra	Fall 2020
•	Teaching assistant for Math 1140 Function of Several Variables	Spring 2020
•	Instructor for Math 0190 Advanced Placement Calculus	Fall 2019
•	Teaching assistant for Math 0100 Introductory Calculus II	Spring 2019
•	Teaching assistant for Math 0100 Introductory Calculus II	Fall 2018
•	Teaching assistant for Math 0180 Intermediate Calculus	Spring 2018
•	Teaching assistant for Math 0200 Intermediate Calculus (Physics/Eng	gineeering)

Fall 2017

### **Publications and Preprints**

- Zhehui Liang and S. Treil, A Dynamical System Approach to the Inverse Spectral Problem for Hankel Operators: the Case of Compact Operators with Simple Singular Values, preprints
- Zhehui Liang and S. Treil, An inverse Spectral Problem for Hankel Operators: the Case of Compact Operators with Non-simple singular values, preprints

## Acknowledgement

I want to express greatest gratitude to the support of people around me. It's impossible to mention all of them, but I would particularly like to thank some of them for their great help and kindness.

First of all, I would like to thank my advisor Sergei Treil. He has certainly played a great role in shaping my view of mathematics. He always manages to find time discussing problems with me, providing inspiring ideas and thoughts for me to explore. I am also grateful for the opportunities of bringing me to some mathematical conferences, meeting great mathematician and sharing ideas together. In addition, I want to express my great thanks for his great support and understanding during the unexpected covid pandemic, especially in the last two years of my Ph.D. education. Thanks for bringing me to the beautiful world of functional analysis.

Thanks to my dissertation committee member: Professor Jill Pipher, Professor Benoit Pausader, for taking time to read this long paper and their valuable advice.

Thanks to all professors and my fellow students at the department that accompanies me in the analysis world. To all great professors in the analysis seminar: Professor Jungang Li, Professor Jose Conde Alonso, and all other great mathematicians. To the lecturer of my real analysis, functional analysis, and complex number theory courses: Professor Paul Dupuis, Professor Hongjie Dong, and Professor Jeremy Kahn. To my fellow graduate students that sharing ideas with me: Linhan Li, Kakaroumpas Spyridom.

Thanks to all teachers that led me to the mathematical world before my graduate studies. To my mathematical Olympics coaches in middle school and high school: Mr. QuanZhang Chen and Mr. Jianguo Li. And also to all professors that have taught me during my bachelor studies.

Thanks to all of my friends that I have met in Brown that made my life here so colorful and fantastic. To my academic siblings: Milen Ivanov, Alex McDonough, Minsik Han, Tangli Ge. Thanks for accompanying me along the first year of Ph.D. journey, and I will always remember the time we discussing homework problems and sharing funny stories together. To all of my other office mates: Feng Li, Steven Obinna, Ming Hao Quek, Sam Freedman. Thanks for your companionship. And also for all other friends I have met here: Yi Dong, Yuliang Guo, Zhijun Liu, Yuliang Chen, Sicheng Liu, Qian Zhang, Lu Lu, Tingwei Meng, Siya Zhu, Qingyu Ye, Hanjun Yang, Zhenghao Rao, Di Jin, Honghong Lin, Zhiyan Wang, Junlin Hua, Fanfei Wang, Tong Cai, Shiyu Bei, etc. I will always remember the time we spend together.

Also great thanks to all friends that I have met before joining Brown. To my roommate in Peking University: Bing Yu, Nan Xu, Zichong Wang (hope that I will not miss your wedding). To all other friends I have met in high school and my bachelor journey: Yufan Yang, Wei Hu, Zhengang Wang, Jie An, Long chen, Weibiao Wang, Xiangyu Huang, Wen Luo, Jingfeng Wu, Hao Wu, Yuchen Zhou, Lu Qi.

Thanks for my wonderful department staff: Doreen Pappas, Audrey Aguiar, Lori Nascimento, Larry Larivee, Carol Oliveira. You took care of all tough things I have encountered and offered great help to me whenever I was in trouble.

Thanks for my senior lecturer Professor Dan Katz. Thanks for your guidance and advice on a freshman lecturer.

Finally I will not forget to thank my parents: Ling Zhang and Jixian Liang. 张玲, 你是世界上最美的妈妈!梁记先,你是世界上最好的爸爸!希望我能成为一个让你们引以为傲的儿子。

Abstract of "An Inverse Spectral Problem for Hankel Operators" by Zhehui Liang, Ph.D., Brown University, May 2021

This dissertation is devoted to the study of inverse spectral problem of Hankel operators. It is well-known that spectral characteristics of a Hankel operator does not uniquely define it: there are many unitarily equivalent Hankel operators. However, as it was noticed in breakthrough papers by P. Gerald and S. Grellier the spectral characteristics of the Hankel operator and its one column truncation completely determine the compact Hankel operator. This turns out to be true for general Hankel operators, which is one of the results of the thesis.

For self-adjoint Hankel operators it is pretty easy to understand what the spectral characteristics are. For the general case the approach is more involved and based on the theory of complex symmetric operators. In both cases we state and prove an abstract theorem that reduces the existence of a Hankel operator with prescribed spectral properties to the asymptotic stability of some contractions, hence the "dynamical system approach" in the title.

We then apply this abstract theorem to the particular case of compact operators; the asymptotic stability there is obtained almost for free, and the description of spectral characteristics can be greatly simplified. We are able to give new proofs for known results by Gerard–Grillier, as well as to prove some new results.

For compact Hankel operator with simple singular values, we will show that it can be uniquely characterized by two sequences of complex numbers whose modulus part satisfy an intertwining relation. While for Hankel operator with non-simple singular values, we will show that it will be uniquely characterized by two sequences of complex numbers with the same requirement, together with two sequences of discrete probability measures on the unit circle.

## CONTENTS

#### **Notation** 0 1 1 **Introduction to Hankel Operators** 3 1.1 Introduction to Hankel Operators 3 Historical Development on the Spectral Problem of Hankel Operators . 71.21.2.1 7Modulus of Hankel Operators 1.2.2Self-adjoint Hankel Operators 8 1.2.3Hankel Operators not Self-adjoint 11 1.2.4The Spectral Data that Uniquely Determines a Hankel Operator 122 **Structure of This Thesis** 16A Result for Self-adjoint Hankel Operators 3 $\mathbf{20}$ Setup of the Inverse Spectral Problem for Self-adjoint Hankel Operators 3.121253.2Proof of Proposition 3.1.3 3.2.1253.2.2 273.2.3Trivial Kernel Condition 29Introduction to Conjugation and C- symmetric Operators 4 31Definition of Conjugation and Complex Symmetric Operators 324.1. . . . . 4.2Generalization of Polar Decomposition for $\mathfrak{C}$ -symmetric operators $\ldots$ 34

	4.3	A Property Related to Rank-one Perturbation	35			
5	Ger	neral Hankel Operators as Complex Symmetric Operators	38			
	5.1	Setup	41			
	5.2	Plan of the Game and Main Result	45			
	5.3	Proof of Proposition 5.2.3	49			
		5.3.1 Existence of Hankel Operator $\Gamma$	49			
		5.3.2 Uniqueness of Hankel Operator $\Gamma$	53			
		5.3.3 The Trivial Kernel Condition of Hankel $\Gamma$	53			
6	Eig	enspace Structure of Compact Rank-one Perturbation	56			
	6.1	Setting and preparation	57			
	6.2	Non-simple Eigenvalues for Operators of Rank-one Perturbation	59			
		6.2.1 Eigenspace Structure on $\mathcal{H}_0$	60			
		6.2.2 Eigenspace Structure on $\mathcal{H}$	64			
	6.3	Canonical Choice for $\varphi$ and $\varphi_1 \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	68			
7	Asymptotic Stability of Contraction					
	7.1	Preparation	73			
	7.2	Case when $p$ is Cyclic	75			
	7.3	Case when $p$ is not Cyclic $\ldots$	78			
	7.4	C.n.u Implies Weakly Asymptotic Stability	80			
	7.5	Main Result	84			
	7.6	Asymptotic Stability under the Canonical Choice	84			
8	Abs	stract Borg's Theorem	91			
	8.1	Proof of Abstract Borg's Theorem: Existence and Uniqueness Part 9				
		8.1.1 Some Preparation Work: the Definition of Scalar Spectral Mea-				
		sure and Cauchy Transform	94			
		8.1.2 Guess and Proof for Function $F$	96			

		8.1.3	Coefficients of the Scalar Spectral Measure $\rho(s)$	100	
		8.1.4	Existence and uniqueness of the triple $(W, W_1, p)$	102	
	8.2	Proof of Abstract Borg's Theorem: Trivial Kernel Condition			
		8.2.1	The Equivalence of Two Identities in Abstract Borg's theorem $% \mathcal{A}^{(n)}$ .	104	
		8.2.2	Trivial Kernel Condition for $W_1$ and the Spectral Measure $\rho_1(s)$	109	
	8.3	3 Abstract Borg's Theorem: Finite Rank Case			
	8.4	Proof	of Generalized Abstract Borg's Theorem	118	
9	Res	ult for	Compact Hankel Operators with Simple Eigenvalues	121	
	9.1	Result	for Self-adjoint Compact Hankel Operators with Simple Eigenvalue	s122	
	9.2	9.2 Result for Compact Hankel Operators with Simple Singular Values as			
	Complex Symmetric Operators			124	
	9.3	Result	for Finite Rank Hankel Operators with Simple Singular Values .	129	
	9.4	Conclu	nsion	131	
10 Result for Compact Hankel Operators with Non-simple Eigenvalues 1					
	10.1	Prepar	cation Work and Main Result	133	
		10.1.1	The setting of Tuple from Previous Chapters	133	
		10.1.2	Behavior of $\varphi, \varphi_1$ on the Eigenspaces $\ldots \ldots \ldots \ldots \ldots$	135	
	10.2	0.2 Proof of Theorem 10.1.5			
		10.2.1	Proof of Existence	141	
		10.2.2	Proof of Uniqueness	146	
	10.3	Result	for Finite Rank Hankel Operators with Multiple Singular Values	149	

#### • Disclaimer

This thesis is highly based on the two submitted preprints on arXiv, and the author of this thesis is one of the coauthors of these two papers.

 (i) S. R. Treil, Z. Liang, A dynamical system approach to the inverse spectral problem for Hankel operators: the case of compact operators with simple singular values, preprints  (ii) S. R. Treil, Z. Liang, An inverse spectral problem for Hankel operators: the case of compact operators with non-simple singular values, preprints

## Chapter 0

## NOTATION

This short section contains the list of main symbols. We consider only separable Hilbert spaces, usually denoted by  $\mathcal{H}$ .

:=	Equal	by	definition;
----	-------	----	-------------

 $\ell^2$  The Hilbert space formed by infinite complex sequence  $x = (x_1, x_2, ...)$  with constraint  $\sum_{i=1}^{\infty} |x_i|^2 < \infty$ ;

S Symbol for shift operator on spaces  $\ell^2$  or  $H^2(\mathbb{T})$ ;

- $\mathbb{T} \qquad \text{The unit circle on the complex plane, } \mathbb{T} := \{ z \in \mathbb{C} : |z| = 1 \};$
- $\mathbb{D} \qquad \text{The unit disk on the complex plane, } \mathbb{D} := \{ z \in \mathbb{C} : |z| \le 1 \};$
- $H^{p}(\mathbb{T}) \qquad \text{The Hardy space on } \mathbb{T} \text{ defined as } H^{p}(\mathbb{T}) := \{f : f \in L^{p}(\mathbb{T}), f = \sum_{k=0}^{\infty} a_{k} z^{k}\}$ for  $1 \leq p < \infty$ ;
  - $A^*$  Adjoint of an operator A;
  - |A| Modulus of an operator A,  $|A| := (A^*A)^{1/2}$ ;
- Ker A The kernel space of an operator A;
- Ran A The range space of an operator A;

- $P_{\mathcal{H}}x$  The orthogonal projection of a vector x onto a subspace  $\mathcal{H}$ .
  - $x^*$  For  $x \in \mathcal{H}$ ,  $x^*$  is the bounded linear functional on  $\mathcal{H}$  defined as  $y \mapsto \langle x, y \rangle$ .
  - $\mathcal{R}$  Symbol for a self-adjoint operator;
  - R Symbol for a positive self-adjoint operator;
  - $\mathcal{T}$  Symbol for a contraction;
  - $\widetilde{\Gamma}$  The restriction of an operator  $\Gamma$  on its essential part, that is  $\Gamma|_{(\text{Ker }\Gamma)^{\perp}}$ .
- $\mathfrak{C}, \mathfrak{J}$  Symbols for conjugations;
- $\mathcal{H}_0^{\perp}$  The orthogonal complement of  $\mathcal{H}_0$  on the whole space;
- $\sigma(A)$  The spectrum of an operator A;
- $\sigma_p(A)$  The point spectrum of an operator A;
  - $\tilde{\rho}$  The normalization of a finite measure  $\rho$ , i.e., the total measure of  $\tilde{\rho}$  equals 1;
  - $D_{\mathcal{T}}$  Defect operator of a contraction  $\mathcal{T}$ , defined as  $D_{\mathcal{T}} := (I \mathcal{T}^* \mathcal{T})^{1/2}$ ;
  - $\mathcal{D}_{\mathcal{T}}$  Defect space of a contraction  $\mathcal{T}$ , defined as  $\mathcal{D}_{\mathcal{T}} := \operatorname{Clos} \operatorname{Ran} D_{\mathcal{T}}$ ;
  - $\mathfrak{D}_{\mathcal{T}}$  Defect indice of a contraction  $\mathcal{T}$ , defined as  $\mathfrak{D}_{\mathcal{T}} := \dim \mathcal{D}_{\mathcal{T}}$ ;
  - $\rightharpoonup$  Symbol for weak convergence. We say a sequence of vector  $\{x_n\} \rightharpoonup x_0$  if and only if  $\lim_{n \to \infty} \langle x_n, y \rangle \to \langle x, y \rangle$  holds for all  $y \in \mathcal{H}$ .

## Chapter 1

## INTRODUCTION TO HANKEL OPERATORS

In this chapter, we will first give the definition and some well-known results about Hankel operators in section 1.1. To follow up, we will introduce some substantial historical progress on the inverse spectral problem of Hankel operators in section 1.2.

### 1.1 Introduction to Hankel Operators

**Definition 1.1.1.** A Hankel matrix is an infinite matrix of form  $\{\gamma_{j+k}\}_{j,k>=0}^{\infty}$ , i.e. a matrix whose entries depend only on the sum of indices.

A Hankel operator  $\Gamma$  is a bounded operator in  $\ell^2 = \ell^2(\mathbb{Z}_+)$  whose matrix in the standard basis  $\{e_n\}_{n=0}^{\infty}$  of  $\ell^2$  is Hankel, i.e.  $\langle \Gamma e_j, e_k \rangle = \gamma_{j+k}$  holds for all  $j, k \in N$ . And we call  $\{\gamma_n\}_{n=0}^{\infty}$  to be the Hankel coefficients of  $\Gamma$ .

Remark 1.1.2. Let S be the shift operator on  $\ell^2$  defined as

$$S(x_0, x_1, x_2, \ldots) = (0, x_0, x_1, x_2, \ldots),$$

and its adjoint  $S^*$  given by

$$S(x_0, x_1, x_2, \ldots) = (x_1, x_2, \ldots).$$

The fact that an operator  $\Gamma$  is Hankel is equivalent to the following equation:

$$\Gamma S = S^* \Gamma, \tag{1.1.1}$$

so the above identity (1.1.1) can also be used as the definition for Hankel operators.

The following result is a well-known theorem proved by Nehari in [2]. We first recall the definition of *bounded mean oscillation functions* (BMO for abbreviation).

**Definition 1.1.3** (Definition of BMO function). We say an integrable function f on  $\mathbb{T}$  is of bounded mean oscillation (BMO) if

$$||f|| := \sup_{I} \frac{1}{|I|} \int_{I} |f - f_{I}| dm < \infty,$$

where the supremum runs over all arcs of  $\mathbb{T}$  and  $f_I = \frac{1}{|I|} \int_I f dm$  is the mean of f over I.

**Theorem 1.1.4** (Nehari, 1957). A Hankel operator  $\Gamma : \ell^2 \to \ell^2$  with coefficients  $\{\gamma_n\}_{n=0}^{\infty}$  is bounded if and only if there exists a function  $f \in L^{\infty}(\mathbb{T})$ , such that  $\forall n \geq 0$ ,  $\hat{f}(n) = \gamma_n$ . Or equivalently, the Fourier series  $\sum_{n\geq 0} \gamma_n e^{inz}$  is BMO.

Another famous result is proved by Hartman in [4], given an equivalent condition for compact Hankel operators on  $\ell^2$ . We first recall the definition of *vanishing mean oscillation functions* (VMO for abbreviation).

**Definition 1.1.5** (Definition of VMO functions). We say an integrable function f on  $\mathbb{T}$  is of vanishing mean oscillation (VMO) if  $f \in BMO(\mathbb{T})$ , and additionally, the mean oscillation of f on I goes to 0 uniformly when the arc length of I goes to 0.

**Theorem 1.1.6** (Hartman, 1958). A Hankel operator  $\Gamma : \ell^2 \to \ell^2$  with coefficients  $\{\gamma_n\}_{n=0}^{\infty}$  is compact if and only if there exists a continuous function f on  $\mathbb{T}$ , such that  $\hat{f}(n) = \gamma_n$  for  $n \ge 0$ . Or equivalently, the Fourier series  $\sum_{n\ge 0} \gamma_n e^{inz}$  is a function belongs to  $VMO(\mathbb{T})$ .

As for finite rank Hankel operators, the following theorem is proved by L.Kronecker in [5], stated as follow:

**Theorem 1.1.7** (Kronecker 1881). A Hankel operator  $\Gamma$  with coefficients  $\{\gamma_n\}_{n=0}^{\infty}$  is finite-rank, if and only if the sum of series

$$R(z) = \sum_{i=0}^{\infty} \frac{\gamma_i}{z^i}$$

is a rational function of z.

*Remark* 1.1.8. There is also another Hardy space representation for Hankel operators, given in [1, Part B Chapter 1, p. 180], which we will just briefly introduce it here. (We only need Definition 1.1.1 in the main part of this thesis).

We first recall the definition of Hardy space  $H^p(\mathbb{T}) := \{f : f \in L^p(\mathbb{T}), f = \sum_{k=0}^{\infty} a_k z^k\}$ for  $1 \leq p < \infty$ . Here  $H^p$  can be naturally identified with the spaces of analytic functions on the unit disk  $\mathbb{D}$ . For  $p = \infty$ , we define  $H^\infty$  to be the space of all bounded analytic function on the unit disk with the supremum norm.

Now we construct a map from  $H^2$  to  $H^2_- := L^2 \ominus H^2$ . For a function  $\varphi \in L^{\infty}(\mathbb{T})$ , we define the operator  $H_{\varphi}$  as:

$$H_{\varphi}f := P_{-}(\varphi f),$$

here  $f \in H^2$  and  $P_-$  is the natural orthogonal projection from  $H^2$  to  $H^2_-$ .

For this  $H_{\varphi}$ :  $H^2 \to H_{-}^2$ , we can check that the representation matrix of  $H_{\varphi}$ , with respect to basis  $\mathcal{B}_1 = \{e^{int}\}_{n\geq 0}$  on  $H^2$ , and basis  $\mathcal{B}_2 = \{e^{int}\}_{n<0}$  for  $H_{-}^2$ , is an infinite symmetric matrix, thus  $H_{\varphi}$  has Hankel structure, and we call this  $H_{\varphi}$  as Hankel operators from  $H^2$  to  $H_{-}^2$ .

Now we can use this  $H_{\varphi}$  to define an operator from  $\ell^2$  to  $\ell^2$  which is Hankel. We

first define an involution operator  $\mathcal{J}$  given by

$$\mathcal{J}e^{ikt} = e^{-i(k+1)t}$$
 holds for all  $k \in \mathbb{Z}$ ,

(easy to check that  $\mathcal{J}H^2 = H^2_-$ ,  $\mathcal{J}H^2_- = H^2$  and  $\mathcal{J}^2 = \mathcal{I}$ ). Also recall the natural Fourier transform mapping  $\mathcal{F}$  defined as

$$f \to (\hat{f}(0), \hat{f}(1), ....),$$
 (1.1.2)

mapping  $H^2$  to  $\ell^2$ . Here  $\hat{f}(n)$  is the Fourier coefficients defined as  $\hat{f}(n) := \int_{\mathbb{T}} f e^{-inz} dz$ . Now we define an operator  $\Gamma : \ell^2 \to \ell^2$  as

$$\Gamma := \mathcal{F}\mathcal{J}H_{\varphi}\mathcal{F}^{-1}.$$

We can see that  $\Gamma$  is a Hankel operator with Hankel coefficients  $\gamma_k = \hat{\varphi}(-k-1)$ .

Remark 1.1.9. For this defined  $H_{\varphi}$ , we can see that it satisfies the following equation

$$H_{\varphi}S = P_{-}SH_{\varphi}, \tag{1.1.3}$$

here S is the shift operator on  $H^2$  defined as  $Sf := e^{it}$ . The above identity can also be used as the definition of Hankel operators from  $H^2$  to  $H^2_-$ .

Remark 1.1.10. From the definition of Hankel operators on Hardy spaces, it's easy to see that Ker  $H_{\varphi}$  is either trivial or infinite-dimensional (thus Ker  $\Gamma$  is also either trivial or infinite-dimensional for Hankel operator  $\Gamma : \ell^2 \to \ell^2$ ). In fact, from (1.1.3), we can see that Ker  $H_{\varphi}$  is S-invariant, thus by Beurling theorem (see [7, Chapter 1]), it must has the form of  $\theta H^2$  where  $\theta$  is an inner function, thus it must be trivial or infinite-dimensional.

## 1.2 Historical Development on the Spectral Problem of Hankel Operators

#### **1.2.1** Modulus of Hankel Operators

The spectral properties of Hankel operators was first studied in [6]. In this paper, Khrushchev and Peller raised the following question:

**Problem 1.2.1.** Given R to be a non-negative self-adjoint operator on  $\mathcal{H}$ , does there exist a Hankel operator  $\Gamma$  such that its modulus  $|\Gamma|$  is unitary equivalent to R?

Easy to see that R needs to satisfy the following two conditions:

- (i) dim Ker R = 0 or  $\infty$ ;
- (ii) R is non-invertible.

We have discussed the proof of condition (i) in Remark 1.1.10. As for condition (ii), it follows from the fact that if  $\Gamma$  is Hankel, then

$$\|\Gamma e_k\|^2 = \sum_{j \ge k} |\alpha_j|^2 \to 0 \quad \text{as } k \to \infty$$

So now the question is whether (i) and (ii) are also sufficient for Problem 1.2.1.

The first progress was made in [8]. It was shown that if R satisfies (i), (ii), and additionally R has simple discrete spectrum, then R is unitary equivalent to  $|\Gamma|$  for some Hankel  $\Gamma$ .

Then in 1990, Treil solved Problem 1.2.1, claiming that the discrete spectrum condition is unnecessary. That is,

**Theorem 1.2.2.** *R* is a non-negative self-adjoint operator satisfying the following assumptions:

(i) R is non-invertible;

(ii) dim Ker R = 0 or  $\infty$ .

Then there exists a Hankel operator  $\Gamma$ , such that  $|\Gamma|$  is unitary equivalent to R.

#### 1.2.2 Self-adjoint Hankel Operators

Now we consider the inverse spectral problem for Hankel operators.

**Problem 1.2.3.** For a given self-adjoint operator  $\mathcal{R}$ , can we find out a Hankel operator  $\Gamma$ , such that  $\mathcal{R}$  is unitary equivalent to  $\Gamma$ ?

To begin stating the equivalent conditions for this problem, we need some preparation on the definition of *Von Neumann integral* and *scalar multiplicity function*.

Given  $\mu$  to be a finite positive Borel measure on  $\mathbb{R}$ , and  $\{\mathcal{H}(t)\}_{t\in\mathbb{R}}$  be a measurable family of Hilbert spaces. Here the definition of measurable families is given as follow:

**Definition 1.2.4** (Definition of measurable families). Let  $\{\mathcal{H}(t)\}_{t\in\mathbb{R}}$  be a family of Hilbert spaces, then we say  $\{\mathcal{H}(t)\}_{t\in\mathbb{R}}$  is a measurable family if and only if there exists at most a countable set  $\Omega$  of functions f such that  $f(t) \in \mathcal{H}(t)$ ,  $\mu$ -a.e., such that

$$\overline{\text{Span}}\{f(t): f \in \Omega\} = \mathcal{H}(t) \qquad \text{for } \mu - almost \ all \ t,$$

and the function

$$t \to \langle f_1(t), f_2(t) \rangle_{\mathcal{H}(t)}$$

is  $\mu$ -measurable for any  $f_1, f_2 \in \Omega$ .

If we denote  $N(t) := \dim \mathcal{H}(t)$ , then we can indeed embed all  $\mathcal{H}(t)$  in a Hilbert space with orthogonal basis  $\{e_n\}_{n=1}^{\infty}$ , and set

$$\mathcal{H}(t) = \overline{\operatorname{Span}}\{e_k : 1 \le k \le N(t)\}$$

**Definition 1.2.5.** We say a function g with values  $g(t) \in \mathcal{H}(t)$  is measurable if the scalar function

$$t \to \langle g(t), f(t) \rangle_{\mathcal{H}(t)}$$

is measurable for all  $f \in \Omega$ .

Now we are ready to define the *Von-Neumann integral space.*, and then followed by the *Von-Neumann theorem*.

**Definition 1.2.6.** The Von-Neumann integral (or direct integral)  $\int \oplus \mathcal{H}(t) d\mu(t)$  is a Hilbert space consists of measurable functions f (see definition 1.2.5), such that

$$||f|| = \left(\int ||f(t)||^2_{\mathcal{H}(t)} d\mu(t)\right)^{1/2} < \infty.$$

And the inner product on the Von-Neumann integral is defined as

$$\langle f,g\rangle = \int \langle f(t),g(t)\rangle_{\mathcal{H}(t)}d\mu(t),$$

where  $f, g \in \int \oplus \mathcal{H}(t) d\mu(t)$ .

With this newly defined direct integral space, Von-Neumann proved that every selfadjoint operators is unitary equivalent to a multiplication by independent variable on some  $L^2$  space. The theorem is stated as follow:

**Theorem 1.2.7** (Von Neumann' theorem, see [11]). Each self-adjoint operator defined on a separable Hilbert space is unitary equivalent to an operator A, which is a multiplication by independent variable defined on a direct integral  $\int \oplus \mathcal{H}(t)d\mu(t)$ :

$$(Af)(t) = tf(t), \qquad f \in \int \oplus \mathcal{H}(t)d\mu(t).$$

And we say  $\mu$  to be the scalar spectral measure of A, and we say the function  $N(t) = \dim \mathcal{H}(t)$  to be the spectral multiplicity function of A.

It is also stated in [11] that two self-adjoint operators  $R_1, R_2$  are unitary equivalent if and only if their scalar spectral measures are mutually absolutely continuous and their spectral multiplicity functions are equal almost everywhere.

Now back to the Problem 1.2.3. A.V. Megretskii, V.V.Peller, and S.R.Treil solved problem 1.2.3 in [12, Theorem 1, p. 245]. The equivalent condition of such existence is stated as follow:

**Theorem 1.2.8.** Let  $\mathcal{R}$  be a bounded self-adjoint operator on Hilbert space, denote  $\mu$  be its scalar spectral measure, and N(t) be its spectral multiplicity function. Then  $\mathcal{R}$  is unitary equivalent to a Hankel operator  $\Gamma$  if and only if all the following conditions hold

- (i) dim Ker  $\mathcal{R} = 0$  or  $\infty$ ;
- (ii)  $\mathcal{R}$  is non-invertible;
- (iii) The spectral multiplicity function N(t) satisfies: |N(t) − N(−t)| ≤ 2, μ<sub>a</sub>-a.e. (absolute continuous part of μ), and |N(t) − N(−t)| ≤ 1, μ<sub>s</sub>-a.e. (singular part of μ).

Here (iii) implies that if one of the N(t), N(-t) is  $\infty$ , then the other one must also be  $\infty$ .

Remark 1.2.9. Notice that condition (iii) also implies that for any self-adjoint Hankel  $\Gamma$ , we have

$$|\dim \operatorname{Ker}(\Gamma - \lambda I) - \dim \operatorname{Ker}(\Gamma + \lambda I)| \le 1 \quad \text{holds for all } \lambda \in \mathbb{R}.$$
(1.2.1)

Furthermore, [12] also shows that the inequality (1.2.1) is true for all bounded Hankel  $\Gamma$  and all  $\lambda \in \mathbb{C}$ , i.e.,

$$\dim \operatorname{Ker}(\Gamma - \lambda I) - \dim \operatorname{Ker}(\Gamma + \lambda I) \leq 1 \tag{1.2.2}$$

holds for all  $\lambda \in \mathbb{C}$ .

From Theorem 1.2.8, we can easily generate another version when R is compact.

**Theorem 1.2.10.** Let  $\mathcal{R}$  be a bounded self-adjoint compact operator on Hilbert space, then  $\mathcal{R}$  is unitary equivalent to a Hankel operator  $\Gamma$  if and only if all the following conditions hold

- (i) dim Ker  $\mathcal{R} = 0$  or  $\infty$ ;
- (ii)  $\mathcal{R}$  is non-invertible;
- (iii)  $|\dim \operatorname{Ker}(\mathcal{R} \lambda I) \dim \operatorname{Ker}(\mathcal{R} + \lambda I)| \leq 1$  holds for all  $\lambda \in \mathbb{R}$ .

#### 1.2.3 Hankel Operators not Self-adjoint

For a general Hankel  $\Gamma$  which is not self-adjoint, the spectrum properties of  $\Gamma$  is also deeply studied. In Remark 1.2.9, we have already given several spectrum properties for a general Hankel  $\Gamma$ :

(i) 
$$0 \in \sigma(\Gamma)$$
;

(ii)  $|\dim \operatorname{Ker}(\Gamma - \lambda I) - \dim \operatorname{Ker}(\Gamma + \lambda I)| \leq 1$  holds for all  $\lambda \in \mathbb{C}$ . In particular, if  $\lambda$  is a multiple eigenvalue of a Hankel  $\Gamma$ , then  $-\lambda$  must be an eigenvalue.

There are a lot of progress concentrating on whether there exists a Hankel  $\Gamma$ , such that  $\sigma(\Gamma) = \{0\}$ . The first progress was made by S. Power in [13], proving that there exists no non-trivial nilpotent Hankel operators ( $\Gamma^n = 0$  for some n > 0). Then in 1991, A. Metretskii constructed a non-trivial quasinilpotent Hankel operator in [14], i.e., a Hankel operator  $\Gamma$  such that  $\|\Gamma^n\|^{1/n} \to 0$  as  $n \to \infty$  (or equivalently,  $\sigma(\Gamma) = \{0\}$ ).

Another substantial result is proved in [15], saying that there are no other constraints on the spectrum of  $\Gamma$  except for {0}. That is:

**Theorem 1.2.11.** Let  $\sigma$  be any compact subset on the complex plane containing zero. Then there exists a Hankel operator  $\Gamma$  such that  $\sigma(\Gamma) = \sigma$  When considering other spectral properties of Hankel operators, recall that the inequality (1.2.2) gives a restriction on the geometric multiplicities of eigenvalues. However, if we consider the algebraic multiplicities of eigenvalues (recall that the *algebraic multiplicity* of an eigenvalue  $\lambda$  is the dimension of the space of all generalized eigenvectors, i.e. the dimension of the space  $\bigcup_{n\geq 1} \operatorname{Ker}(\Gamma - \lambda I)^n$ ), E. Abakumov proved in [16] that there are no restrictions on the algebraic multiplicities of the eigenvalues, for the special case of finite rank Hankel operators.

**Theorem 1.2.12.** Given a finite number of non-zero points  $\lambda_1, \lambda_2, ..., \lambda_n$  and multiplicities  $k_1, ..., k_n$ , there exists a finite rank Hankel operator  $\Gamma$ , such that its non-zero eigenvalues are exactly  $\lambda_1, ..., \lambda_n$ , and the corresponding algebraic multiplicities are exactly  $k_1, ..., k_n$  with 0 an eigenvalue of infinite multiplicity.

#### **1.2.4** The Spectral Data that Uniquely Determines a Hankel Operator

However, for all the results we stated in previous subsections (Theorem 1.2.8, Theorem 1.2.10, Theorem 1.2.11 and Theorem 1.2.12), the constructed Hankel operator is not unique. In other words, there are many unitary invariants of a Hankel operator that share the same spectral property with a given operator R.

Thus we wonder, what type of spectral data that can uniquely determine a Hankel operator?

The first pioneering progress of this problem was made by P. Gérard and S. Grellier in [17]. Their results came up as a byproduct of the study of an integrable Hamiltonian system called the cubic Szegö equation. For every  $u \in H^2(\mathbb{T})$ , they define the Hankel operator of symbol u as:

$$H_u(h) := P_+(u\overline{h}), \qquad h \in H^2(\mathbb{T}).$$

We need to emphasize that this  $H_u$  is conjugate-linear, which is different from the standard definition given in Definition 1.1.1 and Remark 1.1.8. In fact, we can set up

a correspondence between Gérard-Grellier's definition and Definition 1.1.1.

For a given  $u \in H^2(\mathbb{T})$ , we define a Hankel operator  $\Gamma_u$  on  $\ell^2$  with related Hankel coefficients  $\{\gamma_k\}_{k=0}^{\infty}$  satisfies  $\gamma_k = \hat{u}(k)$ , here  $\hat{u}(k)$  is the Fourier coefficients  $\hat{u}(k) = \int_{\mathbb{T}} u e^{-ikz} dz$ . Then we will have the defined  $H_u$  is unitary equivalent to a conjugatelinear operator  $\mathfrak{C}\Gamma$  on  $\ell^2$ . In fact, we have  $H_u = \mathcal{F}^{-1}(\mathfrak{C}\Gamma)\mathcal{F}$ , where  $\mathcal{F}$  is the Fourier transform given in (1.1.2).

Now for the defined  $H_u$ , we can easily set up a rank-one perturbation relation. We define the shift operator on  $H^2(\mathbb{T})$  as  $S_u := e^{ix}u$ , and then we have  $S^*H_u = H_uS =$  $H_{S^*u}$ , where  $S^*$  is the adjoint of S with the representation  $S^*u = P_+(e^{-ix}u)$ . We denote this  $S^*H_u$  as a new operator  $K_u$  (which is also conjugate-linear), and we call this  $K_u$ to be the *shifted Hankle operator* with symbol u. Then we have  $H_u$  and  $K_u$  is related by the following identity

$$K_u^2 = H_u^2 - u\langle , u \rangle$$

For this  $H_u^2, K_u^2$ , we can show that they are unitary equivalent to  $|\Gamma|^2, |\Gamma_1|^2$  respectively:

$$H_u^2 = \mathcal{F}^{-1} |\Gamma|^2 \mathcal{F}, \qquad K_u^2 = \mathcal{F}^{-1} |\Gamma_1|^2 \mathcal{F}.$$

In the Gérard-Grellier's work [17], they first considered a symbol u such that  $H_u$  has rank N, and  $1 \notin \operatorname{Ran} H_u$ , then they showed that the singular values of  $H_u^2, K_u^2$ , denoted as  $\{\lambda_n\}_{n=1}^N, \{\mu_n\}_{n=1}^N$ , are simple and satisfies the following intertwining relation

$$\lambda_1^2 > \mu_1^2 > \dots > \lambda_N^2 > \mu_N^2. \tag{1.2.3}$$

And conversely, given two sequences  $\{\lambda_n\}_{n=1}^N, \{\mu_n\}_{n=1}^N$  satisfy the intertwining relation (1.2.3), we can find a unique u such that  $H_u^2, K_u^2$  have non-zero singular values as  $\{\lambda_n\}_{n=1}^N, \{\mu_n\}_{n=1}^N$  respectively. If we further assume that  $\{\lambda_n\}_{n=1}^N, \{\mu_n\}_{n=1}^N$  are real numbers, then Gérard-Grellier proved in the same paper [17] that we can find a symbol u with real Fourier coefficients  $\{\hat{u}(k)\}_{k=0}^{\infty}$ , such that the induced Hankel operator  $\Gamma_u, S^*\Gamma_u$  on  $\ell^2$  are self-adjoint and have simple eigenvalues as  $\{\lambda_n\}_{n=1}^N, \{\mu_n\}_{n=1}^N$  respectively.

Later in a followup paper [36], Gérard-Grellier' solved the case for general selfadjoint Hankel operators with simple singular values. The result is stated as follow:

**Theorem 1.2.13.** Let  $\{\lambda_n\}_{n=1}^{\infty}$ ,  $\{\mu_n\}_{n=1}^{\infty}$  be two sequences of real numbers such that

$$|\lambda_1| > |\mu_1| > \dots > |\lambda_n| > |\mu_n| > \dots \to 0,$$

then there exists a unique self-adjoint compact Hankel operator  $\Gamma$ , such that

- (i) the non-zero eigenvalues of  $\Gamma$  are simple, and coincides with  $\{\lambda_n\}_{n=1}^{\infty}$ ;
- (ii) the non-zero eigenvalues of  $S^*\Gamma$  are simple, and coincides with  $\{\mu_n\}_{n=1}^{\infty}$ .

Then in 2014, Gérard-Grellier also solved the case for compact Hankel operators with multiple singular values. In [19], they described the spectral data by the corresponding singular values together with some Blaschke products.

In this thesis, we will present a dynamical system approach for the inverse spectral problems of Hankel operators by reducing the problem to the asymptotic stability of a certain contraction. Different from the work by Gérard-Grellier, we solve the problem by means of:

- (i) Instead of using the language of conjugate-linear operators in [17], [36] and [19];
   we treat Hankel operatoars as complex symmetric operators in the non self-adjoint case;
- (ii) To reconstruct the compact Hankel operators according to their singular values, we will present a theorem so-called Abstract Borg's theorem (see Theorem 8.0.1), and then we construct the scalar spectral measure of the Hankel operator from this theorem;

(iii) For the multiple singular values case, we translate the spectral data by their singular values and some probability measures, where those probability measures are exactly the Clark measures of the Blaschke products given in [19].

## Chapter 2

## STRUCTURE OF THIS THESIS

In this chapter, we will briefly introduce the structure of this thesis.

Throughout the historical studies of inverse spectral problems of Hankel operators (see section 1.2), we can see that

- (i) A Hankel operator cannot be uniquely characterized by the spectral data of its own. There are many unitary invariants of it on ℓ<sup>2</sup> which are also Hankel, and share the same spectral properties;
- (ii) From the studies of P. Gérard and S. Grellier in [17] and [18], we can see that the spectral information of  $\Gamma$  and  $S^*\Gamma$  can completely characterize a Hankel operator.

In this thesis, we will further discuss the topic that what spectral data can uniquely determine a Hankel operator.

In chapter 3, we mainly discuss the inverse spectral problem for self-adjoint Hankel operators. A self-adjoint Hankel operator  $\Gamma$  satisfies  $\Gamma^2 - \Gamma_1^2 = uu^*$  where  $\Gamma_1 := \Gamma S = S^*\Gamma$  and  $u = \Gamma e_0$ , thus we construct a tuple  $(\mathcal{R}, \mathcal{R}_1, p)$  with relation  $\mathcal{R}^2 - \mathcal{R}_1^2 = pp^*$  where  $\mathcal{R}, \mathcal{R}_1$  are self-adjoint and Ker  $\mathcal{R} = \{0\}$ . We want to find out whether there exists a self-adjoint  $\Gamma$ , such that the tuple restricted on the essential part of  $\Gamma$ , i.e.,  $(\widetilde{\Gamma} := \Gamma|_{(\text{Ker}\,\Gamma)^{\perp}}, \widetilde{\Gamma}_1 := \Gamma_1|_{(\text{Ker}\,\Gamma)^{\perp}}, u)$  is unitary equivalent to  $(\mathcal{R}, \mathcal{R}_1, p)$ . If such  $\Gamma$  exists,

since we have

$$\widetilde{\Gamma}_1 = S^* \big|_{(\operatorname{Ker} \Gamma)^{\perp}} \widetilde{\Gamma},$$

we require the contraction  $\mathcal{T}$  defined by  $\mathcal{T} := \mathcal{R}_1 \mathcal{R}^{-1}$  to be unitary equivalent to  $S^*|_{(\operatorname{Ker}\Gamma)^{\perp}}$ , thus  $\mathcal{T}$  is asymptotically stable. In fact, we show in Proposition 3.1.3 tells us that there exists a unique Hankel  $\Gamma$  such that  $(\widetilde{\Gamma}, \widetilde{\Gamma}_1, u)$  is unitary equivalent to  $(\mathcal{R}, \mathcal{R}_1, p)$  as long as we guarantee the asymptotic stability of  $\mathcal{T}$ . Thus a self-adjoint Hankel can be uniquely determined by two operators  $\mathcal{R}, \mathcal{R}_1$ .

Then in chapter 5, we discuss the inverse spectral problem for general Hankel operators  $\Gamma$ . Under this case we take  $\Gamma$  as a special type of operators called  $\mathfrak{C}$ -symmetric operators (See Definition 4.1.6), and write the essential part  $\widetilde{\Gamma}, \widetilde{\Gamma}$  as (5.1.5):

$$\widetilde{\Gamma} = \mathfrak{C}\widetilde{\phi}|\widetilde{\Gamma}|\widetilde{\mathfrak{J}}_u, \qquad \widetilde{\Gamma}_1 = \mathfrak{C}\widetilde{\phi}_1|\widetilde{\Gamma}_1|\widetilde{\mathfrak{J}}_u,$$

where here  $\mathfrak{J}_u$  is a conjugation commutes with  $|\Gamma|, |\Gamma_1|, \tilde{\phi}$  is unitary and  $\tilde{\phi}_1$  is partial isometry which is unitary on  $\ell^2 \ominus \operatorname{Ker} \Gamma_1$ .

Since a general Hankel operator  $\Gamma$  satisfies  $|\widetilde{\Gamma}|^2 - |\widetilde{\Gamma}_1|^2 = uu^*$  where  $u := \Gamma^* e_0$ , we construct a tuple  $(R, R_1, p, \varphi, \varphi_1, \mathfrak{J}_p)$  that satisfies

- (i)  $R, R_1$  are two positive, self-adjoint compact operators defined on a Hilbert space  $\mathcal{H}$ . In addition we have Ker  $R = \{0\}$ ;
- (ii)  $R^2 R_1^2 = pp^*$  for a vector p with  $||R^{-1}p|| \le 1$ ;
- (iii)  $\mathfrak{J}_p$  is a conjugation commutes with  $R, R_1$  and preserves p;
- (iv)  $\varphi$  is a  $\mathfrak{J}_p$ -symmetric unitary operator, which commutes with R;
- (v)  $\varphi_1$  is a  $\mathfrak{J}_p$ -symmetric partial isometry with Ker  $\varphi_1 = \text{Ker } R_1$ , which commutes with  $R_1$ . In addition, we have  $\varphi_1|_{(\text{Ker } R_1)^{\perp}}$  is unitary.

We want to find out whether there exists a Hankel operator  $\Gamma$ , such that there exists a conjugation  $\mathfrak{J}_u$ , commuting with  $|\Gamma|$  and  $|\Gamma_1|$  and preserving u, such that induced tuple  $(|\widetilde{\Gamma}|, |\widetilde{\Gamma}_1|, u, \widetilde{\phi}, \widetilde{\phi}_1, \widetilde{\mathfrak{J}}_u)$  is unitary equivalent to  $(R, R_1, p, \varphi, \varphi_1, \mathfrak{J}_p)$ . In fact, we show in Proposition (5.1.6) that with the asymptotic stability of  $\mathcal{T} := R_1 \varphi_1 \varphi^* R^{-1}$ , there exists a unique Hankel operator  $\Gamma$  such that  $(|\widetilde{\Gamma}|, |\widetilde{\Gamma}_1|, u, \widetilde{\phi}, \widetilde{\phi}_1, \widetilde{\mathfrak{J}}_u)$  is unitary equivalent to  $(R, R_1, p, \varphi, \varphi_1, \mathfrak{J}_p)$ . Thus a general Hankel operator can be uniquely determined by two operators  $\mathcal{R}_1 := R_1 \varphi_1, \mathcal{R} := R\varphi$  (which also means that a Hankel operator can be uniquely determined by its essential part).

For the following chapters, we will further consider the case of compact Hankel operators, and describe the spectral data of these two operators in a different way from the results in [36].

We first need to consider the assumptions needed for  $R\varphi$ ,  $R_1\varphi_1$  to guarantee the asymptotic stability of  $\mathcal{T}$ . Finding out the equivalent requirement for the asymptotic stability of  $\mathcal{T}$  is usually a hard issue, but things become easier if we have R,  $R_1$  are compact. In chapter 7, we discuss the requirements we need to guarantee the asymptotic stability of  $\mathcal{T}$ .

- (i) If p is cyclic with respect to R in  $\mathcal{H}$ , then we will show that  $\mathcal{T}$  will be asymptotic stable for free.
- (ii) If p is not cyclic with respect to R in H, then Proposition 7.5.1 gives a criterion for the asymptotic stability of T. In addition, if we set a canonical choice for φ, φ<sub>1</sub> given in Lemma 6.3.2, then Proposition 7.6.3 gives an equivalent condition for φ and φ<sub>1</sub> for the asymptotic stability of T.

Secondly, to translates the spectral data of  $\mathcal{R}_1, \mathcal{R}$  in compact case, we first need to analyze the eigenspace structure  $R_1, R$ , which will be the work in chapter 6. Denoting  $\mathcal{H}_0 = \overline{\text{Span}} \{ R^n p | n \ge 0 \}$ , we have  $R|_{\mathcal{H}_0}, R_1|_{\mathcal{H}_0}$  both have simple eigenvalues, and  $R = R_1$  on  $\mathcal{H}_0^{\perp}$ . The complete description of structure of tuple  $(R, R_1, p)$  is stated in Proposition 6.2.10. With these preparations, now we are ready to translate the spectral data of  $\mathcal{R}_1, \mathcal{R}_1$ . We classify the compact  $\mathcal{R}, \mathcal{R}_1$  into the following situations:

(a)  $\mathcal{H}_0 = \mathcal{H}$  and  $\mathcal{R}_1, \mathcal{R}$  are self-adjoint. Under this case, we have  $\sigma(\varphi) = \sigma_p(\varphi) \subseteq \{\pm 1\}$ ,  $\sigma(\varphi_1) = \sigma_p(\varphi_1) \subseteq \{\pm 1, 0\}$  with Ker  $\varphi_1 = \text{Ker } R_1$ , thus we can write  $\varphi, \varphi_1$  as  $f(R), f_1(R_1)$  where  $f, f_1$  are some measurable unimodular functions that take values in  $\{\pm 1\}$ . With these assumptions, we show in Theorem 9.1.1 that a self-adjoint compact Hankel operator with simple singular values can be uniquely characterized by two sets of real sequences  $\{\lambda_n\}_{n=1}^{\infty}, \{\mu_n\}_{n=1}^{\infty}$  satisfying the following intertwining relation:

$$|\lambda_1| > |\mu_1| > |\lambda_2| > |\mu_2| > \dots > |\lambda_n| > |\mu_n| > \dots \to 0.$$

(b)  $\mathcal{H}_0 = \mathcal{H}$  and  $\mathcal{R}_1, \mathcal{R}$  are not self-adjoint. Under this case,  $\varphi, \varphi_1$  acts unitarily on each one-dimensional eigenspace of  $R, R_1$  respectively. Under this condition, we show in Theorem 9.2.1 that a compact Hankel operator with simple singular values can be uniquely characterized by two sets of complex sequences  $\{\lambda_n\}_{n=1}^{\infty}, \{\mu_n\}_{n=1}^{\infty}$ , whose modulus part satisfy an intertwining relation:

$$|\lambda_1| > |\mu_1| > |\lambda_2| > |\mu_2| > \dots > |\lambda_n| > |\mu_n| > \dots \to 0$$

(c)  $\mathcal{H}_0 \subsetneq \mathcal{H}$ . Under this case, if we consider the canonical choice of  $\varphi, \varphi_1$ , then Lemma 6.3.2 and Proposition 7.6.3 give the behavior of  $\varphi, \varphi_1$  on each eigenspaces of  $R, R_1$ . Denoting  $\{\rho_k\}_{k=1}^{\infty}, \{\rho_k^1\}_{k=1}^{\infty}$  as the scalar spectral measure of  $\varphi, \varphi_1$  restricted to the corresponding eigenspaces with respect to their \*-cyclic vector, then we show in Proposition 10.1.5 that a compact Hankel operator  $\Gamma$  can be uniquely determined by two complex sequences  $\{\lambda_n\}_{n=1}^{\infty}, \{\mu_n\}_{n=1}^{\infty}$ , which serves as the singular values of  $|\Gamma|$  and  $|\Gamma_1|$ , together with two normalized discrete probability measure  $\{\widetilde{\rho}_k\}_{k=1}^{\infty}, \{\widetilde{\rho}_k^1\}_{k=1}^{\infty}$ .

Hence the four sequences  $\{\lambda_n\}_{n=1}^{\infty}, \{\mu_n\}_{n=1}^{\infty}, \{\widetilde{\rho}_k\}_{k=1}^{\infty}, \{\widetilde{\rho}_k\}_{k=1}^{\infty}$  as a whole can be regarded as the spectral data of a general compact Hankel operator.

## Chapter 3

## A RESULT FOR SELF-ADJOINT HANKEL OP-ERATORS

In this chapter, we discuss the inverse spectral problem for self-adjoint Hankel operators.

First in section 3.1, we introduce the setting and some preparation work for the work. For a self-adjoint Hankel operator  $\Gamma$ , we can show that  $\Gamma$  and  $\Gamma_1 := \Gamma S$  satisfy a rank-one perturbation relation (3.1.1), thus restricted on  $(\text{Ker }\Gamma)^{\perp}$  we get a triple  $(\widetilde{\Gamma} := \Gamma|_{(\text{Ker }\Gamma)^{\perp}}, \widetilde{\Gamma}_1 := \Gamma_1|_{(\text{Ker }\Gamma)^{\perp}}, u := \Gamma e_0).$ 

Next in section 3.2, we prove the main result (Proposition 3.1.3) in this chapter. That is, for a given rank-one perturbation  $(\mathcal{R}, \mathcal{R}_1, p)$  satisfying the following conditions:

- (i)  $\mathcal{R}, \mathcal{R}_1$  are self-adjoint operators, Ker  $\mathcal{R} = \{0\}$ ;
- (ii)  $\mathcal{R}^2 \mathcal{R}_1^2 = pp^*;$
- (iii) The contraction  $\mathcal{T} := \mathcal{R}_1 \mathcal{R}^{-1}$  is asymptotically stable,

then there exists a unique self-adjoint Hankel operator  $\Gamma$ , such that the triple restricted on the essential part of  $\Gamma$ :  $(\tilde{\Gamma}, \tilde{\Gamma}_1, u)$  is unitary equivalent to  $(\mathcal{R}, \mathcal{R}_1, p)$ . Thus from this proposition, we can see that a self-adjoint Hankel operator can be uniquely determined by the spectral data of two operators  $\mathcal{R}, \mathcal{R}_1$  which satisfy a rank-one perturbation relation. In addition, we can further translate the spectral data of  $\mathcal{R}, \mathcal{R}_1$  under the selfadjoint compact case. If we further assume that p is cyclic with respect to  $|\mathcal{R}|$  in  $\mathcal{H}$  (under this case we can see that  $\mathcal{R}$  has simple singular values), then we can show that the spectral data of  $\mathcal{R}, \mathcal{R}_1$  can be characterized by two sets of real sequences  $\{\lambda_n\}_{n=1}^{\infty}, \{\mu_n\}_{n=1}^{\infty}$  satisfying an intertwining relation

$$|\lambda_1| > |\mu_1| > |\lambda_2| > |\mu_2| > \dots > |\lambda_n| > |\mu_n| > \dots \to 0.$$
(3.0.1)

In other words, given two real sequences  $\{\lambda_n\}_{n=1}^{\infty}, \{\mu_n\}_{n=1}^{\infty}$  with the intertwining relation (3.0.1), there exists a unique self-adjoint compact Hankel operator  $\Gamma$  such that the non-zero eigenvalues of  $\Gamma, \Gamma_1$  are simple, and coincide with  $\{\lambda_n\}_{n=1}^{\infty}, \{\mu_n\}_{n=1}^{\infty}$  respectively. The detail of work in this paragraph can be found in chapter 8.

## 3.1 Setup of the Inverse Spectral Problem for Self-adjoint Hankel Operators

Let  $\Gamma$  be a self-adjoint Hankel operator. Define  $\Gamma_1 := \Gamma S = S^* \Gamma$ , which is also a self-adjoint Hankel operator, then we can write

$$\Gamma_1^2 = \Gamma S S^* \Gamma = \Gamma (I - e_0 e_0^*) \Gamma = \Gamma^2 - u u^*, \qquad (3.1.1)$$

where  $u := \Gamma e_0$ .

Since  $\Gamma$  is self-adjoint, we have  $\operatorname{Ran} \Gamma \oplus \operatorname{Ker} \Gamma = \ell^2$ , and  $(\operatorname{Ker} \Gamma)^{\perp}$  is a reducing subspace for both  $\Gamma$  and  $\Gamma_1$ . Hence we restrict everything on  $(\operatorname{Ker} \Gamma)^{\perp}$ , and denote

$$\widetilde{\Gamma} := \Gamma \big|_{(\operatorname{Ker} \Gamma)^{\perp}}, \qquad \widetilde{\Gamma}_1 := \Gamma_1 \big|_{(\operatorname{Ker} \Gamma)^{\perp}},$$

then (3.1.1) can be transformed to

$$\widetilde{\Gamma}^2 - \widetilde{\Gamma}_1^2 = uu^* \tag{3.1.2}$$

Now let us try to go from the opposite direction. Suppose that we are given two selfadjoint operators  $\mathcal{R}$ ,  $\mathcal{R}_1$  on a Hilbert space  $\mathcal{H}$ , and  $R, R_1$  are their modulus operator respectively

$$\mathcal{R} = R\varphi, \qquad \mathcal{R}_1 = R_1\varphi_1,$$

where here  $\varphi, \varphi_1$  are partial isometries commuting with  $R, R_1$  respectively.

Our goal here to is find a Hankel operator  $\Gamma$ , such that the essential part  $\widetilde{\Gamma}$  and  $\widetilde{\Gamma}_1$ are simultaneously unitary equivalent to  $\mathcal{R}$  and  $\mathcal{R}_1$  respectively, meaning that we have  $\widetilde{\Gamma} = \mathcal{VRV}^*$ ,  $\widetilde{\Gamma}_1 = \mathcal{VR}_1 \mathcal{V}^*$  for some unitary operator  $\mathcal{V}$ . We can see from (3.1.1) that  $\mathcal{R}$ and  $\mathcal{R}_1$  should satisfy the relation

$$\mathcal{R}^2 - \mathcal{R}_1^2 = pp^* \tag{3.1.3}$$

for a vector  $p \in \operatorname{Ran} \mathcal{R}$ .

To begin with solving the inverse spectral problem, we first need to find a contraction  $\mathcal{T}$ , (hopefully unitary equivalent to the backward shift  $S^*$ ) such that  $\mathcal{R}_1 = \mathcal{T}\mathcal{R}$ . The tool to find  $\mathcal{T}$  is by using the following simple lemma.

**Lemma 3.1.1** (Douglas Lemma). Let A and B be two bounded operators on a Hilbert space  $\mathcal{H}$  such that

$$\|Bh\| \le \|Ah\| \qquad \forall h \in \mathcal{H},$$

or, equivalently,  $B^*B \leq A^*A$ .

Then there exists a contraction T (i.e.  $||T|| \leq 1$ ) such that B = TA. In addition, if Ker  $A = \text{Ker } A^* = \{0\}$ , then the operator  $\widetilde{T} := BA^{-1}$  (initially defined on Clos Ran A) can be extended to a contraction T on the whole space  $\mathcal{H}$ , and its adjoint is given by  $T^* = (A^*)^{-1}B^*$ ; notice that the condition  $||T|| \leq 1$  implies that Ran  $B^* \in \text{Dom}(A^*)^{-1}$ , so the operator  $T^*$  is well defined on all  $x \in \mathcal{H}$ 

For the specific condition, if A has dense range, the operator T is unique.

The specific proof of Lemma 3.1.1 can be found in [20].

When finding a contraction  $\mathcal{T}$  satisfying  $\mathcal{R}_1 = \mathcal{T}\mathcal{R}$ , we usually assume Ker  $\mathcal{R} = \{0\}$  to avoid the non-uniqueness of such  $\mathcal{T}$ , meaning that we want R to be unitarily equivalent to the *essential part of*  $\Gamma$  (instead of  $\Gamma$  itself), i.e., to the operator  $\widetilde{\Gamma} := \Gamma|_{(\text{Ker }\Gamma)^{\perp}}$ . In this case the operator  $\mathcal{T}$  should be unitarily equivalent to  $S^*$  restricted on  $(\text{Ker }\Gamma)^{\perp}$ , meaning that

$$\mathcal{T} = \widetilde{V}^* S^*|_{(\text{Ker } \Gamma)^{\perp}} \widetilde{V}$$
(3.1.4)

for an unitary  $\widetilde{V}$ .

In this situation the vector p from (3.1.3) should be a vector satisfying  $||\mathcal{R}^{-1}p|| \leq 1$ , where here  $q := \mathcal{R}^{-1}p$  can be thinking as

$$q = \widetilde{V}^* P_{(\operatorname{Ker} \Gamma)^{\perp}} e_0.$$

So now we arrive at the following setup. Given a self-adjoint  $\mathcal{R}$  with trivial kernel on  $\mathcal{H}$ , and a vector p with property  $q := \mathcal{R}^{-1}p$  with norm  $||q|| \leq 1$ , then  $\mathcal{R}^2 - pp^* \geq 0$ , and we take a self-adjoint  $\mathcal{R}_1$  such that

$$\mathcal{R}^2 - \mathcal{R}_1^2 = pp^*. \tag{3.1.5}$$

Now applying the Douglas Lemma 3.1.1, there exists a unique contraction  $\mathcal{T}$  such that

$$\mathcal{R}_1 = \mathcal{T}\mathcal{R} = \mathcal{R}\mathcal{T}^*.$$

Thus we can rewrite (3.1.5) as

$$\mathcal{R}(I - \mathcal{T}^*\mathcal{T})\mathcal{R} = \mathcal{R}qq^*\mathcal{R},$$

which implies that

$$I - \mathcal{T}^* \mathcal{T} = qq^* = \langle \cdot, q \rangle q. \tag{3.1.6}$$

Before stating the main result in this chapter, we need to define the *asymptotic* stability of a contraction.

**Definition 3.1.2.** We say an operator  $\mathcal{T}$  is asymptotically stable iff  $\mathcal{T}^n \to 0$  in the strong operator topology as  $n \to \infty$ , i.e. iff for  $\forall x \in \mathcal{H}$ 

$$\lim_{n \mapsto \infty} \|\mathcal{T}^n x\| \to 0$$

Now go back to identity (3.1.4). In order to find a Hankel operator  $\Gamma$  such that  $(\widetilde{\Gamma}, \widetilde{\Gamma}_1, u)$  is unitary equivalent to  $(\mathcal{R}, \mathcal{R}_1, p)$ , we need  $\mathcal{T}$  to be unitary equivalent to  $\widetilde{S}^* := S^*|_{(\text{Ker }\Gamma)^{\perp}}$ , thus we need  $\mathcal{T}$  to be asymptotically stable. In addition, the following proposition shows that this requirement is also sufficient.

**Proposition 3.1.3.** If  $\mathcal{T}$  is asymptotically stable, then there exist a unique self-adjoint Hankel operator  $\Gamma$  and an unitary operator  $\widetilde{\mathcal{V}} : \mathcal{H} \to (\operatorname{Ker} \Gamma)^{\perp}$  such that

$$\widetilde{\Gamma} := \Gamma|_{(\operatorname{Ker} \Gamma)^{\perp}} = \widetilde{\mathcal{V}} \mathcal{R} \widetilde{\mathcal{V}}^*, \qquad (3.1.7)$$

$$\widetilde{\Gamma}_1 := \Gamma_1|_{(\operatorname{Ker} \Gamma)^{\perp}} = \widetilde{\mathcal{V}} \mathcal{R}_1 \widetilde{\mathcal{V}}^*, \qquad (3.1.8)$$

$$\Gamma e_0 = \tilde{\mathcal{V}} p; \tag{3.1.9}$$

note that (3.1.9) implies that

$$P_{(\text{Ker }\Gamma)^{\perp}}e_0 = \widetilde{\mathcal{V}}q, \qquad (3.1.10)$$

And the Hankel coefficients  $\{\gamma_k\}_{k=0}^{\infty}$  can be expressed by

$$\gamma_k = \langle \mathcal{T}^k p, q \rangle.$$

Moreover, Ker  $\Gamma = \{0\}$  if and only if ||q|| = 1 and  $q \notin \operatorname{Ran}\{\mathcal{R}\}$ .

### 3.2 Proof of Proposition 3.1.3

### 3.2.1 Existence of Hankel $\Gamma$

*Proof.* Treating (3.1.6) as an identity for quadratic forms and substituting  $x \in \mathcal{H}$  into it we get

$$||x||^{2} - ||\mathcal{T}x||^{2} = |\langle x, q \rangle|^{2}.$$

Then we substitute x by  $\mathcal{T}^k x$  for an arbitrary  $k \in \mathbb{N}$ , we get

$$\|\mathcal{T}^k x\|^2 - \|\mathcal{T}^{k+1} x\|^2 = |\langle \mathcal{T}^k x, q \rangle|^2,$$

Now taking the sum from k = 0 to k = n we get

$$||x||^{2} - ||\mathcal{T}^{n+1}x||^{2} = \sum_{k=0}^{n} |\langle \mathcal{T}^{k}x, q \rangle|^{2}.$$

Let  $n \to \infty$  and using the asymptotic stability of  $\mathcal{T}$  we see that

$$||x||^2 = \sum_{k=0}^{\infty} |\langle \mathcal{T}^k x, q \rangle|^2,$$

which means that the operator  $V: \mathcal{H} \to \ell^2$ 

$$\mathcal{V}x = \left(\langle x, q \rangle, \langle \mathcal{T}x, q \rangle, \langle \mathcal{T}^2 x, q \rangle, \ldots\right) = \left(\langle \mathcal{T}^k x, q \rangle\right)_{k=0}^{\infty}$$
(3.2.1)

is an isometry.
We can see that

$$\mathcal{VT}x = \left(\langle \mathcal{T}x, q \rangle, \langle \mathcal{T}^2x, q \rangle, \langle \mathcal{T}^3x, q \rangle, \ldots \right) = S^* \mathcal{V}x,$$

so  $\mathcal{T}$  is unitarily equivalent to either  $S^*$  (if  $\operatorname{Ran} \mathcal{V} = \ell^2$ ) or to the restriction of  $S^*$  to Ran  $\mathcal{V}$ , which is a  $S^*$ -invariant subspace (if  $\operatorname{Ran} \mathcal{V} \neq \ell^2$ ). (Here we use a simple fact that an onto isometry is unitary.)

Let  $\widetilde{\mathcal{V}}$  be the operator  $\mathcal{V}$  with the target space restricted to  $\operatorname{Ran} \mathcal{V}$ , so  $\widetilde{\mathcal{V}} : \mathcal{H} \to \operatorname{Ran} \mathcal{V}$ is an unitary operator. Denoting by  $\widetilde{S}^* := S^*|_{\operatorname{Ran} \mathcal{V}}$  (so  $\widetilde{S}^* = S^*$  if  $\operatorname{Ran} \mathcal{V} = \ell^2$ ), we see that

$$\widetilde{\mathcal{V}}\mathcal{T}\widetilde{\mathcal{V}}^* = \widetilde{S}^*, \qquad \widetilde{\mathcal{V}}\mathcal{T}^*\widetilde{\mathcal{V}}^* = (\widetilde{S}^*)^* = P_{\operatorname{Ran}\mathcal{V}}S|_{\operatorname{Ran}\mathcal{V}} =: \widetilde{S}.$$

Define

$$\widetilde{\Gamma} := \widetilde{\mathcal{V}} \mathcal{R} \widetilde{\mathcal{V}}^*, \qquad \widetilde{\Gamma}_1 := \widetilde{\mathcal{V}} \mathcal{R}_1 \widetilde{\mathcal{V}}^*. \tag{3.2.2}$$

Then the relation  $\mathcal{TR} = \mathcal{R}_1 = \mathcal{R}_1^* = \mathcal{RT}^*$  (remember that  $\mathcal{R}$  and  $\mathcal{R}_1$  are self-adjoint) translates to

$$\widetilde{\Gamma}\widetilde{S} = \widetilde{\Gamma}_1 = \widetilde{S}^*\widetilde{\Gamma}.$$

Extending  $\widetilde{\Gamma}$  and  $\widetilde{\Gamma}_1$  to operators  $\Gamma$  and  $\Gamma_1$  on the whole space  $\ell^2$  by setting them to be 0 on  $(\operatorname{Ran} \mathcal{V})^{\perp}$ , we can see that  $\operatorname{Ker} \Gamma = (\operatorname{Ran} V)^{\perp}$  and that

$$\Gamma = \mathcal{VRV}^*, \qquad \Gamma_1 = \mathcal{VR}_1\mathcal{V}^*.$$

Let us now show that  $\Gamma$  satisfies the identity  $\Gamma S = S^*\Gamma$ , i.e. that  $\Gamma$  is indeed a

Hankel operator. For  $\forall f \in \ell^2$  we decompose

$$f = P_{\operatorname{Ran} \mathcal{V}} f + P_{(\operatorname{Ran} \mathcal{V})^{\perp}} f =: f_1 + f_2.$$

We know that  $S^* \operatorname{Ran} \mathcal{V} \subset \operatorname{Ran} \mathcal{V}$ , so  $S(\operatorname{Ran} \mathcal{V})^{\perp} \subset (\operatorname{Ran} \mathcal{V})^{\perp}$ . Therefore  $Sf_2 \perp$ Ran V, so  $\Gamma Sf_2 = 0 = S^*\Gamma f_2$ . As for  $f_1$ , since Ker  $\Gamma = (\operatorname{Ran} \mathcal{V})^{\perp}$ , we have

$$\Gamma S f_1 = \Gamma P_{\operatorname{Ran} \mathcal{V}} S f_1 = \Gamma \widetilde{S} f_1 = \widetilde{\Gamma} \widetilde{S} f_1 = \widetilde{S}^* \widetilde{\Gamma} f_1 = S^* \Gamma f_1,$$

so  $\Gamma$  is indeed a Hankel operator.

Now it remains to show the identities (3.1.7), (3.1.8) and (3.1.9). The first two identities are exactly from (3.2.2). To show (3.1.9), we first derive a representation for  $\mathcal{V}^*$ .

For  $\forall x, y \in \mathcal{H}$ , we have

$$\begin{aligned} \langle \mathcal{V}x, y \rangle &= \sum_{k} \bar{y_k} \langle \mathcal{T}^k x, q \rangle \\ &= \langle x, \sum_{k} y_k (\mathcal{T}^*)^k q \rangle, \end{aligned}$$

thus  $\mathcal{V}^* y = \sum_k y_k(\mathcal{T}^*)^k q$ . In specific, we have  $\mathcal{V}^* e_0 = q$ . In conclusion, we have

$$\Gamma e_0 = \mathcal{VRV}^* \mathcal{V}q = \mathcal{VRq} = \mathcal{Vp} = \mathcal{Vp},$$

and we finish the proof of existence part.

#### **3.2.2** Uniqueness of Hankel $\Gamma$

Proof. Suppose that the identities (3.1.7), (3.1.8), (3.1.9) hold for some unitary operator  $\widetilde{\mathcal{V}} : \mathcal{H} \to \operatorname{Clos} \operatorname{Ran} \Gamma = (\operatorname{Ker} \Gamma)^{\perp} \subset \ell^2$ .

Since for a Hankel operator  $\operatorname{Ker} \Gamma$  is always S-invariant, the subspace  $(\operatorname{Ker} \Gamma)^{\perp}$  is

 $S^*$ -invariant, so the restriction  $S^*|_{(\text{Ker }\Gamma)^{\perp}}$  is well defined. The identities (3.1.7), (3.1.8) and the definition of  $\mathcal{T}$  imply that

$$S^*|_{(\mathrm{Ker}\,\Gamma)^{\perp}} = \widetilde{\mathcal{V}}\mathcal{T}\widetilde{\mathcal{V}}^*. \tag{3.2.3}$$

Now denote  $\widetilde{\Gamma} := \Gamma|_{(\operatorname{Ker} \Gamma)^{\perp}}$  and  $\widetilde{\Gamma}_1 = \Gamma_1|_{(\operatorname{Ker} \Gamma)^{\perp}}$ , we restrict both sides of identity (3.1.1) on  $(\operatorname{Ker} \Gamma)^{\perp}$  and write

$$\widetilde{\Gamma}_1^2 = \widetilde{\Gamma}^2 - uu^*,$$

where

$$u := \Gamma e_0 = \widetilde{\Gamma} P_{(\operatorname{Ker} \Gamma)^{\perp}} e_0,$$

Then identities (3.1.7), (3.1.8) will be unitarily translated to

$$\mathcal{R}_1^2 = \mathcal{R}^2 - \tilde{p}\tilde{p}^*, \quad \text{where} \quad \tilde{p} = \mathcal{R}\tilde{q}, \; \tilde{q} = \widetilde{\mathcal{V}}^* P_{(\text{Ker}\,\Gamma)^{\perp}} e_0.$$

Comparing this with  $\mathcal{R}_1^2 = \mathcal{R}^2 - pp^*$  we conclude that  $p = \alpha \tilde{p}, q = \alpha \tilde{q}$ , for an  $|\alpha| = 1$ .

To compute the Hankel coefficients  $\{\gamma_k\}_{k=1}^{\infty}$  of  $\Gamma$ , we have

$$\gamma_k = (\Gamma e_0, S^k e_0) = \langle (S^*)^k \Gamma e_0, e_0 \rangle = \langle (S^*)^k \Gamma P_{(\operatorname{Ker} \Gamma)^{\perp}} e_0, P_{(\operatorname{Ker} \Gamma)^{\perp}} e_0 \rangle$$
$$= \langle \mathcal{T}^k \mathcal{R} \tilde{q}, \tilde{q} \rangle = \langle \mathcal{T}^k \tilde{p}, \tilde{q} \rangle = \langle \mathcal{T}^k p, q \rangle,$$

meaning that the coefficients  $\{\gamma_k\}_{k=0}^{\infty}$  does not depend on  $\widetilde{\mathcal{V}}$ . So the uniqueness is proved.

#### 3.2.3 Trivial Kernel Condition

Proof. As we discussed in the proof of existence part, Ker  $\Gamma$  is trivial if and only if the operator  $\mathcal{V}$  defined by (3.2.1) satisfies Ran  $V = \ell^2$ . The latter condition is equivalent to  $\widetilde{S}^* = S^*$ , which happens if and only if  $\mathcal{T}$  is unitarily equivalent to  $S^*$ , or, equivalently,  $\mathcal{T}^*$  is unitarily equivalent to S.

For the first direction, suppose Ker  $\Gamma = \{0\}$ , so T is unitarily equivalent to  $S^*$ . We know that

$$I - SS^* = e_0 e_0^*, \qquad I - \mathcal{T}^* \mathcal{T} = qq^*,$$

so by the unitary equivalence  $||q|| = ||e_0|| = 1$ . If  $q \in \operatorname{Ran} \mathcal{R}$ , i.e.  $q = \mathcal{R}f$ , then

$$\mathcal{RT}^*f = \mathcal{TR}f = \mathcal{T}q = 0,$$

and since Ker  $\mathcal{R} = \{0\}$ , we conclude that  $\mathcal{T}^* f = 0$ . But if Ker  $\Gamma = \{0\}$ . then  $\mathcal{T}$  is an isometry, which contradicts  $\mathcal{T}^* f = 0$ . So  $q \notin \operatorname{Ran} \mathcal{R}$ .

For the sufficiency part, suppose ||q|| = 1 and  $q \notin \operatorname{Ran} \mathcal{R}$ . We know that

$$\mathcal{TR}^2\mathcal{T}^* = \mathcal{R}_1^2 = \mathcal{RT}^*\mathcal{TR} = \mathcal{R}(I - qq^*)\mathcal{R},$$

and that  $\operatorname{Ker}(I-qq^*) = \operatorname{Span}\{q\}$ . Since  $q \notin \operatorname{Ran} \mathcal{R}$ , we see that  $\operatorname{Ker} \mathcal{R}(I-qq^*)\mathcal{R} = \{0\}$ , so  $\operatorname{Ker} \mathcal{T}^* = \{0\}$ .

Now applying (3.1.6) to vector q we get that

$$q - \mathcal{T}^* \mathcal{T} q = q_s$$

and since Ker  $\mathcal{T}^* = \{0\}$  we see that  $\mathcal{T}q = 0$ .

Then left and right multiplying (3.1.6) by  $\mathcal{T}$  and  $\mathcal{T}^*$  respectively, we get

$$\mathcal{T}\mathcal{T}^* - \mathcal{T}\mathcal{T}^*\mathcal{T}\mathcal{T}^* = \mathcal{T}qq^*\mathcal{T}^*,$$

and since  $\mathcal{T}q = 0$ , we have  $\mathcal{T}\mathcal{T}^* = (\mathcal{T}\mathcal{T}^*)^2$ , which implies that  $\mathcal{T}\mathcal{T}^*$  is an orthogonal projection. Since Ker  $\mathcal{T}^* = \{0\}$ , we conclude that  $\mathcal{T}\mathcal{T}^* = I$ , i.e.  $\mathcal{T}$  is an isometry.

In addition, we have that  $\mathcal{T}$  is asymptotically stable, thus there is no reduced subspace on which  $\mathcal{T}$  (and so  $\mathcal{T}^*$ ) is unitary. The identity (3.1.6) implies that rank $(I - \mathcal{T}^*\mathcal{T}) = 1$ , i.e. the defect indices of  $\mathfrak{D}_{\mathcal{T}} = 1$ .so  $\mathcal{T}^*$  is unitarily equivalent to the shift operator S, i.e. that

$$S = \mathcal{V}\mathcal{T}^*\mathcal{V}^*$$

for some unitary operator  $\mathcal{V} : \mathcal{H} \to \ell^2$ . Defining  $\Gamma = \widetilde{\mathcal{V}}\mathcal{R}\widetilde{\mathcal{V}}^*$ ,  $\Gamma_1 = \widetilde{\mathcal{V}}\mathcal{R}_1\widetilde{\mathcal{V}}^*$ , we can see that

$$\Gamma_1 = \Gamma S = S^* \Gamma,$$

thus  $\Gamma$  is indeed the Hankel operator with trivial kernel, which satisfies (3.1.7), (3.1.8) and (3.1.9). Since such Hankel  $\Gamma$  that satisfies (3.1.7), (3.1.8) and (3.1.9) is unique, we have finished the proof of trivial kernel condition.

#### Chapter 4

# INTRODUCTION TO CONJUGATION AND $\mathfrak{C}$ -SYMMETRIC OPERATORS

In last chapter, we talk about the inverse spectral problem for self-adjoint Hankel operators. In chapter 4 and chapter 6, we discuss the same topic for a non self-adjoint Hankel operator. In this chapter, we discuss some background definition and properties before stating the main result for a non self-adjoint Hankel operator in chapter 5.

In section 4.1, we introduce the definition of conjugation and a special type of operators called complex symmetric operators, and we show that all Hankel operators are  $\mathfrak{C}$ -symmetric operators where  $\mathfrak{C}$  is the canonical conjugation defined as (4.1.2).

Next in section 4.2 and section 4.3 we will show two propositions that we will frequently use in the following chapters. In section 4.2, we give the description of the generalized polar decomposition (see Theorem 4.2.2) for  $\mathfrak{C}$ -symmetric operators. In section 4.3, we prove a lemma related to rank-one perturbation, stated as below:

**Lemma 4.0.1.** Let a self-adjoint operator  $B = B^*$  be a rank one perturbation of a self-adjoint operator  $A = A^*$ ,

$$B = A - \alpha p p^*, \qquad \alpha \in \mathbb{R}, \ \alpha \neq 0.$$

Then there exists a conjugation  $\mathfrak{J}_p$  commuting with both A and B and preserving p,

 $\mathfrak{J}_p p = p.$ 

Moreover,  $\mathfrak{J}_p|_{\mathcal{H}_0}$  is uniquely determined where  $\mathcal{H}_0 := \overline{\operatorname{Span}}\{A^n p : n \ge 0\}$ . Indeed, we have  $\mathfrak{J}_p A^n p = A^n p$  holds for all  $n \in \mathbb{N}$ .

## 4.1 Definition of Conjugation and Complex Symmetric Operators

To begin with, we need some definitions and preparation work, borrowed from [22].

**Definition 4.1.1.** An operator  $\mathfrak{C}$  in a complex Hilbert space  $\mathcal{H}$  is called a conjugation, if and only if it is

- (i) conjugate-linear:  $\mathfrak{C}(\alpha x + \beta y) = \bar{\alpha}\mathfrak{C}x + \bar{\beta}\mathfrak{C}y$  for all  $x, y \in \mathcal{H}$ ;
- (ii) involutive:  $\mathfrak{C}^2 = I$ ;
- (iii) isometric:  $\|\mathfrak{C}x\| = \|x\|$  for all  $x \in \mathcal{H}$ .

The followings are some typical examples of conjugations.

**Example 4.1.2.** Let  $(X, \mu)$  be a measurable space, then the canonical conjugation  $\mathfrak{C}$ on  $L^2(X, \mu)$  is just the pointwise complex conjugation:

$$(\mathfrak{C}f)x = \overline{f(x)}.$$

Particularly, the canonical conjugation defined on  $\mathbb{C}^n$  is given by

$$\mathfrak{C}(z_1, z_2, ..., z_n) = (\overline{z_1}, \overline{z_2}..., \overline{z_n})$$

**Example 4.1.3.** The Toeplitz conjugation on  $\mathbb{C}^n$  is defined as

$$\mathfrak{C}(z_1, z_2, ..., z_n) = (\overline{z_n}, \overline{z_{n-1}}, ..., \overline{z_1})$$

The following is a well-known result, giving the existence of a set of  $\mathfrak{C}$ -real orthogonal basis  $\{e_n\}_{n=1}^{\infty}$ , stated in [22].

**Definition 4.1.4.** We call a vector  $v \in \mathcal{H}$  a  $\mathfrak{C}$ -real vector if  $\mathfrak{C}v = v$ .

**Lemma 4.1.5.** If  $\mathfrak{C}$  is a conjugation on  $\mathcal{H}$ , then there exists an orthogonal basis  $\{e_n\}_{n=1}^{\infty}$ such that  $\mathfrak{C}e_n = e_n$  holds for all n. In particular,

$$\mathfrak{C}(\sum_n \alpha_n e_n) = \sum_n \overline{\alpha_n} e_n$$

holds for all  $\{\alpha_n\}_{n\geq 0} \in \ell^2$ . And we call such basis  $\{e_n\}_{n=1}^{\infty} \mathfrak{C}$ -real orthogonal basis.

From the definition of  $\mathfrak{C}$ -real orthogonal basis, it's easy to see that for any two vectors  $x, y \in \mathcal{H}$  and any conjugation  $\mathfrak{C}$ , we have

$$\langle \mathfrak{C}x, y \rangle = \langle \mathfrak{C}y, x \rangle.$$
 (4.1.1)

Now we introduce the definition of  $\mathfrak{C}$ -symmetric operators.

**Definition 4.1.6.** Let  $\mathfrak{C}$  be a conjugation on Hilbert space  $\mathcal{H}$ . A bounded linear operator T on  $\mathcal{H}$  is called  $\mathfrak{C}$ -symmetric iff  $T^* = \mathfrak{C}T\mathfrak{C}$ .

If we have an operator T is  $\mathfrak{C}$ -symmetric, and we take a  $\mathfrak{C}$ -real orthogonal basis  $\{e_n\}_{n\geq 0}$ , then we can show that the representation matrix of T with respect to this basis  $\{e_n\}_{n=1}^{\infty}$  is a complex symmetric matrix. In fact,

$$[T]_{ij} = \langle Te_j, e_i \rangle = \langle \mathfrak{C}T^* \mathfrak{C}e_j, e_i \rangle = \langle \mathfrak{C}e_i, T^* \mathfrak{C}e_j \rangle$$
$$= \langle e_i, T^*e_j \rangle = \langle Te_i, e_j \rangle = [T]_{ji}$$

We record here some well-know examples of complex symmetric operators, and their corresponding  $\mathfrak{C}$ -real orthogonal basis.

**Example 4.1.7.** For any Hankel operator  $\Gamma$ , we have  $\Gamma = \mathfrak{C}\Gamma^*\mathfrak{C}$  where here  $\mathfrak{C}$  is the canonical conjugation on  $\ell^2$ :

$$\mathfrak{C}(z_1, z_2, z_3...) = (\overline{z_1}, \overline{z_2}, \overline{z_3}...). \tag{4.1.2}$$

Hence  $\Gamma$  is a  $\mathfrak{C}$ -symmetric operator. Here the  $\mathfrak{C}$ -real orthogonal basis can be simply taken to be the original orthogonal basis  $\{e_n\}_{n=0}^{\infty}$ .

## 4.2 Generalization of Polar Decomposition for C-symmetric operators

In this section we will introduce a generalized polar decomposition for  $\mathfrak{C}$ -symmetric operators. We begin with the definition of partial conjugation.

**Definition 4.2.1** (See [21]). An anti-linear operator  $\mathfrak{J}$  in Hilbert space is called a partial conjugation if  $(\operatorname{Ker} \mathfrak{J})^{\perp}$  is invariant for  $\mathfrak{J}$  and  $\mathfrak{J}|_{(\operatorname{Ker} \mathfrak{J})^{\perp}}$  is a conjugation. Denoting  $\mathfrak{K} := (\operatorname{Ker} \mathfrak{J})^{\perp}$ , then we often say that  $\mathfrak{J}$  is a partial conjugation on  $\mathfrak{K}$ .

Let us recall that an operator U is called a *partial isometry* if its restriction to  $(\operatorname{Ker} U)^{\perp}$  is an isometry; note that  $(\operatorname{Ker} U)^{\perp}$  does not need to be U-invariant. If denoting  $\mathfrak{K} := (\operatorname{Ker} U)^{\perp}$ , we sometimes say that U is a partial isometry on  $\mathfrak{K}$ .

Recall also that any bounded operator T in a Hilbert space admits a unique *polar* decomposition T = U|R|, where  $|R| := (R^*R)^{1/2}$ , and U is a partial isometry with Ker U = Ker T.

Then we have a generalized polar decomposition for complex symmetric operators, given in [21]:

**Theorem 4.2.2.** If T = U|T| is the polar decomposition of a  $\mathfrak{C}$ -symmetric operator T, then  $T = \mathfrak{C}\mathfrak{J}|T|$ , where  $\mathfrak{J}$  is a partial conjugation satisfy  $\operatorname{Ker}\mathfrak{J} = \operatorname{Ker} T$ , which commutes with  $|T| := \sqrt{T^*T}$ . In particular, the partial isometry U is  $\mathfrak{C}$ -symmetric and factors as  $U = \mathfrak{C}\mathfrak{J}$ .

Remark 4.2.3. Note that we can always find a conjugation  $\tilde{\mathfrak{J}}$ , such that  $\tilde{\mathfrak{J}} = \mathfrak{J}$  on  $(\operatorname{Ker} \mathfrak{J})^{\perp}$ . In fact, we can take  $\mathfrak{J}'$  is an arbitrary partial conjugation with support  $\operatorname{Ker} \mathfrak{J}$ , and then define  $\tilde{\mathfrak{J}} := \mathfrak{J} + \mathfrak{J}'$ . With this conjugation  $\tilde{\mathfrak{J}}$ , we can reach several corollaries from Theorem 4.2.2.

Corollary 4.2.4. If T is a bounded  $\mathfrak{C}$ -symmetric operator on  $\mathcal{H}$ , then  $T = \mathfrak{C}\mathfrak{J}|T|$ , where  $\mathfrak{J}$  is a conjugation that commutes with |T|. In addition, the choice of  $\mathfrak{J}$  is unique if and only if Ker  $T = \{0\}$ .

Corollary 4.2.5. If T is a bounded  $\mathfrak{C}$ -symmetric operator, then T = W|T| where W is a  $\mathfrak{C}$ -symmetric unitary operator.

#### 4.3 A Property Related to Rank-one Perturbation

In last chapter, we have shown that a self-adjoint Hankel operator  $\Gamma$  satisfies a rankone perturbation relation:  $\Gamma^2 - \Gamma_1^2 = uu^*$ , where  $\Gamma_1 := \Gamma S$  and  $u = \Gamma e_0$ . For a general Hankel operator  $\Gamma$ , we can derive a similar identity. In fact, since  $\Gamma_1 = \Gamma S = S^*\Gamma$ , we have

$$|\Gamma|^2 - |\Gamma_1|^2 = \Gamma^* \Gamma - \Gamma^* S S^* \Gamma = \Gamma^* (I - S S^*) \Gamma = \Gamma^* (e_0 e_0^*) \Gamma,$$

so denoting  $u := \Gamma^* e_0$ , we have

$$|\Gamma|^2 - |\Gamma_1|^2 = uu^*, \tag{4.3.1}$$

In this section, we will prove the result in lemma 4.0.1, which will be frequently used in later discussion. Applying this lemma to (4.3.1), we can find a conjugation  $\mathfrak{J}_u$ commuting with  $|\Gamma|, |\Gamma_1|$  and preserves u. In addition, this conjugation is unique if uis cyclic with respect to  $|\Gamma|$  in  $\mathcal{H}$ .

Proof of Lemma 4.0.1. To begin with, we need the following lemma.

**Lemma 4.3.1.** (Modified from [24, Theorem 2.2]) Let  $\mathfrak{J}$  be a conjugation on  $L^2(\rho)$ , where  $\rho$  is a Borel measure on  $\mathbb{R}$ , which is compact supported. Then the following two statements are equivalent:

- (i)  $\mathcal{M}_x \mathfrak{J} = \mathfrak{J} \mathcal{M}_x$ , where  $\mathcal{M}_x$  is the operator of multiplication by independent variable;
- (ii) There exists a function  $\phi(x) \in L^{\infty}(\rho)$ ,  $|\phi(x)| = 1 \ \rho$  a.e., such that:  $\Im f = \mathcal{M}_{\phi(x)}\Im_1 f$  holds  $\rho$ -a.e. Here  $\Im_1$  is the canonical conjugation on  $L^2(\rho)$  defined by

$$\mathfrak{J}_1: f(x) \to \overline{f(x)}$$

Proof. We only need to show (i)  $\Longrightarrow$  (ii), while the other direction is easy to check. We know that  $\mathcal{M}_x \mathfrak{J} \mathfrak{J}_1 = \mathfrak{J} \mathcal{M}_x \mathfrak{J}_1 = \mathfrak{J} \mathfrak{J}_1 \mathcal{M}_x$ , hence  $\mathcal{M}_x$  commutes with an unitary operator  $\mathfrak{J} \mathfrak{J}_1$  on  $L^2(\rho)$ . By [Theorem 3.2, [25]],  $\mathfrak{J} \mathfrak{J}_1 = \mathcal{M}_{\phi}$  for a  $\phi(x) \in L^2(\rho)$ ,  $J = \mathcal{M}_{\phi(x)} \mathfrak{J}_1$ . Since  $\mathfrak{J} \mathfrak{J}_1 = \mathcal{M}_{\phi}$  is unitary, we conclude that  $|\phi| = 1 \rho$ -a.e.

Now get back to the proof of Lemma 4.0.1

By Von Neumann's theorem (See Theorem 1.2.7), up to unitary equivalence we can assume that A is the multiplication by independent variable  $\mathcal{M}_x$  on  $L^2(\rho)$ , and p is a function  $f(x) \in L^2(\rho)$ . By Lemma 4.3.1, since  $\mathcal{M}_x \mathfrak{J} = \mathfrak{J} \mathcal{M}_x$ , we have  $\mathfrak{J} = \mathcal{M}_{\phi(x)} \mathfrak{J}_1$  for a certain function  $\phi(x) \in L^{\infty}(\rho)$ ,  $|\phi(x)| = 1 \rho$ - a.e. Then from  $\mathfrak{J}f = f$ , we have

$$\phi(x)\overline{f(x)} = f(x)$$

Thus we can set  $\phi(x) = \frac{f(x)}{f(x)}$  where  $f(x) \neq 0$ , and set  $\phi(x)$  to be any unit value where f(x) = 0. This gives the existence of such conjugation  $\mathfrak{J}_p$ .

To see that  $\mathfrak{J}_p$  is unique on  $\mathcal{H}_0$ , we can show by induction that  $\mathfrak{J}_p A^n p = A^n p$  for all  $n \in \mathbb{N}$ .

In fact, the equation holds trivially when n = 0. Assume that the equation holds

for n = k, then for n = k + 1,

$$\mathfrak{J}_p A^{k+1} p = A \mathfrak{J}_p A^k p = A A^k p = A^{k+1} p,$$

thus  $\mathfrak{J}_p$  is uniquely defined on  $\mathcal{H}_0 = \overline{\operatorname{Span}} \{ A^n p : n \ge 0 \}.$ 

The following lemma gives all such conjugations that satisfy the property in Lemma 4.0.1. It's easy to see that such conjugation  $\mathfrak{J}_p$  is unique if and only if p is a cyclic vector for A.

**Lemma 4.3.2.** If  $\mathfrak{J}_p$  is a conjugation from Lemma 4.0.1, then any other such conjugation  $\mathfrak{J}'_p$  that also satisfies the requirement in Lemma 4.0.1 will be given by  $\mathfrak{J}'_p = \psi \mathfrak{J}_p$ , where  $\psi$  is unitary  $\mathfrak{J}_p$ -symmetric operators commuting with A and preserves p,  $\psi p = p$ .

*Proof.* If  $\mathfrak{J}'_p$  is another conjugation, commuting with A and preserving p, then  $\mathfrak{J}'_p = \psi \mathfrak{J}_p$ , where  $\psi := \mathfrak{J}'_p \mathfrak{J}_p$ . It is east to see that  $\psi$  is a unitary operator. It is also easy to see that  $\psi$  is  $\mathfrak{J}_p$ -symmetric operator commuting with A and that  $\psi p = p$ .

On the other hand, if  $\psi$  is a  $\mathfrak{J}_p$ -symmetric unitary operator, commuting with A and such that  $\psi p = p$ , then the (conjugate-linear) operator  $\mathfrak{J}'_p := \psi \mathfrak{J}_p$  is a conjugation commuting with A and preserving p.

Indeed, the operator  $\mathfrak{J}'_p$  is trivially conjugate-linear, isometric, preserves p and commutes with A (and so with B). To show that  $\mathfrak{J}'_p$  is an involution we use the  $\mathfrak{J}_p$ -symmetry of  $\psi$ :

$$(\mathfrak{J}'_p)^2 = \psi \mathfrak{J}_p \psi \mathfrak{J}_p = \mathfrak{J}_p \psi^* \psi \mathfrak{J}_p = \mathfrak{J}_p^2 = I;$$

here in the second equality we use the fact that  $\psi$  is  $\mathfrak{J}_p$ -symmetric.

Remark 4.3.3. Since  $\psi$  commutes with A and preserves p, it is easy to see that  $\psi|_{\mathcal{H}_0} = I|_{\mathcal{H}_0}$ .

#### Chapter 5

## GENERAL HANKEL OPERATORS AS COMPLEX SYMMETRIC OPERATORS

In chapter 3, we have discussed the inverse spectral problem for a self-adjoint Hankel operator. In this chapter, we will discuss the same topic for a general Hankel operator taken as a  $\mathfrak{C}$ -symmetric operator. We will also show that a Hankel operator can be uniquely determined by the spectral data of two operators satisfying a rank-one perturbation relation.

In section 5.1, we introduce the setting and some preparation work for the game. Briefly speaking, we can find a conjugation  $\mathfrak{J}_u$ , such that this  $\mathfrak{J}_u$  commutes with  $|\Gamma|$  and  $|\Gamma_1|$ , and also preserves  $u := \Gamma^* e_0$ . Then we apply the generalized polar decomposition Theorem 4.2.2 to  $\Gamma, \Gamma_1$ , and restrict  $\Gamma, \Gamma_1$  on the essential part of  $\Gamma$ , thus we can write them as the form in (5.1.5):

$$\mathfrak{C}\widetilde{\Gamma} = \widetilde{\phi}|\widetilde{\Gamma}|\widetilde{\mathfrak{J}}_{u}, \qquad \mathfrak{C}\widetilde{\Gamma}_{1} = \widetilde{\phi}_{1}|\widetilde{\Gamma}_{1}|\widetilde{\mathfrak{J}}_{u}. \tag{5.0.1}$$

We also need to emphasize that the choice of  $\mathfrak{J}_u$  is unique if and only if u is cyclic with respect to  $|\Gamma|$  on  $\mathcal{H}$ . Under this case, the choice of unitary  $\tilde{\phi}$  and partial isometry  $\tilde{\phi}_1$  is also unique.

Next in section 5.2 and section 5.3, we state and prove the main result (Proposition

5.2.3 and Proposition 5.2.4) in this chapter. That is, given a triple  $(R, R_1, \mathfrak{J}_p, \varphi, \varphi_1, p)$  satisfying the following properties

(i)  $R, R_1 \ge 0$ , Ker  $R = \{0\}$ , are self-adjoint operators in  $\mathcal{H}$ , and  $p \in \mathcal{H}$  and such that

$$R^2 - R_1^2 = pp^*.$$

- (ii)  $\mathfrak{J}_p$  is a conjugation commuting with R and  $R_1$  and preserving p,  $\mathfrak{J}_p p = p$ .
- (iii)  $\varphi$  is a  $\mathfrak{J}_p$ -symmetric unitary operator commuting with R.
- (iv)  $\varphi_1$  is a  $\mathfrak{J}_p$ -symmetric partial isometry, Ker  $\varphi_1 = \text{Ker } R_1$ , commuting with  $R_1$ ; note that if Ker  $R_1 = \{0\}$  then  $\varphi_1$  is unitary.
- (v) The contraction  $\mathcal{T} := R_1 \varphi_1 \varphi^* R^{-1}$  is asymptotically stable;

then there exists a unique Hankel operator  $\Gamma$ , such that we can find a conjugation  $\mathfrak{J}_u$  commuting with  $|\Gamma|, |\Gamma_1|$  and preserves u, satisfying that the induced triple  $(|\widetilde{\Gamma}|, |\widetilde{\Gamma}_1|, u, \widetilde{\phi}, \widetilde{\phi}_1, \mathfrak{J}_u)$  is unitary equivalent to  $(R, R_1, p, \varphi, \varphi_1, \mathfrak{J}_p)$ .

Thus we can see that a general Hankel operator can be uniquely determined by the spectral data of two operators  $R\varphi$ ,  $R_1\varphi_1$ , which satisfy a rank-one perturbation relation.

In addition, we can also further translate the spectral data of  $R\varphi$ ,  $R_1\varphi_1$  under the compact case.

Denoting  $\mathfrak{H}_{\mathfrak{o}} := \overline{\operatorname{Span}}\{|\widetilde{\Gamma}|^n u | n \geq 0\}$ . Then Proposition 6.2.2 implies that there exists two positive real sequences  $\{\lambda_k\}_{k=1}^{\infty}, \{\mu_k\}_{k=1}^{\infty}$ , such that the non-zero eigenvalues of  $|\widetilde{\Gamma}||_{\mathfrak{H}_{\mathfrak{o}}}, |\widetilde{\Gamma}_1||_{\mathfrak{H}_{\mathfrak{o}}}$  are simple, and coincide with  $\{\lambda_k\}_{k=1}^{\infty}, \{\mu_k\}_{k=1}^{\infty}$  respectively.

We can further show that  $|\widetilde{\Gamma}|, |\widetilde{\Gamma}_1|$  has no non-zero eigenvalues other than  $\{\lambda_k\}_{k=1}^{\infty}$ ,

 $\{\mu_k\}_{k=1}^{\infty}$ . Thus we can write all eigenspaces of  $|\widetilde{\Gamma}|, |\widetilde{\Gamma}_1|$  as the following:

$$\begin{split} \mathfrak{E}^{1}_{\lambda_{k}} &:= \operatorname{Ker}(|\widetilde{\Gamma}_{1}| - \lambda_{k}I), \qquad \mathfrak{E}_{\lambda_{k}} &:= \operatorname{Ker}(|\widetilde{\Gamma}| - \lambda_{k}I); \\ \mathfrak{E}^{1}_{\mu_{k}} &:= \operatorname{Ker}(|\widetilde{\Gamma}_{1}| - \mu_{k}I), \qquad \mathfrak{E}_{\mu_{k}} &:= \operatorname{Ker}(|\widetilde{\Gamma}| - \mu I), \end{split}$$

For the non self-adjoint compact  $\Gamma$ , we can also take a canonical choice of tuple  $(\mathfrak{J}_u, \widetilde{\phi}, \widetilde{\phi}_1)$ , which is given by Lemma 6.3.2, together with the equivalent condition of asymptotic stability given in Lemma 7.6.3, we have the structure of  $\varphi$  on  $\mathfrak{E}_{\lambda_k}, \mathfrak{E}_{\mu_k}$ , and the structure of  $\varphi_1$  on  $\mathfrak{E}^1_{\lambda_k}, \mathfrak{E}^1_{\mu_k}$ , which is given as follow:

- (i)  $\widetilde{\phi}|_{\mathfrak{E}_{\mu_k}} = I, \qquad \widetilde{\phi}_1|_{\mathfrak{E}^1_{\lambda_k}} = I;$
- (ii)  $u_{\lambda_k}$  is a \*-cyclic vector for  $\widetilde{\phi}|_{\mathfrak{E}_{\lambda_k}}$ , where  $u_{\lambda_k} := P_{\mathfrak{E}_{\lambda_k}} u$ ;
- (iii)  $u_{\mu_k}$  is a \*-cyclic vector for  $\widetilde{\phi}_1|_{\mathfrak{C}^1_{\mu_k}}$ , where  $u_{\mu_k} := P_{\mathfrak{C}^1_{\mu_k}} u$ .

Now with the preparations above, we consider two different situations under the compact case:

(i)  $\mathfrak{H}_{\mathfrak{o}} = \ell^2$ . Then we have  $\mathfrak{E}_{\mu_k}, \mathfrak{E}^1_{\lambda_k}$  are trivial, and  $\mathfrak{E}_{\lambda_k}, \mathfrak{E}^1_{\mu_k}$  are of dimension 1. Under this situation, we have the choice of  $\mathfrak{J}_u$  is unique, and we conclude that  $\Gamma$  will be uniquely determined by the eigenvalues of two operators  $|\widetilde{\Gamma}|\widetilde{\phi}, |\widetilde{\Gamma}_1|\widetilde{\phi}_1$ . In other words, given two sequences of complex numbers  $\{\lambda_n\}_{n=1}^{\infty}, \{\mu_n\}_{n=1}^{\infty}$ , whose modulus satisfy an intertwining relation:

$$|\lambda_1| > |\mu_1| > |\lambda_2| > |\mu_2| > \dots > |\lambda_n| > |\mu_n| > \dots \to 0,$$

we can find a compact Hankel operator  $\Gamma$  with simple singular values, with the uniquely determined tuple  $(\mathfrak{J}_u, \phi, \phi_1)$ , satisfying that the non-zero eigenvalues of  $|\Gamma|\phi, |\Gamma_1|\phi_1$  are simple, and coincide with  $\{\lambda_n\}_{n=1}^{\infty}$ ,  $\{\mu_n\}_{n=1}^{\infty}$  respectively.

The detail of work under this situation can be found in section 9.2 in chapter 8. (ii)  $\mathfrak{H}_{\mathfrak{o}} \subsetneq \ell^2$ , then we can define a measure  $\tilde{\rho}_k(s)$  to be the scalar spectral measure of  $\widetilde{u}_{\lambda_k} := \frac{u_{\lambda_k}}{\|u_{\lambda_k}\|}$  with respect to  $\widetilde{\phi}|_{\mathfrak{E}_{\lambda_k}}$ . And we also similarly define  $\widetilde{\rho}_k^1(s)$  to be the scalar spectral measure of  $\widetilde{\mu}_k := \frac{u_{\mu_k}}{\|u_{\mu_k}\|}$  with respect to  $\widetilde{\phi}_1|_{\mathfrak{E}^1_{\mu_k}}$ :

$$\langle (\widetilde{\phi} - zI)^{-1} \widetilde{u}_{\lambda_k}, \widetilde{u}_{\lambda_k} \rangle = \int_{\mathbb{T}} \frac{d\widetilde{\rho}_k(s)}{s - z}, \qquad \langle (\widetilde{\phi}_1 - zI)^{-1} \widetilde{u}_{\mu_k}, \widetilde{u}_{\mu_k} \rangle = \int_{\mathbb{T}} \frac{d\widetilde{\rho}_k^1(s)}{s - z}$$

We conclude that  $\tilde{\rho}_k(s), \tilde{\rho}_k^1(s)$  are both positive discrete probability measures (see Proposition 10.1.2), and we conclude that the four sequences  $\{\lambda_n\}_{n=1}^{\infty}, \{\mu_n\}_{n=1}^{\infty}, \{\tilde{\rho}_k\}_{k=1}^{\infty}, \{\tilde{\rho}_k\}_{k=1}^{\infty}$  will uniquely determine a Hankel operator (see Theorem 10.1.5).

The specific description of the theorem can be found in chaper 9.

#### 5.1 Setup

Given a bounded Hankel operator  $\Gamma$ , we have mentioned in Example 4.1.7 that  $\Gamma$  and  $\Gamma_1 := S^*\Gamma$  are  $\mathfrak{C}$ -symmetric, thus applying Theorem 4.2.2, there exists partial conjugations  $\mathfrak{J}$  and  $\mathfrak{J}_1$  commuting with  $|\Gamma|$  and  $|\Gamma_1|$  respectively, such that Ker  $\mathfrak{J} =$ Ker  $\Gamma$ , Ker  $\mathfrak{J}_1 =$  Ker  $\Gamma_1$  and

$$\Gamma = \mathfrak{C}\mathfrak{J}[\Gamma], \qquad \Gamma_1 = \mathfrak{C}\mathfrak{J}_1[\Gamma_1], \qquad (5.1.1)$$

where here  $\mathfrak{J}, \mathfrak{J}_1$  commutes with  $|\Gamma|, |\Gamma_1|$  respectively.

Now recall that we have derived in (4.3.1) that

$$|\Gamma|^2 - |\Gamma_1|^2 = uu^*, \tag{5.1.2}$$

where  $u = \Gamma^* e_0$ . We apply Lemma 4.0.1 to (4.3.1) with  $A = |\Gamma|^2, B = |\Gamma_1|^2, \alpha = 1$ , then we have that there exists a conjugation  $\mathfrak{J}_u$  commuting with  $|\Gamma|^2$  and  $|\Gamma_1|^2$  and such that  $\mathfrak{J}_u u = u$ .

We want to show that  $\mathfrak{J}_u$  also commutes with  $|\Gamma|$  and  $|\Gamma_1|$ . We need the following short lemma.

**Lemma 5.1.1.** Let R be a bounded, self-adjoint positive operator on Hilbert space  $\mathcal{H}$ , and  $\mathfrak{J}$  is a conjugation on  $\mathcal{H}$ . Then the following two statements are equivalent:

- (i) R commutes with  $\mathfrak{J}$ ;
- (ii)  $R^2$  commutes with  $\mathfrak{J}$ .

*Proof.* (i)  $\implies$  (ii) is trivial, we only need to prove the other direction.

Since  $R^2$  commutes with  $\mathfrak{J}$ , for any polynomial  $\mathfrak{p}$  with real coefficients,  $\mathfrak{p}(R^2)$  also commutes with  $\mathfrak{J}$ :  $\mathfrak{p}(R^2)\mathfrak{J} = \mathfrak{J}\mathfrak{p}(R^2)$ . Since the scalar spectrum measure  $\sigma(R^2)$  is compact supported, we can take a polynomial sequence  $\{\mathfrak{p}_n(x)\}_{n=1}^{\infty}$  which converges uniformly to  $\phi(x) = \sqrt{x}$  on  $\sigma(R^2)$ . Then by  $\phi(R^2)\mathfrak{J} = \mathfrak{J}\phi(R^2)$  we have

$$R\mathfrak{J} = \mathfrak{J}R$$

Now back to identity (5.1.2), we know that  $\mathfrak{J}_u$  commutes with  $|\Gamma|, |\Gamma_1|$ . Thus we can rewrite the polar decomposition form in (5.1.1) as

$$\Gamma = \mathfrak{C}\phi\mathfrak{J}_u|\Gamma| = \mathfrak{C}\phi|\Gamma|\mathfrak{J}_u, \qquad \Gamma_1 = \mathfrak{C}\phi_1\mathfrak{J}_u|\Gamma_1| = \mathfrak{C}\phi_1|\Gamma_1|\mathfrak{J}_u, \qquad (5.1.3)$$

where  $\phi := \mathfrak{JJ}_u$ ,  $\phi_1 := \mathfrak{J}_1\mathfrak{J}_u$  are partial isometries, commuting with  $|\Gamma|$  and  $|\Gamma_1|$  respectively. Since  $\mathfrak{J}_u$  commutes with  $|\Gamma|$  and with  $|\Gamma_1|$ , both Ker  $\Gamma =$  Ker  $|\Gamma|$  and Ker  $\Gamma_1 =$  Ker  $|\Gamma_1|$  are invariant for  $\mathfrak{J}_u$ , so Ker  $\phi =$  Ker  $\Gamma$ , Ker  $\phi_1 =$  Ker  $\Gamma_1$ .

We can rewrite those above identities (5.1.3) as

$$\mathfrak{C}\Gamma = \phi|\Gamma|\mathfrak{J}_u, \qquad \mathfrak{C}\Gamma_1 = \phi_1|\Gamma_1|\mathfrak{J}_u. \tag{5.1.4}$$

Since  $\mathfrak{J}_u$  commute with  $|\Gamma|$ , both Ker  $\Gamma = \text{Ker} |\Gamma|$  and  $(\text{Ker} \Gamma)^{\perp}$  are invariant for  $\mathfrak{J}_u$ , and therefore  $\mathfrak{J}_u(\text{Ker} \Gamma)^{\perp} = (\text{Ker} \Gamma)^{\perp}$ . Now restricting everything to  $(\operatorname{Ker} \Gamma)^{\perp}$  and denoting  $\widetilde{\Gamma} := \Gamma |_{(\operatorname{Ker} \Gamma)^{\perp}}, \widetilde{\Gamma}_1 := \Gamma_1 |_{(\operatorname{Ker} \Gamma)^{\perp}}, \widetilde{\mathfrak{J}}_u := \mathfrak{J}_u |_{(\operatorname{Ker} \Gamma)^{\perp}}, \widetilde{\phi}_1 = \phi_1 |_{(\operatorname{Ker} \Gamma)^{\perp}}$  we can see that

$$\mathfrak{C}\widetilde{\Gamma} = \widetilde{\phi}|\widetilde{\Gamma}|\widetilde{\mathfrak{J}}_{u}, \qquad \mathfrak{C}\widetilde{\Gamma}_{1} = \widetilde{\phi}_{1}|\widetilde{\Gamma}_{1}|\widetilde{\mathfrak{J}}_{u}. \tag{5.1.5}$$

Remark 5.1.2. We can easily see that (5.1.5) is well-defined. In fact, though  $(\text{Ker }\Gamma)^{\perp}$  is not necessarily an invariant subspace for  $\Gamma$ , it is a reducing subspace for  $|\Gamma|$ , and since  $u \in \text{Ran }\Gamma^* \perp \text{Ker }\Gamma$ , it is also reducing for  $|\Gamma_1|^2$ , and so for  $|\Gamma_1|$ . It is also invariant for  $\phi_1$  and for  $\mathfrak{J}_u$ , so everything in the right hand sides of (5.1.5) is well defined, which means that  $(\text{Ker }\Gamma)^{\perp}$  is invariant for  $\mathfrak{C}\Gamma$  and for  $\mathfrak{C}\Gamma_1$ .

Now back to (5.1.5). Since  $\mathfrak{J}_u(\operatorname{Ker} \Gamma)^{\perp} = (\operatorname{Ker} \Gamma)^{\perp}$ ,  $\widetilde{\mathfrak{J}}_u$  is a conjugation on  $(\operatorname{Ker} \Gamma)^{\perp}$ , commuting with  $|\widetilde{\Gamma}|$  and preserving u (note that  $u = \Gamma^* e_0 \in \operatorname{Ran} |\Gamma|$ ). We can see that  $\widetilde{\phi}$  is unitary since  $\operatorname{Ker} \phi = \operatorname{Ker} \Gamma$ . Note also that the operator  $\widetilde{\Gamma}_1$  can have a non-trivial kernel (at most one-dimensional), and the action of  $\widetilde{\phi}_1$  on this kernel does not matter in (5.1.5). So, let us assume for definiteness that  $\widetilde{\phi}_1|_{\operatorname{Ker}\widetilde{\Gamma}_1} = 0$ , i.e. that  $\widetilde{\phi}_1$  is a partial isometry with  $\operatorname{Ker} \widetilde{\phi}_1 = \operatorname{Ker} \widetilde{\Gamma}_1$ .

Remark 5.1.3. As it was shown above, see Lemma 4.3.2, the conjugation  $\tilde{\mathfrak{J}}_u$  and the operators  $\phi$ ,  $\phi_1$  in (5.1.5) are not unique, and are defined up to equivalence. Namely, the two triples  $(\tilde{\mathfrak{J}}_u, \tilde{\phi}, \tilde{\phi}_1)$  and  $(\tilde{\mathfrak{J}}'_u, \tilde{\phi}', \tilde{\phi}'_1)$  are in the same equivalent class if there exists an  $\tilde{\mathfrak{J}}_u$ -symmetric unitary operator  $\psi$  commuting with  $|\tilde{\Gamma}|$  and preserving  $u, \ \psi u = u$ , such that

$$\widetilde{\mathfrak{J}}'_u = \psi \widetilde{\mathfrak{J}}_u, \qquad \widetilde{\phi}' = \widetilde{\phi} \psi^*, \qquad \widetilde{\phi}'_1 = \widetilde{\phi}_1 \psi^*.$$

We can easily check that (5.1.5) is still true if we substitute  $(\tilde{\mathfrak{J}}_u, \tilde{\phi}, \tilde{\phi}_1)$  by  $(\tilde{\mathfrak{J}}'_u, \tilde{\phi}', \tilde{\phi}'_1)$ :

$$\mathfrak{C}\widetilde{\Gamma} = \widetilde{\phi}' |\widetilde{\Gamma}| \widetilde{\mathfrak{J}}'_u, \qquad \mathfrak{C}\widetilde{\Gamma}_1 = \widetilde{\phi}'_1 |\widetilde{\Gamma}_1| \widetilde{\mathfrak{J}}'_u$$

Now we give an expression of the Hankel coefficients of  $\Gamma$ , i.e.  $\{\gamma_k\}_{k=0}^{\infty}$ , implying

that  $\Gamma$  could be uniquely determined by  $|\widetilde{\Gamma}|, u, \widetilde{\mathfrak{J}}_u, \widetilde{\phi}, \widetilde{\phi}_1$ . In other words, the uniqueness of  $\Gamma$  only depends on the essential part of  $\Gamma$ , i.e.  $\Gamma|_{(\operatorname{Ker}\Gamma)^{\perp}}$ .

**Proposition 5.1.4.** Given a Hankel operator  $\Gamma$ , with the tuple  $(|\widetilde{\Gamma}|, |\widetilde{\Gamma}_1|, u, \widetilde{\phi}, \widetilde{\phi}_1, \mathfrak{J}_u)$ defined in (5.1.5). Then the Hankel coefficients  $\{\gamma_k\}_{k=0}^{\infty}$  can be expressed as

$$\gamma_k = \langle (\mathfrak{S}^*)^k u, v \rangle, \qquad (5.1.6)$$

where  $\mathfrak{S}^* := |\widetilde{\Gamma}_1|\widetilde{\phi}_1^*\widetilde{\phi}|\widetilde{\Gamma}|^{-1}, v := \mathfrak{J}_u e_0$ , and here  $\widetilde{S}^*$  is the restriction of backward shift on  $(\operatorname{Ker} \Gamma)^{\perp}$ , *i.e.* 

$$\widetilde{S}^* := S^*|_{(\operatorname{Ker} \Gamma)^{\perp}}$$

*Proof.* From  $\Gamma_1 = S^*\Gamma$  together with (5.1.4), we have

$$\mathfrak{C}\phi_1|\Gamma_1|\mathfrak{J}_u = S^*\mathfrak{C}\phi|\Gamma|\mathfrak{J}_u$$

Thus

$$\phi_1[\Gamma_1] = S^* \phi[\Gamma]. \tag{5.1.7}$$

Restricted on  $(\operatorname{Ker} \Gamma)^{\perp}$  and denote  $\widetilde{S}^* := S^*|_{(\operatorname{Ker} \Gamma)^{\perp}}$ , we can rewrite it as

$$|\widetilde{\Gamma}_1|\widetilde{\phi}_1 = \widetilde{S}^*|\widetilde{\Gamma}|\widetilde{\phi}, \qquad \widetilde{S}^* = |\widetilde{\Gamma}_1|\widetilde{\phi}_1\widetilde{\phi}^*|\widetilde{\Gamma}|^{-1}$$

Now using the definition of  $\mathfrak{S}^*$ , since we have  $\phi, \phi_1$  are  $\mathfrak{J}_u$ -symmetric, we have

$$\mathfrak{S}^*\widetilde{\mathfrak{J}}_u = \widetilde{\mathfrak{J}}_u |\widetilde{\Gamma}_1| \widetilde{\phi}_1 \widetilde{\phi}^* |\widetilde{\Gamma}|^{-1} = \widetilde{\mathfrak{J}}_u \widetilde{S}^*$$

thus  $\widetilde{\mathfrak{S}}^* = \widetilde{\mathfrak{J}}_u \widetilde{S}^* \widetilde{\mathfrak{J}}_u$ , and  $\mathfrak{S} = (\mathfrak{S}^*)^*$  satisfies

$$|\widetilde{\Gamma}_1|\widetilde{\phi}_1=\widetilde{S}^*|\widetilde{\Gamma}|\widetilde{\phi}=|\widetilde{\Gamma}|\widetilde{\phi}\mathfrak{S}$$

Now using  $v = \mathfrak{J}_u e_0 = \mathfrak{J}_u \mathfrak{C} e_0 = \mathfrak{J}_u \mathfrak{C} \mathfrak{J}_u v$ , and since  $(\text{Ker } \Gamma)^{\perp}$  is an invariant subspace for  $|\Gamma|, \phi$ , we can write

$$u = \Gamma^* e_0 = \Gamma^* \mathfrak{J}_u v$$
  
=  $(|\Gamma| \mathfrak{JC}) \mathfrak{J}_u (\mathfrak{J}_u \mathfrak{CJ}_u v)$   
=  $|\Gamma| \mathfrak{JJ}_u v = |\Gamma| \phi v = |\widetilde{\Gamma}| \widetilde{\phi} v.$  (5.1.8)

Now using (4.1.1), we can express the Hankel symbol  $\gamma_k$  as

$$\gamma_{k} = \langle \Gamma e_{k}, e_{0} \rangle = \langle S^{k} e_{0}, u \rangle = \langle \mathfrak{J}_{u}^{2} e_{0}, (S^{*})^{k} u \rangle$$
$$= \langle \mathfrak{J}_{u}(S^{*})^{k} u, v \rangle = \langle \widetilde{\mathfrak{J}}_{u}(\widetilde{S}^{*})^{k} u, v \rangle$$
$$= \langle (\mathfrak{S}^{*})^{k} u, v \rangle.$$
(5.1.9)

We will use this equation (5.1.6) in a later proof.

#### 5.2 Plan of the Game and Main Result

Given a Hankel operator  $\Gamma$ , in section 5.1 we have setup a tuple  $(|\Gamma|, |\Gamma_1|, \tilde{\mathfrak{J}}_u, \tilde{\phi}, \tilde{\phi}_1, u)$  with the following properties:

- (i)  $|\widetilde{\Gamma}|, |\widetilde{\Gamma}_1|$  are self-adjoint positive operators. In addition,  $|\widetilde{\Gamma}|$  has trivial kernel;
- (ii)  $|\widetilde{\Gamma}|^2 |\widetilde{\Gamma}_1|^2 = uu^*;$

(iii)  $\tilde{\mathfrak{J}}_u$  is a conjugation which commutes with  $|\tilde{\Gamma}|, |\tilde{\Gamma}_1|$  and preserves u, i.e.  $\tilde{\mathfrak{J}}_u u = u$ ;

- (iv)  $\phi$  is a  $\mathfrak{J}_u$ -symmetric unitary operator which commutes with  $|\widetilde{\Gamma}|$ ;
- (v)  $\widetilde{\phi}_1$  is a  $\widetilde{\mathfrak{J}}_u$ -symmetric partial unitary operator which commutes with  $|\widetilde{\Gamma}_1|$ .

Now we consider the inverse direction of the problem. Assume that we are given a tuple  $(R, R_1, \mathfrak{J}_p, \varphi, \varphi_1, p)$ , which shares similar properties as tuple  $(|\Gamma|, |\Gamma_1|, \tilde{\mathfrak{J}}_u, \tilde{\phi}, \tilde{\phi}_1, u)$ :

(i)  $R, R_1 \ge 0$ , Ker  $R = \{0\}$ , are self-adjoint operators in  $\mathcal{H}$ , and  $p \in \mathcal{H}$  and such that

$$R^2 - R_1^2 = pp^*.$$

Note that the above identity implies that  $p \in \operatorname{Ran} R$  and  $||R^{-1}p|| \leq 1$ .

- (ii)  $\mathfrak{J}_p$  is a conjugation commuting with R and  $R_1$  and preserving p,  $\mathfrak{J}_p p = p$ .
- (iii)  $\varphi$  is a  $\mathfrak{J}_p$ -symmetric unitary operator commuting with R.
- (iv)  $\varphi_1$  is a  $\mathfrak{J}_p$ -symmetric partial isometry, Ker  $\varphi_1 = \text{Ker } R_1$ , commuting with  $R_1$ ; note that if Ker  $R_1 = \{0\}$  then  $\varphi_1$  is unitary.

*Remark* 5.2.1. The first thing to notice is that the dimension of Ker  $R_1$  is at most 1. This can been easily seen from a later Lemma 6.2.1.

The second thing is that  $\varphi_1|_{(\operatorname{Ker} R_1)^{\perp}}$  is unitary, implied by (iv). In fact, since we already have  $\varphi_1|_{(\operatorname{Ker} R_1)^{\perp}}$  is isometry, it suffices to show that  $\varphi_1|_{(\operatorname{Ker} R_1)^{\perp}}$  is onto. From equation  $\varphi_1 \mathfrak{J}_p = \mathfrak{J}_p \varphi_1^*$ , since  $\operatorname{Ran} \varphi_1^* = (\operatorname{Ker} R_1)^{\perp}$  and  $(\operatorname{Ker} R_1)^{\perp}$  is a reducing subspace for  $\mathfrak{J}_p$ , we have  $(\operatorname{Ker} R_1)^{\perp} \subseteq \operatorname{Ran}(\mathfrak{J}_p \varphi_1^*)$ , hence  $(\operatorname{Ker} R_1)^{\perp} \subseteq \operatorname{Ran} \varphi_1$ .

We want to know whether we can find a Hankel operator  $\Gamma$  and choose an appropriate a partial conjugation  $\mathfrak{J}_u$ , which commutes with  $|\Gamma|$  and preserving  $u = \Gamma^* e_0$  such that the tuple  $(|\widetilde{\Gamma}|, |\widetilde{\Gamma}_1|, \widetilde{\mathfrak{J}}_u, \widetilde{\phi}, \widetilde{\phi}_1, u)$  defined in equation (5.1.5) is unitary equivalent to  $(R, R_1, \mathfrak{J}_p, \varphi, \varphi_1, p)$ , i.e., the following identities hold

$$|\widetilde{\Gamma}| = \widetilde{\mathcal{V}}R\widetilde{\mathcal{V}}^*, \qquad |\widetilde{\Gamma}_1| = \widetilde{\mathcal{V}}R_1\widetilde{\mathcal{V}}^*;$$
(5.2.1)

$$\widetilde{\phi} = \widetilde{\mathcal{V}}\varphi\widetilde{V}^*, \qquad \widetilde{\phi}_1 = \widetilde{\mathcal{V}}\varphi_1\widetilde{\mathcal{V}}^*, \qquad \widetilde{\mathfrak{J}}_u = \widetilde{\mathcal{V}}\mathfrak{J}_p\widetilde{\mathcal{V}}^*, \qquad u = \widetilde{\mathcal{V}}p, \qquad (5.2.2)$$

for some unitary operator  $\widetilde{\mathcal{V}} : \mathcal{H} \to (\operatorname{Ker} \Gamma)^{\perp}$ .

Remark 5.2.2. For the case when p is a cyclic vector with respect to R in  $\mathcal{H}$ , then by Lemma 4.3.2, we know that  $\mathfrak{J}_p$  is uniquely defined. (Thus with the unitary equivalence relation in (5.2.1) and (5.2.2), we have u is a cyclic vector with respect to  $|\widetilde{\Gamma}|$  on (Ker  $\Gamma$ )<sup> $\perp$ </sup>). In this case, we can write  $\varphi, \varphi_1$  as  $f(R), f_1(R_1)$  for some Borel measurable, unimodular functions  $f, f_1$ .

In this chapter, we will prove the following two propositions as our main results.

**Proposition 5.2.3.** If contraction  $\mathcal{T} := \varphi_1 R_1 R^{-1} \varphi^*$  (see Douglas Lemma 3.1.1) is asymptotically stable (see definition 3.1.2), then there exists a unique Hankel operator  $\Gamma$  such that

$$\mathfrak{C}\widetilde{\Gamma} = \widetilde{\mathcal{V}}R\varphi\mathfrak{J}_p\widetilde{\mathcal{V}}^*,\tag{5.2.3}$$

$$\mathfrak{C}\widetilde{\Gamma}_1 = \widetilde{\mathcal{V}}R_1\varphi_1\mathfrak{J}_p\widetilde{\mathcal{V}}^*, \qquad (5.2.4)$$

$$\Gamma^* e_0 = \widetilde{\mathcal{V}} p \tag{5.2.5}$$

for some unitary operator  $\widetilde{\mathcal{V}} : \mathcal{H} \to (\operatorname{Ker} \Gamma)^{\perp}$ ; here, recall  $\mathfrak{C}$  is the standard conjugation on  $\ell^2$  defined by (4.1.2).

The coefficients  $\{\gamma_k\}_{k=1}^{\infty}$  of the Hankel operator  $\Gamma$  can be calculated as

$$\gamma_k = \langle \mathfrak{T}^k p, q \rangle. \tag{5.2.6}$$

where  $\mathfrak{T} = \varphi_1^* R_1 R^{-1} \varphi = \mathfrak{J}_p \mathcal{T} \mathfrak{J}_p$  and  $q := R^{-1} \varphi^* p = \varphi^* R^{-1} p$ .

Furthermore, Ker  $\Gamma = \{0\}$  if and only if ||q|| = 1 and  $q \notin \operatorname{Ran} R$  (recall that  $||R^{-1}p|| \leq 1$ , so  $||q|| \leq 1$ ).

**Proposition 5.2.4.** The identities (5.2.3), (5.2.4), and (5.2.5) are equivalent to the unitary equivalence of the tuples  $(|\widetilde{\Gamma}|, |\widetilde{\Upsilon}_1|, \widetilde{\mathfrak{J}}_u, \widetilde{\phi}, \widetilde{\phi}_1, u)$  and  $(R, R_1, \mathfrak{J}_p, \varphi, \varphi_1, p)$ , i.e. to the identities (5.2.1), (5.2.2). (In other words, there exists a conjugation  $\mathfrak{J}_u$  defining on

 $\mathcal{H}$ , such that we can construct a tuple  $(|\widetilde{\Gamma}|, |\widetilde{\Gamma}_1|, \widetilde{\mathfrak{J}}_u, \widetilde{\phi}, \widetilde{\phi}_1, u)$  which is unitary equivalent to  $(R, R_1, \mathfrak{J}_p, \varphi, \varphi_1, p)$ , and also satisfying (5.1.5)).

We first show that the two propositions above are equivalent, then we will present the proof of Proposition 5.2.3 in the next subsection.

*Proof.* Substituting (5.2.1) and (5.2.2) into (5.1.5) we immediately get (5.2.3), (5.2.4), and (5.2.5), so we only need to prove the other direction.

Now Assuming (5.2.3), (5.2.4) and (5.2.5), let us show that

$$|\widetilde{\Gamma}| = \widetilde{\mathcal{V}}R\widetilde{\mathcal{V}}^*, \qquad |\widetilde{\Gamma}_1| = \widetilde{\mathcal{V}}R_1\widetilde{\mathcal{V}}^*.$$

We first show that  $\widetilde{\Gamma}^* = \widetilde{\mathcal{V}}\mathfrak{J}_p \varphi^* R(\widetilde{\mathcal{V}})^* \mathfrak{C}$ . Indeed,

$$\begin{split} \langle \widetilde{\Gamma}x, y \rangle &= \langle \mathfrak{C}\widetilde{\mathcal{V}}R\varphi \mathfrak{J}_p \widetilde{\mathcal{V}}^* x, y \rangle = \langle \mathfrak{C}y, \widetilde{\mathcal{V}}R\varphi \mathfrak{J}_p \widetilde{\mathcal{V}}^* x \rangle \\ &= \langle \varphi^* R \widetilde{\mathcal{V}}^* \mathfrak{C}y, \mathfrak{J}_p \widetilde{\mathcal{V}}^* x \rangle = \langle \widetilde{\mathcal{V}}^* x, \mathfrak{J}_p \varphi^* R \widetilde{\mathcal{V}}^* \mathfrak{C}y \rangle \\ &= \langle x, \widetilde{\mathcal{V}}^* \mathfrak{J}_p \varphi^* R \widetilde{\mathcal{V}}^* \mathfrak{C}y \rangle \end{split}$$

Thus  $\widetilde{\Gamma}^* = \widetilde{\mathcal{V}}^* \mathfrak{J}_p \varphi^* R \widetilde{\mathcal{V}}^* \mathfrak{C}$ , and we have

$$\begin{split} |\widetilde{\Gamma}| &= \left( (\widetilde{\Gamma})^* \widetilde{\Gamma} \right)^{1/2} \\ &= \left( \widetilde{\mathcal{V}} \mathfrak{J}_p \varphi^* R^2 \varphi \mathfrak{J}_p (\widetilde{\mathcal{V}})^* \right)^{1/2} \\ &= \left( \widetilde{\mathcal{V}} R^2 \widetilde{\mathcal{V}}^* \right)^{1/2} = \widetilde{\mathcal{V}} R \widetilde{\mathcal{V}}^*. \end{split}$$

Similarly we can get  $|\tilde{\Gamma}_1| = \tilde{\mathcal{V}}R_1\tilde{\mathcal{V}}^*$ . The fact that  $\varphi_1$  is only a partial isometry does not spoil anything; since  $\varphi_1$  commutes with  $R_1$  and Ker  $\varphi_1 = \text{Ker } R_1$ , we have  $\varphi_1^*R_1\varphi = R_1$ , and the rest of the computations follows exactly as for the case of  $|\Gamma|$ .

Next we define a conjugation  $\widetilde{\mathfrak{J}}_u$  on  $(\operatorname{Ker} \Gamma)^{\perp} = \operatorname{Ran} \widetilde{\mathcal{V}}$  by setting  $\widetilde{\mathfrak{J}}_u := \widetilde{\mathcal{V}} \mathfrak{J}_p \widetilde{\mathcal{V}}^*$ . Easy to see that  $\widetilde{\mathfrak{J}}_u$  commutes with  $|\widetilde{\Gamma}| = \widetilde{\mathcal{V}} R \widetilde{\mathcal{V}}^*$ ,  $|\widetilde{\Gamma}_1| = \widetilde{\mathcal{V}} R_1 \widetilde{\mathcal{V}}^*$  and preserves  $\widetilde{\mathcal{V}} p$ . Now we extend  $\widetilde{\mathfrak{J}}_u$  to a conjugation  $\mathfrak{J}_u$  defining on the whole space  $\ell^2$ . This can be done by following the process stated in Remark 4.2.3 (we can set  $\widetilde{\mathfrak{J}}'_u$  to be an arbitrary partial conjugation with support Ker  $\Gamma$ , and then let  $\mathfrak{J}_u = \widetilde{\mathfrak{J}}_u \oplus \widetilde{\mathfrak{J}}'_u$ , thus we have  $\mathfrak{J}_u$  commutes with  $|\Gamma|, |\Gamma_1|$ , and this  $\mathfrak{J}_u$  also preserves  $\widetilde{\mathcal{V}}p$ ).

Define  $\tilde{\phi} := \tilde{\mathcal{V}} \varphi \tilde{\mathcal{V}}^*$ ,  $\tilde{\phi}_1 := \tilde{\mathcal{V}} \varphi_1 \tilde{\mathcal{V}}^*$ . Clearly  $\tilde{\phi}$  is a unitary operator commuting with  $|\tilde{\Gamma}|$ , and  $\tilde{\phi}_1$  is a partial isometry, Ker  $\tilde{\phi}_1 = \text{Ker }\tilde{\Gamma}_1$  commuting with  $|\Gamma_1|$ . Now we can rewrite equations (5.2.3), (5.2.4) as

$$\widetilde{\Gamma} = \mathfrak{C}(\widetilde{\mathcal{V}}R\widetilde{\mathcal{V}}^*)(\widetilde{\mathcal{V}}\varphi\widetilde{\mathcal{V}}^*)(\widetilde{\mathcal{V}}\mathfrak{J}_p\widetilde{\mathcal{V}}^*) = \mathfrak{C}|\widetilde{\Gamma}|\widetilde{\phi}\widetilde{\mathfrak{J}}_u,$$
$$\widetilde{\Gamma}_1 = \mathfrak{C}(\widetilde{\mathcal{V}}R_1\widetilde{\mathcal{V}}^*)(\widetilde{\mathcal{V}}\varphi_1\widetilde{\mathcal{V}}^*)(\widetilde{\mathcal{V}}\mathfrak{J}_p\widetilde{\mathcal{V}}^*) = \mathfrak{C}|\widetilde{\Gamma}|\widetilde{\phi}_1\widetilde{\mathfrak{J}}_u,$$

which are exactly identities (5.1.5) (for the particular choice of  $\tilde{\mathfrak{J}}_u, \tilde{\phi}, \tilde{\phi}_1$ ).

Finally, let us notice that (5.2.5) is just the identity  $u = \widetilde{\mathcal{V}}p$ .

#### 5.3 Proof of Proposition 5.2.3

The whole proof consists of three different parts: existence, uniqueness and the trivial kernel condition.

#### 5.3.1 Existence of Hankel Operator $\Gamma$

*Proof.* Let us first rewrite the equation  $R^2 - R_1^2 = pp^*$ . Since  $\varphi$  is unitary and commutes with R, we can write

$$R^2 = \varphi^* R^2 \varphi. \tag{5.3.1}$$

Similarly, since  $\varphi_1$  is a partial isometry commuting with  $R_1$ , Ker  $\varphi_1 = \text{Ker } R_1$ , we have

$$R_1^2 = (R_1\varphi_1)^*(R_1\varphi_1) = \varphi^* R \mathcal{T}^* \mathcal{T} R \varphi, \qquad (5.3.2)$$

where the last equality follows from the definition of  $\mathcal{T}$ . Denoting  $\hat{q} := \mathfrak{J}_p q$ , we have

$$p = \mathfrak{J}_p p = \mathfrak{J}_p R \varphi q = R \varphi^* \mathfrak{J}_p q = R \varphi^* \hat{q}.$$
(5.3.3)

Combining (5.3.1), (5.3.2) and (5.3.3), the equation  $R^2 - R_1^2 = pp^*$  can be translated to

$$\varphi^* R(I - \mathcal{T}^* \mathcal{T}) R \varphi = R \varphi^* \hat{q} \hat{q}^* R \varphi.$$

Since the operator  $R\varphi$  has trivial kernel, we conclude that

$$I - \mathcal{T}^* \mathcal{T} = (\hat{q})(\hat{q})^*.$$
 (5.3.4)

Applying both sides of (5.3.4) to x and taking inner product with x, we get

$$||x||^{2} - ||\mathcal{T}x||^{2} = |\langle x, \hat{q} \rangle|^{2}.$$

Replacing x in the above identity by  $\mathcal{T}x, (\mathcal{T})^2 x, ..., (\mathcal{T})^{n-1} x$ , and summing up all n equations, we see that

$$||x||^{2} - ||\mathcal{T}^{n}x||^{2} = \sum_{i=0}^{n-1} |\langle \hat{q}, \mathcal{T}^{i}x \rangle|^{2}.$$

Letting  $n \to \infty$  and using the asymptotic stability of  $\mathcal{T}$ , we conclude that

$$||x||^{2} = \sum_{n=0}^{\infty} |\langle \mathcal{T}^{n} x, \hat{q} \rangle|^{2}, \qquad (5.3.5)$$

which implies that the operator  $\mathcal{V}:H\to\ell^2$  defined by

$$\mathcal{V}x := \left( \langle x, \hat{q} \rangle, \langle \mathcal{T}x, \hat{q} \rangle, \langle \mathcal{T}^2 x, \hat{q} \rangle, \ldots \right) = \left( \langle \mathcal{T}^k x, \hat{q} \rangle \right)_{k=0}^{\infty}$$
(5.3.6)

is an isometry. The above identity (5.3.6) implies that

$$S^* \mathcal{V} = \mathcal{V} \mathcal{T}. \tag{5.3.7}$$

Denote  $\tilde{\mathcal{V}}$  as the operator  $\mathcal{V}$  with the restricted target space on Ran  $\mathcal{V}$ ; then the operator  $\tilde{\mathcal{V}}: \mathcal{H} \to \operatorname{Ran} \mathcal{V}$  is unitary. Trivially (5.3.6) implies that Ran  $\mathcal{V}$  is  $S^*$ -invariant, so we can define  $\tilde{S}^*: \operatorname{Ran} \mathcal{V} \to \operatorname{Ran} \mathcal{V}$  as  $\tilde{S}^*:= S^*|_{\operatorname{Ran} \mathcal{V}}$ . Denote by  $\tilde{S}$  the adjoint of  $\tilde{S}^*, \tilde{S}:=(\tilde{S}^*)^*=P_{\operatorname{Ran} \mathcal{V}}S|_{\operatorname{Ran} \mathcal{V}}$ . Then the identity (5.3.7) implies that

$$\widetilde{S}^* = \widetilde{\mathcal{V}}\mathcal{T}\widetilde{\mathcal{V}}^*$$
 and  $\widetilde{S} = \widetilde{\mathcal{V}}\mathcal{T}^*\widetilde{\mathcal{V}}^*.$  (5.3.8)

Now we define operator  $\Gamma$  and  $\Gamma_1$  as following

$$\Gamma := \mathfrak{CV}R\varphi\mathfrak{J}_p\mathcal{V}^* \qquad \Gamma_1 := \mathfrak{CV}R_1\varphi_1\mathfrak{J}_p\mathcal{V}^*.$$
(5.3.9)

We will show that  $\Gamma$  is a Hankel operator by proving that  $\Gamma S = \Gamma_1 = S^* \Gamma$ .

To show that  $\Gamma S = \Gamma_1$ , we recall that  $\mathfrak{T}^* = \varphi^* R^{-1} R_1 \varphi_1$  and that  $\mathfrak{T}^* \mathfrak{J}_p = \mathfrak{J}_p \mathcal{T}^*$ ; together with identity  $\mathcal{T}^* \mathcal{V}^* = \mathcal{V}^* S$  (which is just the adjoint of (5.3.7)) it gives us

$$\Gamma_{1} = \mathfrak{CV}R_{1}\varphi_{1}\mathfrak{J}_{p}\mathcal{V}^{*} = \mathfrak{CV}R\varphi\mathfrak{T}^{*}\mathfrak{J}_{p}\mathcal{V}^{*}$$
$$= \mathfrak{CV}R\varphi\mathfrak{J}_{p}\mathcal{T}^{*}\mathcal{V}^{*} = \mathfrak{CV}R\varphi\mathfrak{J}_{p}\mathcal{V}^{*}S$$
$$= \Gamma S.$$

And for the identity  $\Gamma_1 = S^*\Gamma$ , recalling that  $\mathcal{T} = R_1\varphi_1R^{-1}\varphi^*$  and using (5.3.7), we have

$$\mathfrak{C}\Gamma_1 = \mathcal{V}R_1\varphi_1\mathfrak{J}_p\mathcal{V}^* = \mathcal{V}\mathcal{T}R\varphi\mathfrak{J}_p\mathcal{V}^*$$
$$= S^*\mathcal{V}R\varphi\mathfrak{J}_p\mathcal{V}^* = S^*\mathfrak{C}\Gamma.$$

Since  $\mathfrak{C}$  commutes with  $S^*$ , we see that  $\mathfrak{C}\Gamma_1 = \mathfrak{C}S^*\Gamma$ , and left multiplying this identity by  $\mathfrak{C}$  we get the desired result.

Thus we know  $\Gamma$  is a Hankel operator. It remains to show that  $\Gamma$  satisfies (5.2.3), (5.2.4) and (5.2.5).

Using the definition of  $\Gamma$ ,  $\Gamma_1$  in (5.3.9), we know Ker  $\Gamma = (\operatorname{Ran} \mathcal{V})^{\perp}$ , hence we can write

$$\widetilde{\Gamma} := \Gamma|_{(\operatorname{Ker} \Gamma)^{\perp}} = \mathfrak{C} \mathcal{V} R \varphi \mathfrak{J}_p \widetilde{\mathcal{V}}^* = \mathfrak{C} \widetilde{\mathcal{V}} R \varphi \mathfrak{J}_p \widetilde{\mathcal{V}}^*, \qquad (5.3.10)$$

$$\widetilde{\Gamma}_1 := \Gamma_1|_{(\operatorname{Ker} \Gamma)^{\perp}} = \mathfrak{C}\widetilde{\mathcal{V}}R_1\varphi_1\mathfrak{J}_p\widetilde{\mathcal{V}}^*; \qquad (5.3.11)$$

Thus (5.2.3) and (5.2.4) holds.

For (5.2.5), we first derive the expression for  $\mathcal{V}^*$  and show that  $\mathcal{V}^* e_0 = \hat{q}$ . From the definition of  $\mathcal{V}$  in (5.3.6), for  $\forall x \in \mathcal{H}, \forall y \in \ell^2$ , we have

$$\begin{aligned} \langle \mathcal{V}x, y \rangle &= \sum_{k=0}^{\infty} \overline{y_k} \langle \mathcal{T}^k x, \hat{q} \rangle \\ &= \langle x, \sum_{k=0}^{\infty} y_k (\mathcal{T}^*)^k \hat{q} \rangle \end{aligned}$$

Hence  $\mathcal{V}^* y = \sum_{k=0}^{\infty} y_k (\mathcal{T}^*)^k \hat{q}$ , and we have  $\mathcal{V}^* e_0 = \mathfrak{S}^0 \hat{q} = \hat{q}$ . Together with  $\Gamma^* = \mathcal{V}\mathfrak{J}_p \varphi^* R \mathcal{V}^* \mathfrak{C}$  gained from (5.3.9), we have

$$\begin{split} \Gamma^* e_0 &= \mathcal{V} \mathfrak{J}_p \varphi^* R \mathcal{V}^* e_0 \\ &= \mathcal{V} \mathfrak{J}_p \varphi^* R \hat{q} \\ &= \mathcal{V} \mathfrak{J}_p p = \mathcal{V} p = \widetilde{\mathcal{V}} p. \end{split}$$

-	_	_	_	

#### 5.3.2 Uniqueness of Hankel Operator $\Gamma$

Proof. Now let's discuss the uniqueness by showing that the Hankel symbol  $\gamma_k = \langle \Gamma e_k, e_0 \rangle$  must have representation (5.2.6), which is independent of  $\mathcal{V}$ . Assuming that we are given Hankel operator  $\Gamma$  and  $\Gamma_1 = \Gamma S$  that satisfy (5.2.3), (5.2.4) and (5.2.5), we have already built up the unitary equivalence between the tuple  $\langle |\widetilde{\Gamma}|, |\widetilde{\Gamma}_1|, \widetilde{\mathfrak{J}}_u, \widetilde{\phi}, \widetilde{\phi}_1, u \rangle$  and  $\langle R, R_1, \mathfrak{J}_p, \varphi, \varphi_1, p \rangle$  in the proof of Proposition 5.2.4.

Now applying Proposition 5.1.4, we have  $\gamma_k = \langle (\mathfrak{S}^*)^k u, v \rangle$ , where  $\mathfrak{S}^* = |\widetilde{\Gamma}_1| \phi_1^* \phi |\widetilde{\Gamma}|^{-1}$ is unitary equivalent to  $\mathfrak{T} := \varphi_1^* R_1 R^{-1} \varphi$ , and the definition of v is also given in Proposition 5.1.4.

We first show that  $q = \widetilde{\mathcal{V}}^* v$ . Indeed from (5.1.8), we have

$$\widetilde{\mathcal{V}}p = u = |\widetilde{\Gamma}|\widetilde{\varphi}v = \widetilde{\mathcal{V}}R\varphi\widetilde{\mathcal{V}}^*v,$$

thus  $p = R\varphi \widetilde{\mathcal{V}}^* v$  and  $q = \widetilde{\mathcal{V}}^* v$ . Now from (5.1.6), we can write

^

$$\begin{aligned} \gamma_k &= \langle (\mathfrak{S}^*)^k u, v \rangle \\ &= \langle \widetilde{\mathcal{V}} \mathfrak{T}^k \widetilde{\mathcal{V}}^* \widetilde{\mathcal{V}} p, \widetilde{\mathcal{V}} q \rangle \\ &= \langle \mathfrak{T}^k p, q \rangle. \end{aligned}$$
(5.3.12)

Thus we get (5.2.6), and the coefficients  $\{\gamma_k\}_{k=0}^{\infty}$  only depends on the tuple  $\langle R, R_1, p, \varphi, \varphi_1 \rangle$ , which is independent of  $\widetilde{\mathcal{V}}$ . The uniqueness is done.

#### 5.3.3 The Trivial Kernel Condition of Hankel $\Gamma$

Proof. As discussed above,  $\operatorname{Ker} \Gamma = \{0\}$  if and only if  $\mathcal{V} : \mathcal{H} \to \ell^2$  defined by (5.3.6) satisfies  $\operatorname{Ran} \mathcal{V} = \ell^2$ , and this is equivalent to  $\mathcal{T}$  being unitary equivalent to  $S^*$ .

If Ker  $\Gamma = \{0\}$ , then  $\mathfrak{S}^*$  is unitarily equivalent to the backward shift  $S^*$ , and

comparing identities

$$I - SS^* = e_0 e_0^*, \qquad I - \mathcal{T}^* \mathcal{T} = (\mathfrak{J}_p q) (\mathfrak{J}_p q)^*$$

we conclude that  $||q|| = ||\mathfrak{J}_q q|| = 1$ . Now assuming  $q \in \operatorname{Ran} R$ , q = Rx, we will lead to a contradiction.

Take a vector  $f = \varphi_1 \varphi^* \mathfrak{J}_p x$ , thus  $x = \mathfrak{J}_p \varphi \varphi_1^* f$  and  $\mathfrak{J}_p q = R \varphi \varphi_1^* f$ , Hence

$$\mathcal{T}R\varphi\varphi_1^*f = \mathcal{T}(\mathfrak{J}_pq) = S^*e_0 = 0.$$

But on the other hand we have

$$\mathcal{T}R\varphi\varphi_1^* = R_1 = \varphi_1\varphi^*R\mathcal{T}^*,$$

where the last equality follows from  $R_1$  is a self-adjoint operator. Hence we have

$$\varphi_1 \phi^* R \mathcal{T}^* f = 0,$$

so  $\mathcal{T}^* f = 0$ , which contradicts to the fact that  $\mathcal{T}^*$  is an isometry. Hence  $q \notin \operatorname{Ran} R$ .

Now we prove the sufficiency part. Suppose ||q|| = 1 and  $q \notin \operatorname{Ran} R$ . We first show that  $\operatorname{Ker} R_1 = \{0\}$ .

Let  $R_1 x = 0$  for a  $x \neq 0$ . Applying to x the identity

$$R_1^2 = R\bigg(I - \langle \cdot, \varphi q \rangle \varphi q\bigg)R,$$

we get that

$$Rx = \langle Rx, \varphi q \rangle \varphi q;$$

note that  $Rx \neq 0$  because R has trivial kernel. This implies  $R\varphi^*x = \varphi^*Rx = \alpha q$ ,

 $\alpha = \langle Rx, \varphi q \rangle^{-1}$  which contradicts the assumption  $q \notin \operatorname{Ran} R$ .

Now from the definition of  $\mathfrak{S}^*$ :  $\varphi_1^* R_1 = \varphi^* R \mathcal{T}^*$ , we know Ker  $\mathcal{T}^* = \{0\}$  since Ker  $R_1 = \{0\}$ .

In addition, we apply  $\mathfrak{J}_p q$  on both sides of (5.3.4) we get  $\mathcal{T}^* \mathcal{T} \mathfrak{J}_p q = 0$ , together with Ker  $\mathcal{T}^* = \{0\}$ , we have  $\mathcal{T} \mathfrak{J}_p q = 0$ , hence by the definition of  $\mathfrak{T} = R_1 \varphi_1^* \varphi R^{-1}$  we also get  $\mathfrak{T} q = 0$ .

Now left and right multiplying (5.3.4) by  $\mathfrak{S}^*$  and  $\mathfrak{S}$  respectively, we get

$$\mathcal{T}\mathcal{T}^* - \mathcal{T}\mathcal{T}^*\mathcal{T}\mathcal{T}^* = \mathcal{T}\bigg(\langle\cdot, \mathfrak{J}_p q\rangle \mathfrak{J}_p q\bigg)\mathcal{T}^* = 0,$$

hence  $\mathcal{T}^*\mathcal{T}$  is a projection. Furthermore, since Ker  $\mathcal{T}^* = \{0\}$ , we have  $\mathcal{T}\mathcal{T}^* = I$  and  $\mathcal{T}^*$  is an isometry.

By Wold Decomposition Theorem [26, Theorem 1.1, p. 3], there exists an orthogonal decomposition for the whole Hilbert space:  $\mathcal{H} = \mathcal{H}_0 \oplus \mathcal{H}_1$ , such that  $\mathfrak{S}|_{\mathcal{H}_0}$  is unitary and  $\mathfrak{S}|_{\mathcal{H}_1}$  is a unilateral shift. Since  $\mathfrak{S}^*$  is asymptotically stable proved in subsection 5.4 which has no unitary part, we have  $\mathcal{H}_0 = \{0\}$  and thus there exists a wandering subspace  $\mathcal{L}$ , such that  $\mathcal{H} = \bigoplus_{n=0}^{\infty} (\mathfrak{S})^n \mathcal{L}$ .

To show that  $\mathcal{T}$  unitary equivalent to backward shift operator  $S^*$ , it's suffices to show that dim  $\mathcal{L} = 1$ .

Now apply vector  $\mathcal{T}^*x$  for an arbitrary x into (5.3.4), we get

$$\langle \mathcal{T}^* x, \mathfrak{J}_p q \rangle \mathfrak{J}_p q = \mathcal{T}^* x - \mathcal{T}^* (\mathcal{T} \mathcal{T}^*) x = \mathcal{T}^* x - \mathcal{T}^* x = 0,$$

hence  $\mathfrak{J}_p q \perp \operatorname{Ran} \mathcal{T}^*$ , and  $\mathcal{L}$  is the space spanned by  $\mathfrak{J}_p q$ , which is of dimension 1. This implies  $\mathcal{T}$  is unitarily equivalent to  $S^*$ , and  $\operatorname{Ker} \Gamma = \{0\}$ .

#### Chapter 6

## EIGENSPACE STRUCTURE OF COMPACT RANK-ONE PERTURBATION

In chapter 2 and chapter 4, we have discussed the inverse spectral problem for self-adjoint Hankel operators and non self-adjoint Hankel operators as  $\mathfrak{C}$ -symmetric operators respectively. Starting from this chapter, we will mainly focus on the category of compact Hankel operators.

Recall that for a general Hankel operator  $\Gamma$ ,  $\Gamma$  and  $\Gamma_1 := \Gamma S$  satisfy a rank-one perturbation relation:  $|\Gamma|^2 - |\Gamma_1|^2 = uu^*$ . And also recall that in section 5.1, we construct a triple  $(|\widetilde{\Gamma}|, |\widetilde{\Gamma}_1|, u, \widetilde{\phi}, \widetilde{\phi}_1, \widetilde{\mathfrak{J}}_u)$ . Thus in this chapter, we start with a triple  $(R, R_1, p, \varphi, \varphi_1, \mathfrak{J}_p)$  with similar properties:

- (i)  $R, R_1$  are two positive, self-adjoint compact operator defined on a Hilbert space  $\mathcal{H}$ . In addition we have Ker  $R = \{0\}$ ;
- (ii)  $R^2 R_1^2 = pp^*$  for a vector p with  $||R^{-1}p|| \le 1$ ;
- (iii)  $\mathfrak{J}_p$  is a conjugation commutes with  $R, R_1$  and preserves p.
- (iv)  $\varphi$  is a  $\mathfrak{J}_p$ -symmetric unitary operator, which commutes with R;
- (v)  $\varphi_1$  is a  $\mathfrak{J}_p$ -symmetric partial isometry with Ker  $\varphi_1 = \text{Ker } R_1$ , which commutes with  $R_1$ . In addition, we have  $\varphi_1|_{(\text{Ker } R_1)^{\perp}}$  is unitary (See Remark 5.2.1);

(vi) The contraction  $\mathcal{T}$  defined as  $\mathcal{T} := \varphi_1 R_1 R^{-1} \varphi^*$  is asymptotically stable.

In this chapter, we will mainly focus on the eigenspaces structure of  $R, R_1$ . Denoting  $\mathcal{H}_0 = \overline{\text{Span}}\{R^n p | n \ge 0\}$ , we start this chapter by stating an equivalent condition of  $\mathcal{H}_0 = \mathcal{H}$ , i.e., the cyclicity of p (see Lemma 6.1.1). Then in section 6.2, we first analyze the eigenspace structure of  $R, R_1$  on  $\mathcal{H}_0$ , and then we derive the structure of eigenspaces of  $R, R_1$  on the whole space  $\mathcal{H}$ .

Finally, we close this chapter by generating a canonical choice of  $\varphi, \varphi_1$ , which is given in Lemma 6.3.2. For the canonical choice of  $\varphi, \varphi_1$ , we can further analyze the behavior of  $\varphi, \varphi_1$  acting on each eigenspaces of  $R, R_1$  (see Proposition 7.6.3). We can also show that the conjugation  $\mathfrak{J}_p$  will be uniquely determined if taking the canonical choice of  $\varphi.\varphi_1$  (see Remark 7.6.6)

#### 6.1 Setting and preparation

We start this section with a tuple  $(R, R_1, p, \varphi, \varphi_1, \mathfrak{J}_p)$  which satisfies (i), (ii), (iii), (iv), (v), (vi) stated at the beginning of this chapter. Denoting  $\mathcal{H}_0 := \overline{\text{Span}}\{R^n p | n \ge 0\}$ , we first derive an equivalent condition of  $\mathcal{H}_0 = \mathcal{H}$ , i.e., the cyclicity of vector p.

**Lemma 6.1.1.** With the assumptions above, p is cyclic with respect to R if and only if

(i) For every  $\lambda \in \sigma(R)$ , we have dim Ker $(R - \lambda I) \leq 1$ ;

(ii)  $\operatorname{Proj}_{\operatorname{Ker}(R-\lambda I)} p \neq 0.$ 

*Proof.* If we have the cyclicity of vector p, we will prove (i) and (ii) by contradiction.

If dim Ker $(R - \lambda I) \geq 2$  for a certain  $\lambda$ , then  $H_0 \cap \text{Ker}(R - \lambda I)$  have at most dimension 1 spanned by  $\text{Proj}_{\text{Ker}(R-\lambda I)} p$  (will be a trivial space if  $\text{Proj}_{\text{Ker}(R-\lambda I)} p = 0$ ), thus p can't be cyclic with respect to R.

For (ii), if  $\operatorname{Proj}_{\operatorname{Ker}(R-\lambda I)} p = 0$ , then  $\mathcal{H}_0 \cap \operatorname{Ker}(R-\lambda I) = \emptyset$ , thus p can't be cyclic.

For the other direction, under (i) and (ii), we denote the spectrum of R as

$$\lambda_1 > \lambda_2 > \dots > \lambda_n \to 0,$$

and we write  $p = \sum_{k=1}^{\infty} p_k$  where  $p_k \in \text{Ker}(R - \lambda_k I)$ . (The case when R is of finiterank, i.e. finitely many eigenvalues, is trivially true, so we only consider case when dim Ran  $R = \infty$ )

We can show that

$$\lim_{n \to \infty} \left(\frac{1}{2\lambda_1}\right)^n (R + \lambda_1 I)^n p \rightharpoonup p_1.$$

Since  $\mathcal{H}_0$  is weakly closed, thus we have  $p_1 \in \mathcal{H}_0$ . In fact we have

$$\left(\frac{1}{2\lambda_1}\right)^n (R+\lambda_1 I)^n p = \sum_{k=0}^\infty \left(\frac{1}{2} + \frac{\lambda_k}{2\lambda_1}\right)^n p_k,$$

thus

$$\lim_{n \to \infty} \langle \left(\frac{1}{2\lambda_1}\right)^n (R + \lambda_1 I)^n p, p_i \rangle = \begin{cases} \|p_1\|^2 & \text{if } i = 1\\ 0 & \text{else} \end{cases}$$

so we have  $p_1 \in \mathcal{H}_0$ .

Similarly we can also show that

$$\lim_{n \to \infty} \left(\frac{1}{2\lambda_2}\right)^n \left\{ (R + \lambda_2 I)^n p - (\lambda_1 + \lambda_2)^n p_1 \right\} \rightharpoonup p_2,$$

and  $p_2 \in \mathcal{H}_0$ . Thus followed by the process of induction, if we have  $p_1, p_2, ..., p_k \in \mathcal{H}_0$ , then since

$$\lim_{n \to \infty} \left(\frac{1}{2\lambda_{k+1}}\right)^n \left\{ (R + \lambda_{k+1}I)^n p - \sum_{i=1}^k (\lambda_i + \lambda_{k+1})^n p_i \right\} \rightharpoonup p_{k+1},$$

then we will have  $p_{k+1} \in \mathcal{H}_0$ . Hence  $p_k \in \mathcal{H}_0$  holds for all k and  $\mathcal{H} = \overline{\text{Span}}\{p_k | k \ge 1\} \subset \mathcal{H}_0$ , p is cyclic.

Remark 6.1.2. In fact, Lemma 6.1.1 is also true for a general compact operator without given the trivial kernel condition. The statement can be modified as follow:

Proposition 6.1.3. Let A be a compact operator on a Hilbert space  $\mathcal{H}$  with simple eigenvalues, then there exists a vector  $x \in A$ , such that x is a cyclic vector for A in  $\mathcal{H}$ .

In fact, if we write all eigenvalues  $\{\lambda_k\}_{k=1}^{\infty}$  in the non-increasing order:

$$|\lambda_1| \ge |\lambda_2| \ge \dots,$$

then we can take x to be

$$x = e_0 + \sum_{k=1}^{\infty} \frac{1}{k^2} e_k,$$

here  $e_k$  is a unit vector in Ker $(A - \lambda_k I)$ , and  $e_0$  is a unit vector in Ker A if  $0 \in \sigma_p(A)$ , otherwise we take  $e_0 = 0$ . The proof is very much similar to the proof of Lemma 6.1.1.

In this section we discuss the case when operator R is not cyclic (that is, there doesn't exist any vector p such that p is cyclic with respect to R in  $\mathcal{H}$ .). Then according to Lemma 6.1.1 R have some non-simple singular values.

### 6.2 Non-simple Eigenvalues for Operators of Rank-one Perturbation

We begin this part with the following simple lemma.

**Lemma 6.2.1.** For  $\forall \lambda \in \mathbb{R}^+ \cup \{0\}$ , we have

$$\left|\dim \operatorname{Ker}(R - \lambda I) - \dim \operatorname{Ker}(R_1 - \lambda I)\right| \leq 1.$$

Proof. For all vector  $x \in \text{Ker}(R - \lambda I) \cap p^{\perp}$ , we have  $x \in \text{Ker}(R_1^2 - \lambda^2 I)$ , thus  $x \in \text{Ker}(R_1 - \lambda I)$  as  $R_1$  is positive. Thus  $\text{Ker}(R - \lambda I) \cap p^{\perp} \subseteq \text{Ker}(R_1 - \lambda I)$ , and we have

$$\dim \operatorname{Ker}(R - \lambda I) \leq \dim \operatorname{Ker}(R_1 - \lambda I) + 1.$$

Similarly,  $\operatorname{Ker}(R_1 - \lambda I) \cap p^{\perp} \subseteq \operatorname{Ker}(R - \lambda I)$  and we have

$$\dim \operatorname{Ker}(R_1 - \lambda I) \le \dim \operatorname{Ker}(R - \lambda I) + 1.$$

So the dimension of two kernel space at most differ by 1.

From Lemma 6.2.1, it's easily seen that dim Ker  $R_1 \leq 1$ . Now we first restrict  $R^2 - R_1^2 = pp^*$  on  $\mathcal{H}_0$ :

$$R^2|_{\mathcal{H}_0} - R_1^2|_{\mathcal{H}_0} = pp^*.$$

Since p is cyclic with respect to  $\mathcal{H}_0$ , thus  $R|_{\mathcal{H}_0}$  have simple singular values. For the special case, if  $R|_{\mathcal{H}_0}$  has finite rank, then  $R_1$  is also finite rank. For the following discussion, we assume that dim  $\operatorname{Ran} R|_{\mathcal{H}_0} = \infty$ .

In the next section, we will show the following proposition, which gives the eigenspace structure for  $(R, R_1)$  restricted on  $\mathcal{H}_0$ .

**Proposition 6.2.2.** There exists an intertwining sequence

$$\lambda_1 > \mu_1 > \lambda_2 > \mu_2 > \dots \to 0,$$
 (6.2.1)

such that the non-zero eigenvalues of  $R|_{\mathcal{H}_0}$ ,  $R_1|_{\mathcal{H}_0}$  are simple, and coincide with  $\{\lambda_k\}_{k=1}^{\infty}$ .  $\{\mu_k\}_{k=1}^{\infty}$  respectively.

#### 6.2.1 Eigenspace Structure on $\mathcal{H}_0$

We show Proposition 6.2.2 in the following steps.

**Proposition 6.2.3.** The dimension of any eigenspace of  $R_1|_{\mathcal{H}_0}$  is no more than 1, i.e.  $R_1|_{\mathcal{H}_0}$  has simple singular values.

*Proof.* For any  $\lambda$ , if dim Ker $(R_1 - \lambda I) \ge 3$ , since we have

$$\operatorname{Ker}(R_1 - \lambda I) \cap p^{\perp} \subseteq \operatorname{Ker}(R - \lambda I), \tag{6.2.2}$$

this contradicts to the fact that R has simple singular values on  $\mathcal{H}_0$ .

If dim Ker $(R_1 - \lambda I) = 2$  for some  $\lambda$ , then by (6.2.2), we can take a non-zero vector x such that

$$x \in \operatorname{Ker}(R - \lambda I) \cap \operatorname{Ker}(R_1 - \lambda I) \cap \mathcal{H}_0,$$

then applying this x to  $R^2 - R_1^2 = pp^*$  we get  $\langle x, p \rangle = 0$ , which contradicts to the property that  $P_{\text{Ker}(R-\lambda I)}p \neq 0$  given in Lemma 6.1.1.

**Proposition 6.2.4.** Denoting the non-zero eigenvalues of  $R, R_1$  as  $\{\lambda_k\}_{k=1}^{\infty}, \{\mu_k\}_{k=1}^{\infty}$ respectively with

$$\lambda_1 > \lambda_2 > \dots \to 0, \qquad \mu_1 > \mu_2 > \dots \to 0$$

then we have

$$\lambda_k \ge \mu_k, \qquad \mu_k \ge \lambda_{k+1}$$

holds for all k.

*Proof.* Here we use the mini-max definition for singular values. That is,

$$\lambda_k^2 = \min_{subspace \mathcal{F} \subseteq \mathcal{H}_0, codim \mathcal{F} = k-1} \max_{x \in \mathcal{F}, \|x\| = 1} \langle R^2 x, x \rangle$$
(6.2.3)

$$\mu_k^2 = \min_{subspace \mathcal{G} \subseteq \mathcal{H}_0, codim \mathcal{G} = k-1} \max_{x \in \mathcal{G}, \|x\|=1} \langle R_1^2 x, x \rangle$$
(6.2.4)
To show  $\lambda_k \geq \mu_k$ , we take a subspace  $\mathcal{F} \subseteq \mathcal{H}_0$  with codim = k - 1 where (6.2.3) reaches its equality. Then we have

$$\mu_k^2 \leq \max_{x \in \mathcal{F}, \|x\| = 1} \langle R_1^2 x, x \rangle \leq \max_{x \in \mathcal{F}, \|x\| = 1} \langle R^2 x, x \rangle = \lambda_k^2$$

To show that  $\mu_k \geq \lambda_{k+1}$ , we take a subspace  $\mathcal{G} \subseteq \mathcal{H}_0$  with codim = k - 1 where (6.2.4) reaches its equality. Then denoting  $\mathcal{F}_1 := \mathcal{G} \cap p^{\perp}$ , then  $\mathcal{F}_1$  has codim no more than k where R coincides with  $R_1$  on  $\mathcal{F}_1$ . Thus we have

$$\mu_k^2 = \max_{x \in \mathcal{G}, \|x\|=1} \langle R_1^2 x, x \rangle \ge \max_{x \in \mathcal{F}_1, \|x\|=1} \langle R_1^2 x, x \rangle = \max_{x \in \mathcal{F}_1, \|x\|=1} \langle R^2 x, x \rangle \ge \lambda_{k+1}^2$$

**Proposition 6.2.5.** (6.2.1) holds for  $\{\lambda_k\}_{k=1}^{\infty}$ ,  $\{\mu_k\}_{k=1}^{\infty}$  defined in Proposition 6.2.4. Equivalently saying, we have

$$\lambda_k > \mu_k, \qquad \mu_k > \lambda_{k+1}$$

*Proof.* We prove the two inequalities by contradiction.

(i) If  $\lambda_k = \mu_k$ , we take a subspace  $\mathcal{F}$  as

$$\mathcal{F} = \bigoplus_{i=k}^{\infty} \big( \operatorname{Ker}(R - \lambda_i I) \cap \mathcal{H}_0 \big),$$

then  $codim\mathcal{F} = k - 1$  in  $\mathcal{H}_0$  and  $\max_{x \in \mathcal{F}, ||x|| = 1} \langle R^2 x, x \rangle = \lambda_k^2$ . The maximum can only be reached when  $x \in \operatorname{Ker}(R - \lambda_k I)$ .

Now for any vector  $x \in \mathcal{F}$ , we have

$$\langle R_1^2 x, x \rangle = \langle R^2 x, x \rangle - |\langle x, p \rangle|^2 \le \lambda_k^2 ||x||^2 - |\langle x, p \rangle|^2 \le \lambda_k^2 ||x||^2.$$
(6.2.5)

Thus if we have  $\mu_k = \lambda_k$ , then according to (6.2.5), there exists a vector  $x \in \mathcal{F}$ , such

that  $||Rx|| = \lambda_k ||x||$  and  $x \perp p$ . This contradicts to the fact that  $\operatorname{Proj}_{\operatorname{Ker}(R-\lambda_k I)\cap\mathcal{H}_0} p \neq 0$ , which is given in Lemma 6.1.1

(ii) If  $\mu_k = \lambda_{k+1}$ , we take a subspace  $\mathcal{G} \subseteq \mathcal{H}_0$  as

$$\mathcal{G} = \operatorname{Ker} R_1 \bigoplus_{i=k}^{\infty} \left( \operatorname{Ker}(R_1 - \mu_i I) \cap \mathcal{H}_0 \right),$$

then we have  $\max_{x \in \mathcal{G}, \|x\|=1} \langle R_1^2 x, x \rangle = \mu_k^2$ , and the maximum can only be achieved when  $x \in \operatorname{Ker}(R_1 - \mu_k)$ .

Denote again  $\mathcal{F}_1 := \mathcal{G} \cap p^{\perp}$ , since

$$\mu_k^2 = \max_{x \in \mathcal{G}, \|x\|=1} \langle R_1^2 x, x \rangle \ge \max_{x \in \mathcal{F}_1, \|x\|=1} \langle R_1^2 x, x \rangle = \max_{x \in \mathcal{F}_1, \|x\|=1} \langle R^2 x, x \rangle \ge \lambda_{k+1}^2, \quad (6.2.6)$$

where here the maximum of first equality in (6.2.6) holds when  $x \in \text{Ker}(R_1 - \mu_k) \cap \mathcal{H}_0$ , and the first inequality inside (6.2.6) is reached when this  $x \perp p$ . Thus if we have  $\mu_k = \lambda_{k+1}$ , we have  $p \perp \text{Ker}(R_1 - \mu_k I)$ 

Now take a vector  $y \in \text{Ker}(R_1 - \mu_k I)$ , since  $y \perp p$ , we have  $Ry = R_1 y = \mu_k y$ , and  $\mu_k$  is a simple singular value for both  $R|_{\mathcal{H}_0}$  and  $R_1|_{\mathcal{H}_0}$ . Together with the property that  $R = R_1$  on  $\mathcal{H}_0^{\perp}$ , we have

$$\operatorname{Ker}(R - \mu_k I) = \operatorname{Ker}(R_1 - \mu_k I)$$
 on  $\mathcal{H}$ .

Now apply a  $x \in \text{Ker}(R - \mu_k I)$  to the equation  $R_1\varphi_1 = \mathcal{T}R\varphi$ , we have  $\varphi_1 x = \mathcal{T}\varphi x$ . Since  $\text{Ker}(R - \mu_k I) = \text{Ker}(R_1 - \mu_k I)$  is a reducing subspace for  $R, R_1, \varphi, \varphi_1$ , thus  $\mathcal{T}$  maps unitarily on  $\text{Ker}(R - \mu_k I)$ , which leads to a contradiction to the asymptotic stability of  $\mathcal{T}$ .

*Remark* 6.2.6. For the special case when  $\mathcal{H}_0$  has finite dimension, we have the following result:

Proposition 6.2.7. If dim  $\mathcal{H}_0 < \infty$ , there exists an intertwining sequence

$$\lambda_1 > \mu_1 > \dots > \lambda_N > \mu_N \ge 0$$

such that  $R|_{\mathcal{H}_0}, R_1|_{\mathcal{H}_0}$  have simple eigenvalues, and their eigenvalues coincides with  $\{\lambda_k\}_{k=1}^N$ .  $\{\mu_k\}_{k=1}^N$  respectively.

#### 6.2.2 Eigenspace Structure on $\mathcal{H}$

With the result in Proposition 6.2.2, we are ready to deep into the eigenspace structure of  $(R, R_1)$  on  $\mathcal{H}$ . We derive the structure of eigenspaces of  $R, R_1$  by the following steps.

**Lemma 6.2.8.** For a non-zero  $\tau \neq \lambda_k, \mu_k$  for all k, we have

$$\operatorname{Ker}(R - \tau I) = \operatorname{Ker}(R_1 - \tau I) = 0$$

Proof. We first show that  $\operatorname{Ker}(R-\tau I) = \operatorname{Ker}(R_1-\tau I)$ . For all  $x \in \operatorname{Ker}(R-\tau I)$ , we have  $x \perp \operatorname{Ker}(R - \lambda_k I)$  for all k, thus  $x \perp H_0$ .  $R = R_1$  on  $\mathcal{H}_0^{\perp}$ , we have  $x \in \operatorname{Ker}(R_1 - \tau I)$ . Thus  $\operatorname{Ker}(R - \tau I) \subseteq \operatorname{Ker}(R_1 - \tau I)$ . Similarly for all  $x \in \operatorname{Ker}(R_1 - \tau I)$ , we have  $x \perp \mathcal{H}_0$  and  $x \in \operatorname{Ker}(R - \tau I)$ , thus

$$\operatorname{Ker}(R - \tau I) = \operatorname{Ker}(R_1 - \tau I).$$

Now we show that  $\mathcal{T}$  can't be asymptotically stable if the kernel space is non-trivial. If  $\operatorname{Ker}(R - \tau x) \neq \emptyset$ , then apply  $x \in \operatorname{Ker}(R - \tau I)$  into equation  $\mathcal{T}\varphi R = \varphi_1 R_1$ , we get  $\mathcal{T}\varphi x = \varphi_1 x$ . Since  $\varphi$  commutes with R, and  $\varphi_1$  commutes  $R_1$ , thus we have  $\operatorname{Ker}(R - \tau I)$  is a reducing subspace with respect to  $R, R_1, \varphi, \varphi_1$ , and  $\mathcal{T}$  maps unitarily on  $\operatorname{Ker}(R - \tau I)$ , which contradicts to the asymptotic stability of  $\mathcal{T}$ . Now we know that  $R, R_1$  has no eigenvalues other than  $\{\lambda_k\}_{k=1}^{\infty}, \{\mu_k\}_{k=1}^{\infty}$  and 0. We denote

$$E_{\lambda_k} := \operatorname{Ker}(R - \lambda_k I), \qquad E_{\mu_k} := \operatorname{Ker}(R - \mu_k I); \tag{6.2.7}$$

$$E_{\lambda_k}^1 := \operatorname{Ker}(R_1 - \lambda_k I), \qquad E_{\mu_k}^1 := \operatorname{Ker}(R_1 - \mu_k I).$$
 (6.2.8)

Note that all of those eigenspaces have finite dimension since  $R, R_1$  are compact. In addition, Lemma 6.2.1 implies that the dimension between  $E_{\lambda_k}$  and  $E_{\lambda_k}^1$ ,  $E_{\mu_k}$  and  $E_{\mu_k}^1$  at most differs by 1.

**Proposition 6.2.9.**  $E_{\lambda_k}, E_{\lambda_k}^1, E_{\mu_k}, E_{\mu_k}^1$  satisfy the following properties

- (i)  $E_{\lambda_k}^1 \subseteq \mathcal{H}_0^{\perp}, \ E_{\mu_k} \subseteq \mathcal{H}_0^{\perp};$
- (ii) Denote

$$p_k := P_{\mathcal{H}_0 \cap \operatorname{Ker}(R - \lambda_k I)} p, \qquad p_k^1 := P_{\mathcal{H}_0 \cap \operatorname{Ker}(R_1 - \mu_k I)} p, \tag{6.2.9}$$

then we have

$$E_{\lambda_k} = E_{\lambda_k}^1 \oplus \operatorname{Span}\{p_k\}, \qquad E_{\mu_k}^1 = E_{\mu_k} \oplus \operatorname{Span}\{p_k^1\}.$$

*Proof.* For (i), since we have

$$\mathcal{H}_0 = \bigoplus_{k=1}^{\infty} \left( \operatorname{Ker}(R - \lambda_k I) \cap \mathcal{H}_0 \right) = \bigoplus_{k=1}^{\infty} \left( \mathcal{H}_0 \cap \operatorname{Ker}(R_1 - \mu_k I) \right) \bigoplus \left( \mathcal{H}_0 \cap \operatorname{Ker}(R_1) \right),$$

and we have  $\operatorname{Ker}(R_1 - \lambda_k I) \perp \operatorname{Ker}(R_1 - \mu_j I)$  for all j and  $\operatorname{Ker}(R_1 - \lambda_k I) \perp \operatorname{Ker}(R_1, k)$ thus  $\operatorname{Ker}(R_1 - \lambda_k I) \perp \mathcal{H}_0$ . Similarly we also have  $\operatorname{Ker}(R - \mu_k) \perp \mathcal{H}_0$  for all k.

Now for (ii), since  $R = R_1$  on  $H_0^{\perp}$ , which implies  $R = R_1$  on  $E_{\lambda_k}^1$  and  $E_{\mu_k}$  for all k. Thus  $E_{\lambda_k}' \subseteq E_{\lambda_k}, E_{\mu_k} \subseteq E_{\mu_k}'$ .

On the other hand, we also know that  $p_k \in E_{\lambda_k}$  and  $p_k^1 \in E_{\mu_k}^1$  (easy to see that

 $p_k, p_k^1 \neq 0$  because p is cyclic with respect to R and  $R_1$  on  $\mathcal{H}_0$ ). Hence

$$\operatorname{Span}\{p_k\} \oplus E^1_{\lambda_k} \subseteq E_{\lambda_k}, \qquad \operatorname{Span}\{p_k^1\} \oplus E_{\mu_k} \subseteq E^1_{\mu_k}, \qquad (6.2.10)$$

then applying Lemma 6.2.1 to (6.2.10), and we finish the proof of (ii).

So far, we can describe the complete structure of all eigenspaces of  $R, R_1$ .

**Proposition 6.2.10.** Let  $(R, R_1, p, \varphi, \varphi_1, \mathfrak{J}_p)$  be a tuple satisfying the setting in subsection 6.1, and we again denote  $\mathcal{H}_0 = \overline{\text{Span}}\{R^n p : n \ge 0\}$ . If R is not finite rank, then there exists an intertwining sequences

$$\lambda_1 > \mu_1 > \lambda_2 > \mu_2 > \dots > \to 0,$$

such that

(i)  $R|_{\mathcal{H}_0}, R_1|_{\mathcal{H}_0}$  have simple eigenvalues  $\{\lambda_k\}_{k=1}^{\infty}, \{\mu_k\}_{k=1}^{\infty}$  respectively;

(ii)

$$\mathcal{H}_{0}^{\perp} = \left(\bigoplus_{k=1}^{\infty} E_{\lambda_{k}}^{1}\right) \bigoplus \left(\bigoplus_{k=1}^{\infty} E_{\mu_{k}}\right)$$
(6.2.11)

(iii)

$$E_{\lambda_k} = E_{\lambda_k}^1 \oplus \operatorname{Span}\{p_k\}, \qquad E_{\mu_k}^1 = E_{\mu_k} \oplus \operatorname{Span}\{p_k^1\},$$

where  $p_k, p_k^1$  are defined in (6.2.9).

For the special case when  $R, R_1$  are finite rank operators, proposition 6.2.10 will be modified as follow:

**Proposition 6.2.11.** Let  $(R, R_1, p, \varphi, \varphi_1, \mathfrak{J}_p)$  be a tuple satisfying the setting in subsection 6.1. In addition, R is a finite rank operator. Then there exists an intertwining sequence

$$\lambda_1 > \mu_1 > \dots > \lambda_N > \mu_N \ge 0,$$

such that

(i) 
$$R|_{\mathcal{H}_0}, R_1|_{\mathcal{H}_0}$$
 have simple eigenvalues  $\{\lambda_k\}_{k=1}^N, \{\mu_k\}_{k=1}^N$  respectively;  
(ii)  $\mathcal{H}_0^{\perp} = \left(\bigoplus_{k=1}^N E_{\lambda_k}^1\right) \bigoplus \left(\bigoplus_{k=1}^N E_{\mu_k}\right);$   
(iii)

$$E_{\lambda_k} = E_{\lambda_k}^1 \oplus \operatorname{Span}\{p_k\}, \qquad E_{\mu_k}^1 = E_{\mu_k} \oplus \operatorname{Span}\{p_k^1\},$$

where  $p_k, p_k^1$  are defined in (6.2.9).

For the special case when  $\mu_N = 0$ , we have dim  $E^1_{\mu_N} = 1$ , and  $E_{\mu_N}$  is trivial.

*Remark* 6.2.12. Since we have the decomposition of the whole space

$$\mathcal{H} = \mathcal{H}_0 \bigoplus \left( \bigoplus_{k=1}^N E_{\lambda_k}^1 \right) \bigoplus \left( \bigoplus_{k=1}^N E_{\mu_k} \right)$$

implied by (6.2.11), we can also analyze the structure of  $\varphi, \varphi_1$  on  $\mathcal{H}$ . In fact, since  $\varphi, \varphi_1$  commutes with  $R, R_1$  respectively, we have

- (i)  $\varphi$  acts unitarily on  $E_{\lambda_k}, E_{\mu_k}$  for all k.
- (ii) In Remark 5.2.1, we already know  $\varphi_1|_{(\operatorname{Ker} R_1)^{\perp}}$  is unitary. Since  $E_{\lambda_k}^1, E_{\mu_k}^1 \subseteq (\operatorname{Ker} R_1)^{\perp}$  are all invariant subspaces for  $\varphi_1$ . Thus we have  $\varphi_1$  acts unitarily on  $E_{\lambda_k}^1, E_{\mu_k}^1$ . Another way to see this is that  $\varphi_1|_{E_{\lambda_k}^1}, \varphi_1|_{E_{\mu_k}^1}$  are both finite-rank isometry, thus they are onto.

## **6.3** Canonical Choice for $\varphi$ and $\varphi_1$

In this section, we generate a canonical choice for  $\varphi$ ,  $\varphi_1$  and  $\mathfrak{J}_p$  on  $\mathcal{H}$ . As we have discussed in Remark 5.1.3, the choice of  $\mathfrak{J}_u$  is not unique and we say two triples  $(\tilde{\mathfrak{J}}_u, \tilde{\phi}, \tilde{\phi}_1)$  and  $(\tilde{\mathfrak{J}}'_u, \tilde{\phi}', \tilde{\phi}'_1)$  are in the same equivalent class iff there exists a  $\mathfrak{J}_u$ -symmetric unitary operator  $\psi$  such that

$$\widetilde{\mathfrak{J}}'_u = \psi \widetilde{\mathfrak{J}}_u, \qquad \widetilde{\phi}' = \widetilde{\phi} \psi^*, \qquad \widetilde{\phi}'_1 = \widetilde{\phi}_1 \psi^*.$$

As we have stated in Proposition 5.2.4, we have built up unitary equivalence between tuples  $(\mathfrak{J}_p, \varphi, \varphi_1)$  and  $(\tilde{\mathfrak{J}}_u, \tilde{\phi}, \tilde{\phi}_1)$ . Now we define a similar equivalence class relation for the tuple  $(\mathfrak{J}_p, \varphi, \varphi_1)$ .

**Definition 6.3.1.** We say two tuples  $(\mathfrak{J}_p, \varphi, \varphi_1)$  and  $(\mathfrak{J}'_p, \varphi', \varphi'_1)$  are in the same equivalent class if and only if there exists and  $\mathfrak{J}_p$ -symmetric unitary operator  $\psi$ , such that

$$\mathfrak{J}'_p = \psi \mathfrak{J}_p, \qquad \varphi' = \varphi \psi^*, \qquad \varphi'_1 = \varphi_1 \psi^*.$$

Easy to check that the two tuples  $(\mathfrak{J}_p, \varphi, \varphi_1)$  and  $(\mathfrak{J}'_p, \varphi', \varphi'_1)$  from the same equivalent class define the same Hankel  $\Gamma$  in Proposition 5.2.3 and the same contraction  $\mathfrak{S}^* := \varphi_1 R_1 R^{-1} \varphi^*$ .

The following lemma gives the canonical choice of  $\varphi, \varphi_1$  from a given equivalent class.

**Lemma 6.3.2.** In each equivalent class of  $(\mathfrak{J}_p, \varphi, \varphi_1)$ , there exists a unique triple  $(\mathfrak{J}'_p, \varphi', \varphi'_1)$  such that

$$\varphi_1'|_{E_{\lambda_k}^1} = I, \qquad \varphi'|_{E_{\mu_k}} = I.$$

*Proof.* As we have discussed in subsection 6.2,  $\mathcal{H}_0^{\perp}$  can be decomposed as

$$\mathcal{H}_0^{\perp} = \left(\bigoplus_{k=1}^{\infty} E_{\lambda_k}^1\right) \bigoplus \left(\bigoplus_{k=1}^{\infty} E_{\mu_k}\right).$$

Now we define an operator  $\psi$ , such that

- (i)  $\psi = \varphi$  on  $E_{\mu_k}$  for all  $k \in \mathbb{N}$ ;
- (ii)  $\psi = \varphi_1$  on  $E^1_{\lambda_k}$  for all  $k \in \mathbb{N}$ ;
- (iii)  $\psi = I$  on  $\mathcal{H}_0$ .

Since we have assumed that  $\varphi$ ,  $\varphi_1$  are  $\mathfrak{J}_p$ -symmetric in subsection 6.1, and  $E^1_{\lambda_k}$ ,  $E_{\mu_k}$ are all reducing subspaces for  $\mathfrak{J}_p$ , we know  $\psi$  is  $\mathfrak{J}_p$ -symmetric. In addition, from Remark 6.2.12, we know that  $\varphi$ ,  $\varphi_1$  acts unitarily on  $E_{\mu_k}$ ,  $E^1_{\lambda_k}$  respectively, hence  $\psi$  is unitary, and the tuple  $(\mathfrak{J}'_p, \varphi', \varphi'_1)$  defined as

$$\mathfrak{J}'_p := \psi \mathfrak{J}_p, \qquad \varphi' := \varphi \psi^*, \qquad \varphi'_1 := \varphi_1 \psi^*$$

satisfy

$$\left. \varphi' \right|_{E_{\mu_k}} = I, \qquad \left. \varphi'_1 \right|_{E^1_{\lambda_k}} = I.$$

Here the uniqueness of such triple follows from the uniqueness of  $\psi$ .

We say the triple  $(\mathfrak{J}_p, \varphi, \varphi_1)$  which satisfies

$$\varphi_1|_{E^1_{\lambda_k}} = I, \qquad \varphi|_{E_{\mu_k}} = I \qquad \text{hold for all } \mathbf{k} \in \mathbb{N}$$

to be the canonical choice of  $(\mathfrak{J}_p, \varphi, \varphi_1)$ . Indeed, we can show that given the choice of  $\varphi, \varphi_1$  and the asymptotic stability of  $\mathcal{T}, \mathfrak{J}_p$  will be uniquely determined. The explicit representation of  $\mathfrak{J}_p$  will be given in a later Remark 7.6.6.

## Chapter 7

## ASYMPTOTIC STABILITY OF CONTRACTION

In chapter 2 and chapter 4, we have studied the inverse spectral problem for selfadjoint Hankel operators and non self-adjoint Hankel operators as  $\mathfrak{C}$ -symmetric operators respectively. Notice that in both cases (Proposition 3.1.3 and Proposition 5.2.3) we require the asymptotic stability of a defined contraction  $\mathcal{T}$  (In chapter 2,  $\mathcal{T}$  is given as  $\mathcal{T} = \mathcal{R}_1 \mathcal{R}^{-1}$ . And in chapter 4,  $\mathcal{T}$  is given as  $\mathcal{T} = \mathcal{R}_1 \varphi_1 \varphi^* \mathcal{R}^{-1}$ ).

However, finding out an equivalent condition for the asymptotic stability is not an easy thing. The following *stability test* is a criterion for asymptotic stability given in [12].

**Lemma 7.0.1.** Let  $\mathcal{T}$  be a contraction on a Hilbert space. If  $\mathcal{T}$  has no eigenvalues on the unit circle  $\mathcal{T}$  and the set  $\sigma(\mathcal{T}) \cap \mathcal{T}$  is at most countable, then  $\mathcal{T}$  is asymptotically stable.

In addition, Nagy gave an equivalent condition for asymptotic stability in his book [26]. Here we first introduce the definition of *defect operators*, *defect indices*, and *minimal unitary dilation*, which will also be used in later chapters.

**Definition 7.0.2.** Let  $\mathcal{T}$  be a contraction on a Hilbert space  $\mathcal{H}$  (thus  $\mathcal{T}^*\mathcal{T} \leq I$  and  $\mathcal{T}\mathcal{T}^* \leq I$ ), we define the operators

$$D_{\mathcal{T}} = (I - \mathcal{T}^* \mathcal{T})^{1/2}, \qquad D_{\mathcal{T}^*} = (I - \mathcal{T} \mathcal{T}^*)^{1/2},$$

which are self-adjoint, and bounded by 0 and 1. We call  $D_{\mathcal{T}}$ ,  $D_{\mathcal{T}^*}$  be the defect operators of  $\mathcal{T}$ ,  $\mathcal{T}^*$  respectively.

In addition, we call  $\mathcal{D}_A := \operatorname{Clos} \operatorname{Ran} D_A$  to  $\mathcal{D}_{A^*} := \operatorname{Clos} \operatorname{Ran} D_{A^*}$  to be the defect spaces, and

$$\mathfrak{D}_{\mathcal{T}} := \dim \overline{D_{\mathcal{T}} \mathcal{H}}, \qquad \mathfrak{D}_{\mathcal{T}^*} := \dim \overline{D_{\mathcal{T}^*} \mathcal{H}}$$

to be the defect indices of  $\mathcal{T}$  and  $\mathcal{T}^*$  respectively.

**Definition 7.0.3.** Let  $\mathcal{A}$  be an operator on a Hilbert space  $\mathcal{H}_1$ , and  $\mathcal{B}$  be an operatoar on a Hilbert space  $\mathcal{H}_2$  which containing  $\mathcal{H}_1$ . We call  $\mathcal{B}$  a dilation of  $\mathcal{A}$  if

$$\mathcal{A}^n = P_{\mathcal{H}_1} \mathcal{B}^n$$

**Definition 7.0.4** (Theorem 4.2, Chapter 1,[26]). For every contraction  $\mathcal{T}$  on a Hilbert space  $\mathcal{H}_1$ , there exists a unitary dilation U on a space  $\mathcal{H}_2$  containing  $\mathcal{H}_1$  as a subspace, which is minimal, that is,

$$\mathcal{H}_2 = \overline{\mathrm{Span}} \{ U^n \mathcal{H}_1 | n \in \mathbb{Z} \}.$$

This minimal unitary dilation is determined up to isomorphism, and thus is called the minimal unitary dilation of  $\mathcal{T}$ .

With the preparations of those definitions above, Nagy gave an equivalent condition of asymptotic stability in his book [26], stated as below:.

**Theorem 7.0.5** (Proposition 1.3, Chapter 2, [26]). Let  $\mathcal{T}$  be a contraction on a Hilbert space  $\mathcal{H}_1$ , and U be its minimal dilation on a Hilbert space  $\mathcal{H}_2$  containing  $\mathcal{H}_1$ . Then  $\mathcal{T}$  is asymptotically stable if and only if the following two properties hold:

(i) The defect index  $\mathfrak{D}_{\mathcal{T}}$  is finite;

(ii) The minimal unitary dilation U is a bilateral shift of multiplicity equal to 𝔅<sub>T</sub>. That is, there exists a subspace 𝔅 ⊆ 𝔅<sub>2</sub> (called wandering space) with dimension 𝔅<sub>T</sub>, satisfies

$$\mathcal{H}_2 = \bigoplus_{n=-\infty}^{\infty} U^n \mathcal{L}$$

For the special case of complete non-unitary dilation (See Definition 7.3.1), the following result was stated:

**Theorem 7.0.6** (Proposition 6.7, Chapter 2, [26]). Let  $\mathcal{T}$  be a completely nonunitary contraction, and suppose that the intersection of the spectrum of  $\mathcal{T}$  with the unit circle  $\mathbb{T}$  has Lebesgue measure 0, then both  $\mathcal{T}$  and  $\mathcal{T}^*$  are asymptotically stable.

However, finding the spectral property of  $\mathcal{T}$  and its minimal unitary dilation U is still not an easy thing, but things become easier if we assume that  $R, R_1$  are compact. In this chapter, we mainly discuss the asymptotic stability of  $\mathcal{T} := R_1 \varphi_1 \varphi^* R^{-1}$  under the condition that  $R, R_1$  are compact.

We first restate the setting. We are given a tuple  $(R, R_1, p, \varphi, \varphi_1, \mathfrak{J}_p)$  satisfying

- (i)  $R, R_1$  are two positive, self-adjoint compact operators defined on a Hilbert space  $\mathcal{H}$ . In addition we have Ker  $R = \{0\}$ ;
- (ii)  $R^2 R_1^2 = pp^*$  for a vector p with  $||R^{-1}p|| \le 1$ ;
- (iii)  $\mathfrak{J}_p$  is a conjugation commutes with  $R, R_1$  and preserves p, implied by Lemma 4.0.1 and Lemma 5.1.1;
- (iv)  $\varphi$  is a  $\mathfrak{J}_p$ -symmetric unitary operator, which commutes with R;
- (v)  $\varphi_1$  is a  $\mathfrak{J}_p$ -symmetric partial isometry with Ker  $\varphi_1 = \text{Ker } R_1$ , which commutes with  $R_1$ . In addition, we have  $\varphi_1|_{(\text{Ker } R_1)^{\perp}}$  is unitary (See Remark 5.2.1);

In addition,  $\mathcal{T}$  is the unique contraction which satisfies

$$R_1\varphi_1 = \mathcal{T}R\varphi,\tag{7.0.1}$$

implied by Douglas Lemma 3.1.1.

In this chapter, we will show that

- (i) If  $\mathcal{H}_0 = \mathcal{H}$ , i.e., p is a cyclic vector for R. Then  $\mathcal{T}$  is automatically asymptotically stable without any further assumptions;
- (ii) If  $\mathcal{H}_0 \subsetneq \mathcal{H}$ , then the condition (ii) in Proposition 7.5.1 gives a criterion for the asymptotically stability of  $\mathcal{T}$ .

Finally, in section 7.6, we give an equivalent condition for the asymptotic stability of  $\mathcal{T}$  in Proposition 7.6.3 when taking the canonical choice of  $\varphi, \varphi_1$ . With this proposition, we can further analyze the behavior of  $\varphi, \varphi_1$  restricted on each eigenspaces of  $R, R_1$ , which will be discussed later in chapter 9.

### 7.1 Preparation

We first discuss the case when p is a cyclic vector with respect to R. We introduce the following lemma, which is a slight modification of [12, lemma 3.2].

**Lemma 7.1.1.** Let  $||\mathcal{T}|| \leq 1$ , and let K be a compact operator with dense range. Assume that an operator A satisfies

$$\mathcal{T}K = KA. \tag{7.1.1}$$

If A is weakly asymptotically stable, meaning that  $A^n \to 0$  in the weak operator topology (W.O.T) as  $n \to \infty$ , then T is asymptotically stable.

Proof. Iterating (7.1.1) we get that  $\mathcal{T}^n K = KA^n$ ,  $n \ge 1$ . Take  $x \in \mathcal{H}$ . Since  $A^n \to 0$  in W.O.T. and K is compact, we have that  $||KA^n x|| \to 0$ .

So  $\lim_{n\to\infty} \|\mathcal{T}^n y\| = 0$  for all  $y \in \operatorname{Ran} K$ . Thus, we have strong convergence on a dense set S.

For any  $y \in \mathcal{H}$  and any positive  $\varepsilon$ , we can find a x such that  $||x - y|| \leq \frac{\varepsilon}{3}$ . Then we can find a sufficiently large N, such that  $||\mathcal{T}^n x - x|| \leq \frac{\varepsilon}{3}$  holds for all  $n \geq N$ . Then we have

$$\|\mathcal{T}^{n}y - y\| \le \|y - x\| + \|x - \mathcal{T}^{n}x\| + \|\mathcal{T}^{n}x - \mathcal{T}^{n}y\| \le 2\|y - x\| + \|x - \mathcal{T}^{n}x\| \le \varepsilon$$

Thus we conclude (by  $\varepsilon/3$ -Theorem) that  $\mathcal{T}^n \to 0$  in the strong operator topology.

Recall that for an operator R (in a Hilbert space) its *modulus* |R| is defined as  $|R| := (R^*R)^{1/2}$ 

**Lemma 7.1.2.** For the operators R and  $\mathcal{T}$  from (7.0.1), there exists a unique contraction A, such that

$$\mathcal{T}R^{1/2} = R^{1/2}A$$

*Proof.* Since  $R_1\varphi_1 = \mathcal{T}R\varphi$ , we have

$$R_1^2 = (R_1\varphi_1)(R_1\varphi_1)^* = \mathcal{T}R\varphi\varphi^*R\mathcal{T}^* = \mathcal{T}R^2\mathcal{T}^*.$$

Hence by  $\mathcal{T}^*\mathcal{T} \leq I$ , we have

$$R^2 \ge R_1^2 = \mathcal{T}R^2\mathcal{T}^* \ge \mathcal{T}R\mathcal{T}^*\mathcal{T}R\mathcal{T}^* = (\mathcal{T}R\mathcal{T}^*)^2$$

This tells us  $R \geq \mathcal{T} R \mathcal{T}^*$  and

$$||R^{1/2}x|| \ge ||R^{1/2}\mathcal{T}^*x|| \qquad \text{holds for all} \qquad x \in H$$

Thus by Douglas Lemma 3.1.1, we can find a contraction denoted as  $A^*$ , satisfying

$$A^* R^{1/2} = R^{1/2} \mathcal{T}^*.$$

Taking the adjoint for the equation above, and we finish the proof of this lemma.  $\Box$ 

By Lemma 7.1.1 and Lemma 7.1.2 we can see that, in order to show that  $\mathcal{T}$  is asymptotically stable, it's sufficient to show that A is weakly asymptotically stable.

### 7.2 Case when *p* is Cyclic

To prove the weak asymptotic stability of A we need to investigate its structure in more detail.

We know that  $R_1^2 = R^2 - pp^* \le R^2$ . By the Löwner–Heinz inequality with  $\alpha = 1/2$ we have that  $R_1 \le R$ , so by Lemma 3.1.1 there exists an unique contraction Q such that

$$R_1^{1/2} = QR^{1/2} \tag{7.2.1}$$

The following simple proposition (modified from [12, Lemma 3.5]) gives an expression for operator A.

**Proposition 7.2.1.** The operator A from Lemma 7.1.2 is given by

$$A = Q^* \varphi_1 Q \varphi^*.$$

*Proof.* We calculate the representation of  $R_1\varphi_1$  in two different ways. First we have

$$R_1\varphi_1 = \mathcal{T}R^{1/2}R^{1/2}\varphi = R^{1/2}AR^{1/2}\varphi.$$

On the other hand,

$$R_1\varphi_1 = R^{1/2}Q^*QR^{1/2}\varphi_1.$$

Since Ker  $R = \{0\}$ , combining the two equations above, we get

$$A = Q^* Q R^{1/2} \varphi_1 \varphi^* R^{-1/2} = Q^* (R_1^{1/2} \varphi_1 \varphi^* R^{-1/2}) = Q^* \varphi_1 Q \varphi^*, \qquad (7.2.2)$$

which finishes our proof.

In addition, the following proposition gives the structure of Q.

**Proposition 7.2.2.** Let  $\mathcal{H}_0$  be the smallest invariant subspace of R that contains p. Then the operator Q with respect to the decomposition  $\mathcal{H} = \mathcal{H}_0 \oplus \mathcal{H}_0^{\perp}$  has the block structure

$$Q = \begin{pmatrix} Q_0 & 0\\ 0 & I \end{pmatrix},$$

where  $Q_0$  defined on  $\mathcal{H}_0$  is a strict contraction (i.e.  $||Q_0x|| < ||x||$  for all  $x \neq 0$ ).

*Proof.* We know that

$$R_1^2 = R^2 - pp^*, (7.2.3)$$

so  $\mathbb{R}^2$  coincides with  $\mathbb{R}^2_1$  on  $p^{\perp}$ .

One can easily see that  $\mathcal{H}_0$  is an invariant subspace for  $R^2$  and for  $R_1^2$ , and therefore so is  $\mathcal{H}_0^{\perp}$ . That means  $\mathcal{H}_0$  and  $\mathcal{H}_0^{\perp}$  are reducing subspaces for both  $R^2$  and  $R_1^2$ , i.e. that these operators in the decomposition  $\mathcal{H} = \mathcal{H}_0 \oplus \mathcal{H}_0^{\perp}$  are block diagonal. Therefore, the same is true for  $R^{1/2}$ 

Easy to see that  $R^{1/2}$  and  $R_1^{1/2}$  coincide on  $\mathcal{H}_0^{\perp}$ , which is a reducing space for both operators, so only need to show that  $Q_0$  is a strict contraction.

Using (7.2.3) and the identity  $R_1^{1/2} = QR^{1/2} = R^{1/2}Q^*$ , we can write

$$R^2 - pp^* = R_1^2 = R^{1/2} Q^* Q R Q^* Q R^{1/2}.$$

Recalling that  $p = R\varphi q$ , we can rewrite the above identity as

$$R^{1/2} \left( R - (R^{1/2} \varphi q) (R^{1/2} \varphi q)^* \right) R^{1/2} = R^{1/2} Q^* Q R Q^* Q R^{1/2}.$$

Since  $\operatorname{Ker} R = \{0\}$ , we have

$$Q^*QRQ^*Q = R - (R^{1/2}\varphi q)(R^{1/2}\varphi q)^*.$$
(7.2.4)

Applying both sides to x, and taking the inner product with x, we get

$$(RQ^*Qx, Q^*Qx) = (Rx, x) - |(x, R^{1/2}\phi(R)q)|^2.$$
(7.2.5)

Now, take x such that ||Qx|| = ||x||. Since  $||Q|| \le 1$ , we have

$$\langle x, x \rangle = \langle Qx, Qx \rangle = \langle x, Q^*Qx \rangle \le ||x|| ||Q^*Qx|| \le ||x||^2.$$

Thus we have  $x = Q^*Qx$ . The equation (7.2.5) can be rewritten in this case as

$$(Rx, x) = (Rx, x) - |(x, R^{1/2}\varphi q)|^2,$$

which implies that  $x \perp R^{1/2} \varphi q$ . Applying equation (7.2.4) to such x, and using again the fact that  $Q^*Qx = x$ , we get that

$$Q^*QRx = Rx.$$

Hence set  $\mathcal{H}_1 := \{h \in \mathcal{H} : h \in \mathcal{H}, \|Qh\| = \|h\|\} = \operatorname{Ker}(I - Q^*Q)$  is an invariant sub-

space for R, which is orthogonal to  $\varphi R^{1/2}q$ . Therefore

$$\mathcal{H}_1 \perp \overline{\operatorname{Span}} \{ R^n R^{1/2} \varphi q : n \ge 0 \} = \overline{\operatorname{Span}} \{ R^n p : n \ge 0 \} = \mathcal{H}_0,$$

and so  $Q_0 = Q|_{\mathcal{H}_0}$  is a strict contraction.

Now from Proposition 7.2.2, we have Q is a pure contraction since p is a cyclic vector for R, thus by proposition 7.2.1 we know that A is also a pure contraction. *Remark* 7.2.3. Indeed, we can also show that  $Q_0^*$  is also a pure contraction on  $\mathcal{H}_0$ . For all  $x, y \in \mathcal{H}_0$ , we have

$$\langle Q_0^* x, y \rangle = \langle x, Q_0 y \rangle \le ||x|| ||Q_0 y|| < ||x|| ||y||$$

Take  $y = Q_0^* x$ , then we have  $||Q_0^* x|| < ||x||$ .

#### 7.3 Case when p is not Cyclic

When p is not cyclic with respect to R inside  $\mathcal{H}$ , instead we show that A is a completely non-unitary contraction under certain assumptions of  $\varphi, \varphi_1$ . Beforehead, we recall the definition of completely nonunitary (c.n.u) contraction.

**Definition 7.3.1.** (From [26]) We call a contraction  $\mathcal{T} : \mathcal{H} \to \mathcal{H}$  a completely nonunitary contraction (c.n.u) if and only if for no nonzero reducing subspace E for  $\mathcal{T}$  is  $\mathcal{T}|_E$ is a unitary operator.

**Proposition 7.3.2.** The contraction A is completely non-unitary if and only if  $\varphi_1 \varphi^*$ does not have any non-zero reducing subspace  $E \subset \mathcal{H}_0^{\perp}$  such that  $\varphi^* E \perp \mathcal{H}_0$ .

To prove this proposition we need the following simple observation.

**Lemma 7.3.3.** Let A be a contraction and E be a subspace. The following statements are equivalent:

- (i) The subspace E is a reducing subspace for A such that  $A|_E$  is unitary;
- (ii) The operator A acts isometrically on E (i.e. ||Ax|| = ||x|| for all  $x \in E$ ) and AE = E.

*Proof of lemma 7.3.3.* The proof of this lemma is simple.

Suppose that we have (i), then we have  $A|_{E}$  is isometric and onto, thus (ii) holds.

For the other direction, suppose that we have (ii). Since we already have  $A|_E$  is isometric and onto, which implies that  $A|_E$  is unitary. It's sufficient to show that  $AE^{\perp} \subset E^{\perp}$ .

In fact, for  $\forall x \in E, y \in E^{\perp}$ , we have

$$\langle Ax, Ay \rangle = \langle A^*Ax, y \rangle = \langle x, y \rangle = 0,$$

here the first equality results  $A|_E$  is isometric. Thus E is reducing subspace for A, and (i) holds.

Proof of Proposition 7.3.2. Assume that  $E \subset \mathcal{H}_0^{\perp}$  is a reducing subspace for  $\varphi_1 \varphi^*$  such that  $\varphi^* E \subset \mathcal{H}_0^{\perp}$ .

Since  $Q = Q^* = I$  on  $\mathcal{H}_0^{\perp}$ , we have for any  $x \in E$ ,

$$Ax = Q^* \varphi_1 Q \varphi^* x = Q^* \varphi_1 \varphi^* x = \varphi_1 \varphi^* x; \qquad (7.3.1)$$

in the second equality we used the fact that  $\varphi^* x \in \mathcal{H}_0^{\perp}$ , and in the last one the fact that  $\varphi_1 \varphi^* x \in E \subset \mathcal{H}_0^{\perp}$ .

Since  $\varphi_1$  acts isometrically on  $\mathcal{H}_0^{\perp}$  (Ker  $\varphi_1$  can only belongs to  $\mathcal{H}_0$ ), we have

$$||Ax|| = ||\varphi_1 \varphi^* x|| = ||x||,$$

i.e. A acts isometrically on E.

Also, this implies that  $\varphi_1 \varphi^*$  acts isometrically on E. In addition,  $\varphi$  is unitary, and

(see Remark 5.2.1) we know  $\varphi_1|_{(\text{Ker }R_1)^{\perp}}$  is unitary, therefore  $\varphi \varphi_1^*$  acts unitarily on its reducing subspace E. Now from (7.3.1) we have

$$AE = \varphi_1 \varphi^* E = E.$$

Applying Lemma 7.3.3, we can see that E is a reducing subspace for A such that  $A|_{E}$  is unitary, so A is not c.n.u.

Now let's move on to the sufficiency. If A is not c.n.u., then we can find a reducing subspace E for A, such that  $A|_E = Q^* \varphi_1 Q \varphi^*|_E$  is unitary. Using that Q is a pure contraction on  $H_0$ , we have

$$\varphi^* E \subset \mathcal{H}_0^{\perp}, \qquad \varphi_1 \varphi^* E \subset \mathcal{H}_0^{\perp};$$
  
$$AE = \varphi_1 \varphi^* E = E.$$

Thus  $E = \varphi_1 \varphi^* E \subset \mathcal{H}_0^{\perp}$ . In addition, since  $\varphi_1 \varphi^* E^{\perp} \perp \varphi_1 \varphi^* E$ , thus  $\varphi_1 \varphi^* E^{\perp} \subseteq E^{\perp}$ , and E is a reducing subspace for  $\varphi_1 \varphi^*$ . This gives a contradiction, so we finish the proof.

#### 7.4 C.n.u Implies Weakly Asymptotic Stability

Recall that in Subsection 7.2, we have shown that the operator A is a pure contraction, and in Subsection 7.3 we have shown that A is c.n.u under certain assumptions for  $\varphi, \varphi_1$ . Since a pure contraction is certainly c.n.u from the Definition 7.3.1, it's sufficient to prove that completely non-unitary implies asymptotical stability.

The following proposition and its proof is inspired from [27].

**Proposition 7.4.1.** Let A be a completely non unitary contraction on a Hilbert space  $\mathcal{H}$ , then A is weakly asymptotically stable.

*Proof.* Denote set  $H_1$  as

$$H_1 := \left\{ x \big| \|A^n x\| = \|A^{*n} x\| = \|x\| \text{ holds for } \forall n \in \mathbb{N} \right\},$$
(7.4.1)

we firstly show that  $H_1 = \{0\}$ .

For  $\forall x \in H_1$  and  $\forall n \in \mathbb{N}$ ,

$$||x||^{2} = ||A^{n}x|| = \langle A^{n}x, A^{n}x \rangle = \langle A^{*n}A^{n}x, x \rangle \le ||A^{*n}A^{n}x|| ||x|| \le ||x||^{2},$$

thus we have  $A^{*n}A^nx = x$  holds for all n, and similarly we have  $A^nA^{*n}x = x$ .

On the contrary, if a vector **x** satisfies that  $A^{*n}A^nx = A^nA^{*n}x = x$  holds for all n, then we have

$$||A^n x||^2 = \langle A^{*n} A^n x, x \rangle = \langle x, x \rangle = ||x||^2,$$

and similarly we have  $||A^{*n}x|| = ||x||$ , thus we have

$$H_1 = \left\{ x \middle| A^{*n} A^n x = A^n A^{*n} x = x \text{ holds for } \forall n \in \mathbb{N} \right\}.$$
 (7.4.2)

Thus  $H_1$  is a subspace. Furthermore,  $H_1$  is a reducing subspace for A and  $A^*$ . In fact, for all  $x \in H_1$ , we have

$$||A^{n}Ax|| = ||x||, \qquad ||A^{*n}Ax|| = ||A^{*(n-1)}x|| = ||x||$$
$$||A^{n}A^{*}x|| = ||A^{n-1}x|| = ||x||, \qquad ||A^{*n}A^{*}x|| = ||A^{*(n+1)}x|| = ||x||$$

Thus by (7.4.1), we have  $Ax, A^*x \in H_1$ . Since we also have  $A|_{H_1}, A^*|_{H_1}$  is onto due to (7.4.2), applying Lemma 7.3.3 we have  $A|_{H_1}$  is unitary. Since we are given that A is c.n.u, we have  $H_1 = \{0\}$ .

Now we prove that A is weakly asymptotically stable by contradiction. Assume

that A is not weakly asymptotically stable, then there exists  $x, y \in \mathcal{H}$ , a positive  $\varepsilon$ , and a subsequence  $\{n_j\}_{j=1}^{\infty}$  such that

$$|\langle A^{n_j}x, y \rangle| > \varepsilon. \tag{7.4.3}$$

We know that the set  $\{A^{n_j}x\}_{j=1}^{\infty}$  is bounded, and since  $\mathcal{H}$  is reflexive and separable, thus weakly compactness implies sequentially weakly compactness, hence there exists a subsequence of  $\{A^{n_j}x\}_{j=1}^{\infty}$  which is weakly convergent. Without lose of generality, we take this subsequence to be  $\{A^{n_j}x\}_{j=1}^{\infty}$  itself. Denote  $A^{n_j}x \rightharpoonup x_0$  for a  $x_0$ .

Thus for a fixed  $k \in \mathbb{N}$  we have

$$A^{*k}A^kA^{n_j}x \rightharpoonup A^{*k}A^kx_0 \text{ when } n_j \to \infty.$$
(7.4.4)

On the other hand, we can show that  $A^{*k}A^kA^{n_j}x \rightharpoonup x_0$ . In fact we have

$$\begin{split} \|A^{*k}A^{k}A^{n_{j}}x - A^{n_{j}}x\|^{2} &= \|A^{*k}A^{k}A^{n_{j}}x\|^{2} + \|A^{n_{j}}x\|^{2} - 2\langle A^{*k}A^{k}A^{n_{j}}x, A^{n_{j}}x\rangle \\ &\leq \|A^{k+n_{j}}x\|^{2} + \|A^{n_{j}}x\|^{2} - 2\|A^{k+n_{j}}x\|^{2} \\ &= \|A^{n_{j}}x\|^{2} - \|A^{k+n_{j}}x\|^{2}. \end{split}$$

Here the right hand side of the above inequality goes to 0 when  $n_j \to \infty$ , because sequence  $||A^n x||_{n=1}^{\infty}$  is monotonically decreasing, hence convergent. And since  $A^{n_j} x \to x_0$ , we have

$$A^{*k}A^kA^{n_j}x \to x_0 \text{ when } n_j \to \infty.$$
 (7.4.5)

Together with (7.4.4), we have  $A^{*k}A^kx_0 = x_0$ . Similar to the process above, we can get  $A^kA^{*k}x_0 = x_0$ . By (7.4.2), we have  $x_0 \in H_1$ , which implies that  $x_0 = 0$ . Hence  $A^{n_j}x \to 0$  weakly which contradicts to (7.4.3).

Remark 7.4.2. We can also use functional model theory to shorten the proof of Propo-

sition 7.4.1.

Since every completely non-unitary operator A is unitary equivalent to a so-called model operator  $M_{\theta}$  defined on the functional space  $K_{\theta}$ . Here  $\theta(z)$  is called to be the *characteristic function* of A, which is an analytic function defined as

$$\theta_A(z) = \left( -A + z D_{A^*} (I - z A^*)^{-1} D_A \right) \Big|_{\mathcal{D}_A}, \qquad z \in \text{ unit disk } D.$$
(7.4.6)

For a given z satisfying that  $I - zA^*$  is bounded invertible,  $\theta_A(z)$  is a bounded operator mapping  $\mathcal{D}_A := \operatorname{Clos}\operatorname{Ran} D_A$  to  $\mathcal{D}_{A^*} := \operatorname{Clos}\operatorname{Ran} D_{A^*}$ .

And the model space  $K_{\theta}$  is an appropriately constructed subspace of a weighted space  $L^2(E^* \oplus E, W)$  on the unit circle  $\mathbb{T}$  with respect to the normalized Lebesgue measure W on  $\mathbb{T}$ , where  $E^*, E$  are some Hilbert spaces with dimensions  $\mathfrak{D}_{A^*}, \mathfrak{D}_A$  respectively (See Definition 7.0.2). And the model operator  $M_{\theta}$  is the compression of the multiplication operator  $M_z$  onto  $K_{\theta}$ . That is,

$$M_{\theta} = P_{\theta} M_z \big|_{K_{\theta}},\tag{7.4.7}$$

Here  $P_{\theta}$  is the orthogonal projection onto  $K_{\theta}$ . The specific choice of  $K_{\theta}$  can be found in [26, Proposition 2.1, Chapter VI]. (As for the foundation of functional model theory, details can be found in [26, Chapter VI] and [28, P. 109-115]).

From (7.4.7), we have

$$M_{\theta}^{n} = P_{K_{\theta}} M_{z}^{n} \big|_{K_{\theta}},$$

since  $M_z^n \to 0$  when  $n \to 0$  in the weak operator topology of  $B(L^2(E^* \oplus E, W))$  (the set of all bounded operators on this space), we claim that  $M_{\theta}^n \to 0$  when  $n \to \infty$  in the weak operator topology of  $B(K_{\theta})$  as well, thus  $A^n \to 0$  in the weak operator topology.

#### 7.5 Main Result

Combining the results in subsections 6.1-6.4, we reach the following proposition.

**Proposition 7.5.1.**  $\mathcal{T}$  defined by  $\mathcal{T} := \varphi_1 R_1 R^{-1} \varphi^*$  is asymptotically stable if one of the following two conditions hold

(i)  $\mathcal{H}_0 = \mathcal{H}$ , *i.e.* p is cyclic with respect to R in  $\mathcal{H}$ ;

(ii)  $\varphi_1 \varphi^*$  does not have any non-zero reducing subspace  $E \subset \mathcal{H}_0^{\perp}$  such that  $\varphi^* E \perp \mathcal{H}_0$ .

Proof of Proposition 7.3.3. To show that  $\mathcal{T}$  is asymptotically stable, it suffices to show that the operator A defined by

$$\mathcal{T}R^{1/2} = R^{1/2}A$$

is weakly asymptotically stable according to Lemma 7.1.1 and Lemma 7.1.2. And furthermore by Lemma 7.4.1, it's sufficient to show that A is a completely non-unitary contraction.

When  $\mathcal{H}_0 = \mathcal{H}$  holds, by Proposition 7.2.1 we have  $A = Q^* \varphi_1 Q \varphi^*$ . Since Q is a pure contraction according to Proposition 7.2.2, we have A is also a pure contraction, thus a completely non-unitary contraction.

When  $\mathcal{H}_0 \subsetneq \mathcal{H}$ , according to Proposition 7.3.2, we still have A is a c.n.u. So we finish the proof.

#### 7.6 Asymptotic Stability under the Canonical Choice

In this section, we consider the asymptotic stability of contraction  $\mathcal{T}$  under the canonical choice of  $\varphi, \varphi_1$ . We will show that the requirement in Proposition 7.5.1 can be substituted by a sufficient and necessary condition.

Recall the definition of four eigenspaces  $E_{\lambda_k}$  and  $E_{\lambda_k}^1$ ,  $E_{\mu_k}$  and  $E_{\mu_k}^1$  stated in (6.2.7), the definition of  $p_k$ ,  $p_k^1$  given in (6.2.9), and the canonical choice of  $\varphi$ ,  $\varphi_1$  given in Lemma 6.3.2:

$$\varphi_1|_{E^1_{\lambda_k}} = I, \qquad \varphi|_{E_{\mu_k}} = I. \tag{7.6.1}$$

We first state the definition of star-cyclicity.

**Definition 7.6.1.** Here we say a vector x is \*-cyclic for A in  $\mathcal{H}$  if and only if

$$\overline{\operatorname{Span}}\big\{A^n x, (A^*)^n x | n \ge 0\big\} = \mathcal{H}.$$

An equivalent definition in [29, Def 3.1, Chapter IX, p. 268] is given as follow:

We say a vector x is \*-cyclic for A in  $\mathcal{H}$  if and only if  $\mathcal{H}$  is the smallest reducing subspace for A that contains x.

**Definition 7.6.2.** We say that an operator A on  $\mathcal{H}$  is cyclic if and only if there exists a vector  $x \in \mathcal{H}$ , such that

$$\overline{\operatorname{Span}}\left\{A^n x | n \ge 0\right\} = \mathcal{H}.$$

Similarly, we say an operator A on  $\mathcal{H}$  is \*-cyclic if and only if there exists a vector  $x \in \mathcal{H}$ , such that

$$\overline{\operatorname{Span}}\left\{A^n x, (A^*)^n x | n \ge 0\right\} = \mathcal{H}.$$

J. Bram proved in [30] that a normal operator A is \*-cyclic if and only if A is cyclic.

The results below gives an equivalent condition of  $\varphi, \varphi_1$  which guarantees the asymptotic stability of  $\mathcal{T}$ .

**Proposition 7.6.3.** Under the choice of  $\varphi, \varphi_1$  given in (7.6.1),  $\mathcal{T}$  is asymptotically stable if and only if both of the two conditions below are satisfied:

(i) For all k the vector  $p_k$  is \*-cyclic for the operator  $\varphi|_{E_{\lambda_k}}$ ;

(ii) For all k the vector  $p_k^1$  is \*-cyclic for the operator  $\varphi_1\Big|_{E_{l_1}^1}$ .

Proof of Lemma 7.6.3. If for some k the vector  $p_k$  is not \*-cyclic for  $\varphi|_{E_{\lambda_k}}$ , then there exists a subspace  $H_k \subset E_{\lambda_k}$ ,  $H_k \perp p_k$  (and so  $H_k \subset E_{\lambda_k}^1$ ) which is a reducing subspace for  $\varphi|_{E_{\lambda_k}}$  (and thus for  $\varphi$ ).

We know that  $R = R_1$  on  $E_{\lambda_k}^1$ , and that  $\varphi_1|_{E_{\lambda_k}^1} = I$ , so the representation  $\mathcal{T} = \varphi_1 R_1 R^{-1} \varphi^*$  implies that

$$\mathcal{T}x = \varphi^* x \in H_k \qquad \forall x \in H_k$$

Therefore  $H_k$  is an invariant subspace for  $\mathcal{T}$  (in fact, one can show that it is a reducing subspace, but we do not need this in the proof) on which  $\mathcal{T}$  acts isometrically, and so  $\mathcal{T}$  cannot be asymptotically stable.

Similarly, if  $p_k^1$  is not \*-cyclic for  $\varphi_1|_{E_{\mu_k}^1}$  for a certain k, then we can find a subspace of  $E_{\mu_k}^1$  denoted as  $H_k^1$ ,  $H_k^1 \perp p_k^1$  (simply set  $H_k^1 := E_{\mu_k}^1 \ominus \overline{\text{Span}}\{(\varphi_1^*)^n p_k^1, \varphi_1^n p_k^1 : n \ge 0\})$ , which is a reducing subspace for  $\varphi_1|_{E_{\mu_k}^1}$ .

Since  $\varphi|_{E_{\mu_k}} = I$ , applying any  $x \in H_k^1$  to  $R_1\varphi_1 = \mathcal{T}R\varphi$  implies that

$$\mathcal{T}x = \varphi_1 x \qquad \forall x \in H_k^1$$

Hence  $H_k^1$  is an invariant subspace for  $\mathcal{T}$ . Noticing that  $\varphi_1$  is an isometry restricted on  $\mathcal{H}_0^{\perp}$ , thus  $\mathcal{T}$  can't be asymptotically stable.

So, we proved that the conditions (i), (ii) are necessary for the asymptotic stability of  $\mathcal{T}$ .

To prove the sufficiency of these conditions, we will show that under (i), (ii),  $A = Q^* \varphi_1 Q \varphi^*$  is c.n.u. Then proposition 7.4.1 implies that A is weakly asymptotically stable, and so by Lemma 7.1.1 and Lemma 7.1.2,  $\mathcal{T}$  is asymptotically stable.

So, let us assume that conditions (i), (ii) are satisfied, but A is not c.n.u. Then there exists a reducing subspace L where  $A|_L$  is unitary. Since  $A^* = \varphi Q^* \varphi_1^* Q$  acts isometrically on L, we have  $L \subseteq \mathcal{H}_0^{\perp}$ . From  $A|_L = (Q^* \varphi_1 Q \varphi^*)|_L$  is unitary, and we also know from Proposition 7.2.2 and Remark 7.2.3 that both Q and  $Q^*$  are pure contractions on  $\mathcal{H}_0$ , thus

$$\varphi^* L \subseteq \mathcal{H}_0^{\perp}, \qquad \varphi_1 \varphi^* L = L. \tag{7.6.2}$$

Now take a vector  $x \in L \subseteq \mathcal{H}_0^{\perp}$ . We write x = a + b where  $a \in \bigoplus_{k=1}^{\infty} E_{\lambda_k}^1$  and  $b \in \bigoplus_{k=1}^{\infty} E_{\mu_k}$ . Then

$$\varphi^* x = \varphi^* a + \varphi^* b = \varphi^* a + b.$$

Here  $\varphi^* a \in \bigoplus_k E_{\lambda_k} = \bigoplus_k E_{\lambda_k}^1 \oplus \mathcal{H}_0$ . On the other hand, from (7.6.2) we have  $\varphi^* x \in \mathcal{H}_0^{\perp}$ , thus  $\varphi^* a \in \bigoplus_k E_{\lambda_k}^1$ .

So now we have

$$Ax = \varphi_1 \varphi^* x = \varphi^* a + \varphi_1 b,$$

where here we use  $\varphi_1|_{E^1_{\lambda_k}} = I$  in the second identity. Since  $Ax \in L \subseteq \mathcal{H}_0^{\perp}$ , we have  $\varphi_1 b \in \bigoplus_{k=1}^{\infty} E_{\mu_k}$ . Hence by induction, we will get

$$A^{n}x = (\varphi^{*})^{n}a + (\varphi_{1})^{n}b, \qquad (7.6.3)$$

holds for all  $n \in \mathbb{N}$ , where  $(\varphi^*)^n a \in \bigoplus_k E^1_{\lambda_k}, \varphi_1^n b \in \bigoplus_k E_{\mu_k}$ .

And similar property holds for  $A^*$ , with the same notation of x, a, b, we have

$$(A^*)^n x = \varphi^n a + (\varphi_1^*)^n b, \tag{7.6.4}$$

where  $\varphi^n a \in \bigoplus_k E^1_{\lambda_k}, (\varphi_1^*)^n b \in \bigoplus_k E_{\mu_k}.$ 

Now we need the following simple lemma.

**Lemma 7.6.4.** Let U be an arbitrary operator on a Hilbert space  $\mathcal{H}$ , and x is a \*-cyclic vector with respect to U on  $\mathcal{H}$ , then for  $\forall y \in \mathcal{H}$ , we have

$$P_x \overline{\operatorname{Span}} \{ U^n y, (U^*)^n y | n \ge 0 \} \neq 0.$$

Here P is the projection operator.

Proof of Lemma 7.6.4. If the projection is zero, then we have  $x \perp U^n y, (U^*)^n y$  for all n. Thus accordingly, we have

$$y \perp U^n x, (U^*)^n x$$
 for all  $n$ ,

from the \*-cyclicity of x, we have  $y \perp \mathcal{H}$ , which gives a contradiction.

Now back to the proof of Proposition 7.6.3. For the chosen vector  $x \in L$ , there exists a  $k \in \mathbb{N}$  such that  $x_k^1 := P_{E_{\lambda_k}^1} x \neq 0$  or  $x_k := P_{E_{\mu_k}} x \neq 0$ .

If  $x_k^1 \neq 0$ , then applying Lemma 7.6.4 we have

$$P_{p_k}\overline{\operatorname{Span}}\left\{\varphi^n x_k^1, (\varphi^*)^n x_k^1 | n \ge 1\right\} \neq 0.$$

Thus we can find a  $n \in \mathbb{N}$  such that  $P_{p_k}(\varphi^n x_k^1) \neq 0$  or  $P_{p_k}(\varphi^*)^n x_k^1 \neq 0$ .

If  $P_{p_k}(\varphi^n x_k^1) \neq 0$ , then it contradicts to the equation (7.6.4) where we have  $\varphi^n a \in \bigoplus_k E_{\lambda_k}^1$ . If  $P_{p_k}(\varphi^*)^n x_k^1 \neq 0$ , then it contradicts to the equation (7.6.3) where we have  $(\varphi^*)^n a \in \bigoplus_k E_{\lambda_k}^1$ .

Similarly, if  $x_k \neq 0$  for a k, applying Lemma 7.6.4 we have

$$P_{p_k^1}\overline{\operatorname{Span}}\big\{\varphi_1^n x_k, (\varphi_1^*)^n x_k | n \ge 1\big\} \neq 0.$$

Then we can also find a  $n \in \mathbb{N}$  such that  $P_{p_k^1}(\varphi_1^n x_k) \neq 0$  or  $P_{p_k^1}(\varphi_1^*)^n x_k \neq 0$ . Thus we have  $\varphi_1^n b \notin \bigoplus_k E_{\mu_k}$  or  $(\varphi_1^*)^n b \notin \bigoplus_k E_{\mu_k}$ , which contradicts to (7.6.3) and (7.6.4) respectively, and it will also leads to a contradiction. So there doesn't exist such reducing subspace L satisfying  $A|_L$  is unitary. Hence A is c.n.u, and we finish the proof of sufficiency part.

Remark 7.6.5. Since  $\varphi|_{E_{\lambda_k}}$  is unitary and finite rank, with this restriction, we can show that only cyclicity (instead of \*-cyclicity) is required. That is,

- (i) For all k the vector  $p_k$  is cyclic for the operator  $\varphi|_{E_{\lambda_k}}$  on  $E_{\lambda_k}$ ;
- (ii) For all k the vector  $p_k^1$  is cyclic for the operator  $\varphi_1\Big|_{E^1_{\mu_k}}$  on  $E^1_{\mu_k}$ .

The specific explanantion of this part can be found in Remark 10.1.4.

*Remark* 7.6.6. We need to mention here that the conjugation  $\mathfrak{J}_p$  will be uniquely determined under the canonical choice of  $\varphi, \varphi_1$ . Since all eigenspaces of  $R, R_1$  are invariant subspaces for  $\mathfrak{J}_p$ , and we have

$$\mathcal{H} = \big(\bigoplus_{k=1}^{\infty} E_{\lambda_k}\big) \oplus \big(\bigoplus_{k=1}^{\infty} E_{\mu_k}\big),$$

it's sufficient to determine value of  $\mathfrak{J}_p$  on  $E_{\lambda_k}$  and  $E^1_{\mu_k}$  (since  $E_{\mu_k} \subset E^1_{\mu_k}$ ) for all k.

We first show that  $\mathfrak{J}_p p_k = p_k, \mathfrak{J}_p p_k^1 = p_k^1$  hold for all k. In fact, since  $\mathfrak{J}_p R = R\mathfrak{J}_p$ , apply  $p_k$  on both sides we get

$$R\mathfrak{J}_p p_k = \mathfrak{J}_p R p_k = \lambda_k \mathfrak{J}_p p_k.$$

Since  $\mathfrak{J}_p p_k \in \mathcal{H}_0$ , we have  $\mathfrak{J}_p p_k = \alpha_k p_k$  for a  $\alpha_k \in \mathbb{C}$  with module 1. Hence we have

$$p = \mathfrak{J}_p p = \mathfrak{J}_p \sum_{k=1}^{\infty} p_k = \sum_{k=1}^{\infty} \overline{\alpha_k} p_k,$$

thus  $\alpha_k = 1$  and  $\mathfrak{J}_p p_k = p_k$  hold for all k.  $\mathfrak{J}_p p_k^1 = p_k^1$  can be shown in a similar process.

Now since  $\mathfrak{J}_p p_k = p_k$  and  $\varphi|_{E_{\lambda_k}}$  is  $\mathfrak{J}_p$ -symmetric, thus for all  $n \in \mathbb{N}$  we have

$$\mathfrak{J}_p \varphi^n p_k = (\varphi^*)^n \mathfrak{J}_p p_k = (\varphi^*)^n p_k,$$
$$\mathfrak{J}_p (\varphi^*)^n p_k = \varphi^n \mathfrak{J}_p p_k = \varphi^n p_k.$$

thus  $\mathfrak{J}_p|_{E_{\lambda_k}}$  is uniquely determined implied by Proposition 7.6.3.

Similarly we have  $\mathfrak{J}_p\big|_{E^1_{\mu_k}}$  is uniquely determined by

$$\mathfrak{J}_p(\varphi_1^n p_k^1) = (\varphi_1^*)^n p_k^1, \qquad \mathfrak{J}_p((\varphi_1^*)^n p_k^1) = \varphi_1^n p_k^1 \qquad \text{for all } n.$$

Thus the uniqueness of  $\mathfrak{J}_p$  is proved.

## Chapter 8

## **ABSTRACT BORG'S THEOREM**

In a previous Proposition 6.2.2, we have shown that for a given rank-one perturbation  $(R, R_1, p, \varphi, \varphi_1)$  with  $R^2 - R_1^2 = pp^*$ , under the cyclicity of vector p and the asymptotical stability of contraction  $\mathcal{T}$ , there exists an intertwining sequence

$$\lambda_1 > \mu_1 > \lambda_2 > \mu_2 > \dots \to 0,$$
 (8.0.1)

such that  $R|_{\mathcal{H}_0}, R_1|_{\mathcal{H}_0}$  has simple eigenvalues as  $\{\lambda_k\}_{k=1}^{\infty}$ .  $\{\mu_k\}_{k=1}^{\infty}$  respectively.

Thus in this chapter we consider the inverse problem. Given an intertwining real sequence

$$\lambda_1 > \mu_1 > \lambda_2 > \mu_2 > \dots \to 0,$$

can we find a rank-one perturbation  $(R, R_1, p)$ , such that  $R, R_1$  has the corresponding eigenvalues? Fortunately the answer to this question is positive, and in this chapter we will prove the following so-called The Abstract Borg's Theorem:

**Theorem 8.0.1** (Abstract Borg's Theorem). Given two sequences  $\{\lambda_k^2\}_{k\geq 1}$  and  $\{\mu_k^2\}_{k\geq 1}$ satisfying intertwining relations (8.0.1) and such that  $\lambda_k^2 \to 0$  as  $k \to \infty$ , there exists an unique (up to unitary equivalence) triple  $(W, W_1, p)$ , such that  $W = W^* \geq 0$ , Ker  $W = \{0\}$  is a compact operator with simple eigenvalues  $\{\lambda_k^2\}_{k=1}^\infty$ , and the operator  $W_1 = W - pp^*$  has  $\{\mu_k^2\}_{k=1}^{\infty}$  as its (simple) non-zero eigenvalues ( $W_1$  can also have a simple eigenvalue at 0).

Moreover, the two identities  $||W^{-1/2}p|| = 1$ ,  $||W^{-1}p|| = \infty$  hold if and only if:

$$\sum_{j=1}^{\infty} \left( 1 - \frac{\mu_j^2}{\lambda_j^2} \right) = \infty, \qquad (8.0.2)$$

$$\sum_{j=1}^{\infty} \left( \frac{\mu_j^2}{\lambda_{j+1}^2} - 1 \right) = \infty.$$
(8.0.3)

Remark 8.0.2. The original Borg's theorem [32] states that the potential q of a Schrödinger operator L, Ly = y'' + q(x)y on an interval is uniquely defined by the two sets of eigenvalues, corresponding to two specific boundary conditions. Later Levinson [33] extended this result by showing that essentially any non-degenerate pair of self-adjoint boundary conditions would work.

Changing boundary conditions for a Schrödinger operator is essentially a rank one perturbation (by an unbounded operator). Namely, if  $L_1$  and  $L_2$  are Schrödinger operators on an interval with the same potential, but with two different self-adjoint boundary conditions, then for any  $\lambda \notin \sigma(L_1) \cup \sigma(L_2)$  the difference  $(L_1 - \lambda I)^{-1} - (L_2 - \lambda I)^{-1}$  is a rank one operator (and the operators  $(L_1 - \lambda I)^{-1}$ ,  $(L_2 - \lambda I)^{-1}$  are compact). Thus, by picking a real  $\lambda$  the problem can be reduced to rank one perturbations of compact self-adjoint operators.

Our Theorem 8.0.1 deals with rank one perturbations of (abstract) compact selfadjoint operators, hence we give its name. We do not assume that our operators came from Schrödinger operators, so we only reconstruct the spectral measure, and are not concerned with the reconstruction of the potential. However, it is well known how to reconstruct the potential from the spectral measure, or, more precisely, from the Titchmarsh–Weyl m-function, so it should be possible to get the Borg's result from our abstract theorem.

Note also, that Theorem 8.0.1 gives not only uniqueness, but the existence as well.

*Remark* 8.0.3. Note also that the triple  $(W, W_1, p)$  we constructed in Theorem 8.0.1 satisfies that  $P_{\text{Ker}(W-\lambda_k^2 I)}p \neq 0$  for all k, hence by Lemma 6.1.1 the cyclicity of p with respect to W is automatically satisfied. In fact, if there is a k such that

$$p_k := P_{\operatorname{Ker}(W - \lambda_k^2 I)} p = 0$$

Then  $\lambda_k^2$  is a common eigenvalue for  $W, W_1$ , which gives us contradiction.

Similarly we also have  $P_{\text{Ker}(W_1-\mu_k^2 I)}p \neq 0$  holds for all k.

In section 8.1, we prove the existence and uniqueness part of the abstract Borg's theorem. Inside this proof, we give an expression for the scalar spectral measure of W with respect to p (see definition in (8.1.3)), denoted as  $\rho(s)$ , and its coefficients has the form in (8.1.9).

In section 8.2, we will translate the trivial kernel condition of  $\Gamma$  we get from Proposition 3.1.3 and Proposition 5.2.3, in terms of  $\{\lambda_k\}_{k=1}^{\infty}$  and  $\{\mu_k\}_{k=1}^{\infty}$ . Within this part, we also give an equivalent condition of Ker  $W_1 = \{0\}$ , and furthermore, give an expression for the scalar spectral measure of  $W_1$  with respect to p (see the definition in (8.1.4)), denoted as  $\rho_1(s)$ , and its coefficients has the form in Proposition 8.2.9.

In section 8.3, we discuss a special case, i.e., a finite-rank version for The Abstract Borg's Theorem. Here the proof for Theorem 8.3.1 is similar to the proof of Theorem 8.0.1, so we omit it.

Finally we close this chapter by considering the inverse problem for the case of rank-one perturbation  $(W, W_1, p)$ , when p is not cyclic with respect to W. Given an intertwining real sequence

$$\lambda_1 > \mu_1 > \lambda_2 > \mu_2 > \dots \to 0$$

with their corresponding geometric multiplicities, we can also reconstruct the triple  $(W, W_1, p)$ . In this case, we will have the following result, which is a simple generalized

version of Abstract Borg's theorem:

**Theorem 8.0.4** (Generalized abstract Borg's theorem). Given positive sequences  $\{\lambda_k\}_{k=1}^{\infty}$ and  $\{\mu_k\}_{k=1}^{\infty}$  satisfying the following intertwining property:

$$\lambda_1 > \mu_1 > \lambda_2 > \mu_2 > \dots > \lambda_k > \mu_k \dots \to 0,$$

and also integer sequences  $\{m(\lambda_k)\}_{k=1}^{\infty}$ ,  $\{m(\mu_k)\}_{k=1}^{\infty}$  satisfying for  $\forall k \geq 0$ ,  $m(\lambda_k) \geq 1$ ,  $m(\mu_k) \geq 0$ , then there exists a unique triple  $(W, W_1, p)$  satisfying:

- (i)  $W, W_1$  are positive, compact self-adjoint operators;
- (ii)  $W W_1 = pp^*;$
- (iii) For all k ∈ N, W has singular values λ<sup>2</sup><sub>k</sub> with multiplicity m(λ<sub>k</sub>), and singular values μ<sup>2</sup><sub>k</sub> with multiplicity m(μ<sub>k</sub>);
- (iv) For all  $k \in \mathbb{N}$ ,  $W_1$  has singular values  $\lambda_k^2$  with multiplicity  $m(\lambda_k) 1$ , and singular values  $\mu_k^2$  with multiplicity  $m(\mu_k) + 1$ ;

We will prove this Theorem 8.0.4 in the last section 8.4 of this chapter.

# 8.1 Proof of Abstract Borg's Theorem: Existence and Uniqueness Part

# 8.1.1 Some Preparation Work: the Definition of Scalar Spectral Measure and Cauchy Transform

Since everything is defined up to unitary equivalence, by Von Neumann Theorem 1.2.7 we can then assume without loss of generality that W is the multiplication  $M_s$  by the independent variable s in the weighted space  $L^2(\rho)$ , and the vector p is represented by the function 1 in  $L^2(\rho)$ . Here  $\rho(s)$  is called to be the scalar spectral measure with

respect to the vector p, defined as following:

$$\left\langle (W - zI)^{-1}p, p \right\rangle = \int_{\sigma(W)} \frac{d\rho(s)}{s - z}.$$
(8.1.1)

We also call  $F(z) := \int \frac{d\rho(s)}{s-z}$  to be the Cauchy transform of  $\rho(s)$ .

Since W is a compact operator with eigenvalues  $\{\lambda_k^2\}_{k\geq 1}$ , the measure  $\rho$  is purely atomic, that is:

$$\rho = \sum_{k \ge 1} a_k \delta_{\lambda_k^2}, \qquad a_k > 0.$$
(8.1.2)

Since the choice of  $a_k$  can uniquely determine the triple  $W, W_1, p$  up to unitary equivalence. We want to find out values of  $\{a_k\}_{k=1}^{\infty}$  according to the given intertwining sequence.

The following proposition gives the Cauchy transform of an operator under a rank one perturbation, also stated in cf. [11, Chapter 9], [34, Theorem 5.8.1, p. 335]

**Proposition 8.1.1** (Aronszajn-Krein formula). Assume that  $W_{\alpha}$  is a rank-one perturbation of operator  $W: W_{\alpha} = W - \alpha pp^*$ , then the Borel transform of W and  $W^{\alpha}$ defined as

$$F(z) = \langle (W - zI)^{-1}p, p \rangle, \qquad F_{\alpha}(z) = \langle (W_{\alpha} - zI)^{-1}p, p \rangle$$

have the following relation

$$F_{\alpha}(z) = \frac{F(z)}{1 - \alpha F(z)}.$$

Back to our main problem. Denote the Cauchy transform of W and  $W_1$  with respect

to p to be

$$F(z) = \left\langle (W - zI)^{-1}p, p \right\rangle = \int_{\sigma(W)} \frac{d\rho(s)}{s - z};$$
(8.1.3)

$$F_1(z) = \left\langle (W_1 - zI)^{-1}p, p \right\rangle = \int_{\sigma(W_1)} \frac{d\rho_1(s)}{s - z}.$$
(8.1.4)

Applying Proposition 8.1.1 and take  $\alpha = 1$ , we have

$$1 - F = \frac{F}{F_1}.$$
 (8.1.5)

#### 8.1.2 Guess and Proof for Function F

In order to find out the coefficients  $\{a_k\}_{k=1}^{\infty}$  in the scalar spectral measure (8.1.2), we need to give a guess for the function F.

Denote  $\sigma = \{\lambda_k\}_{k=1}^{\infty} \cup \{0\}$  and  $\sigma_1 = \{\mu_k\}_{k=1}^{\infty} \cup \{0\}$ , then we know F(z) has simple poles at  $\{\lambda_k\}_{k=1}^{\infty}$  and analytic at  $\mathbb{C} \setminus \sigma$ ,  $F_1(z)$  has simple poles at  $\{\mu_k\}_{k=1}^{\infty}$  and analytic at  $\mathbb{C} \setminus \sigma_1$ . Thus  $1 - F = \frac{F}{F_1}$  should be a function which is analytic function on  $\mathbb{C} \setminus \sigma$ , and has simple zeros at  $\{\mu_k\}_{k=1}^{\infty}$  and simple poles at  $\{\lambda_k\}_{k=1}^{\infty}$ .

In the following part of this section, we will prove our guess for the function F, which is:

**Proposition 8.1.2.** We have the following equation for function F hold

$$1 - F(z) = \frac{F}{F_1} = \prod_{k=1}^{\infty} \left( \frac{z - \mu_k^2}{z - \lambda_k^2} \right).$$
(8.1.6)

We first prove that the right hand side of (8.1.6) converges uniformly on compact subset of  $\mathbb{C} \setminus \sigma$ .

**Lemma 8.1.3.** 
$$\Phi_N(z) = \prod_{k=1}^N \left( \frac{z - \mu_k^2}{z - \lambda_k^2} \right)$$
 converges uniformly on compact subset of  $\mathbb{C} \setminus \sigma$ .

*Proof.* We use the trivial fact that the convergence of  $\sum_{k=0}^{\infty} |f_k(z) - 1|$  implies the conver-

gence of  $\prod_{k=0}^{\infty} f_k(z)$  (convergence here always means the uniform convergence on compact subset). Since  $\left|1 - \frac{z - \mu_k^2}{z - \lambda_k^2}\right| = \left|\frac{\mu_k^2 - \lambda_k^2}{z - \lambda_k^2}\right|$ , it is sufficient to show that  $\sum_{k=1}^{\infty} \left|\frac{\mu_k^2 - \lambda_k^2}{z - \lambda_k^2}\right|$  converges. For z in a compact  $K \subset \mathbb{C} \setminus \sigma$ , and sufficiently large k (i.e. for all k > N)

$$\left|\frac{\mu_k^2 - \lambda_k^2}{z - \lambda_k^2}\right| \le C(K, N) |\lambda_k^2 - \mu_k^2| \le C(K, N) (\lambda_k^2 - \lambda_{k-1}^2),$$

thus:

$$\sum_{k=N}^{\infty} \left| \frac{\mu_k^2 - \lambda_k^2}{z - \lambda_k^2} \right| \le C(K, N) \lambda_N^2 < \infty,$$

and the convergence follows trivially.

Now denote  $\lim_{N\to\infty} \Phi_N(z) \to \Phi(z)$  on  $\mathbb{C} \setminus \sigma$ . Before we introduce several properties for this function  $\Phi(z)$ , let us recall the definition of Nevanlinna function and its integral representation.

**Definition 8.1.4.** Nevanlinna function is an analytic function on the open half plane and image has non-negative imaginary part.

**Theorem 8.1.5** (Integral representation). Every Nevanlinna function f admits a following integral representation

$$f(z) = C + Dz + \int_{R} \left( \frac{1}{\lambda - z} - \frac{\lambda}{1 + \lambda^{2}} \right) d\mu(\lambda)$$
(8.1.7)

where C is a real constant, D is non-negative,  $\mu$  is a Borel measure on R satisfying  $\int_{R} \frac{d\mu(\lambda)}{1+\lambda^{2}} < \infty.$ 

Conversely, if a function has this type of form, then it's a Nevanlinna function, and the representation is unique.

Now let us get back to discuss function  $\Phi(z)$ .

**Lemma 8.1.6.** Function  $\Phi(z)$  satisfies the following properties:
- (i)  $\lim_{z \to \infty} \Phi(z) = 1;$
- (ii) function  $-\Phi(z)$  is a Nevanlinna function, restricted on  $\mathbb{C}^+$ ;
- (iii)  $\Phi(z)$  is symmetric, i.e.  $\overline{\Phi(z)} = \Phi(\overline{z})$ , in particular,  $\Phi(z)$  is real for all  $x \in \mathbb{R} \setminus \sigma$ ;
- (iv)  $\Phi(z)$  has simple poles at  $\{\lambda_k^2\}_{k=1}^{\infty}$ , simple zeros at  $\{\mu_k^2\}_{k=1}^{\infty}$ .

*Proof.* (i) This is true because

$$\sum_{k=1}^{\infty} \log |\frac{z - \mu_k^2}{z - \lambda_k^2}| \le \sum_{k=1}^{\infty} \frac{|\lambda_k^2 - \mu_k^2|}{|z - \lambda_k^2|} \le \sum_{k=1}^{\infty} \frac{|\lambda_k^2 - \lambda_{k+1}^2|}{|z| - |\lambda_1|^2} \le \frac{|\lambda_1^2|}{|z| - |\lambda_1|^2}$$

goes to 0 when  $z \to \infty$ , hence  $\lim_{z \to \infty} \Phi(z) = 1$ .

(ii) It's equivalent to show that  $\frac{1}{\Phi(z)} = \prod_{i=1}^{\infty} \left( \frac{z - \lambda_i^2}{z - \mu_i^2} \right)$  is a Nevanlinna function on  $\mathbb{C}^+$ . Only need to show that  $0 < \arg \frac{1}{\Phi(z)} < \pi$ .

Denoting Z to be the point z on the complex plane  $\mathbb{C}^+$ , and  $A_1, A_2, ...$  to be the point of sequence  $\{\lambda_n\}$  on the real axis, and  $B_1, ..., B_n, ...$  to be  $\{\mu_n\}$ . Then  $\arg \frac{1}{\Phi(z)}$  is given by

$$0 < \measuredangle B_1 Z A_1 + \dots + \measuredangle B_n Z A_n < \pi,$$

While this is trivially true because  $\{\lambda_n\}$  and  $\{\mu_n\}$  are two interwining sequences, and all those angles don't intersect with each other.

Here (iii) and (iv) is trivial, so we omit it.

Now we prove the following property.

**Lemma 8.1.7.** The function  $\Phi(z)$  given above is the only function that satisfies (i),(ii),(iii), and (iv) in Lemma 8.1.6

*Proof.* Assume that  $\Phi_1(z)$  is another function that satisfies those properties. Denote their ratio to be:  $\Psi := \Phi_1/\Phi$ . Additionally, we have both functions have simple poles at  $\mu_n$  and simple zeros at  $\lambda_n$ , hence  $\Psi(z)$  is analytic and zero-free in  $\mathbb{C} \setminus \{0\}$ .

In addition, we have  $\lim_{z\to\infty} \Psi(z) = 1$  since both  $\Phi(z), \Phi_1(z)$  satisfy (i) in Lemma 8.1.6.

Moreover, for  $x \in \mathbb{R} \setminus \{0\}$  we have  $\Psi(x) > 0$ . Indeed, on  $\mathbb{R} \setminus \sigma \setminus \sigma_1$  functions  $\Phi_1$  and  $\Phi$  are real and have the same sign (If  $\Phi(x), \Phi_1(x)$  have different signs for a certain  $x \in \mathbb{R}/\sigma/\sigma_1$ , then  $\lim_{x \to +\infty} \Phi(x)$  and  $\lim_{x \to +\infty} \Phi(x)$  have different signs, which gives a contradiction to (i) in Lemma 8.1.6), so  $\Psi(x) > 0$  on  $\mathbb{R} \setminus \sigma \setminus \sigma_1$ . Since  $\Psi$  is continuous and zero-free on  $\mathbb{R} \setminus \{0\}$ , this tells us that  $\Psi$  is positive on  $\mathbb{R} \setminus \{0\}$ .

Next, let us notice that  $\Psi(z)$  does not take real negative values. If Im z > 0, then according to (ii) in Lemma 8.1.6,  $\text{Im } \Phi_1(z) < 0$ ,  $\text{Im } \Phi(z) < 0$ , so  $\Psi(z) = \Phi_1(z)/\Phi(z)$ cannot be negative real. If Im z < 0, the symmetry  $\Psi(\overline{z}) = \overline{\Psi(z)}$  implies the same conclusion. And, as we just discussed above, on the real line  $\Psi$  takes positive real values.

So  $\Psi$  omits infinitely many points, therefore by the Picard's Theorem, 0 is not the essential singularity for  $\Psi$ . Trivial analysis shows that 0 cannot be a pole, otherwise  $\frac{1}{\Psi}$  is analytic at 0, which also contradicts to the fact that  $\Psi$  can't take negative real values on real axis. Hence we have 0 is a removable singularity for function  $\Psi$ , and  $\Psi$ is an entire function. By Liouville's Theorem, condition  $\Psi(\infty) = 1$  implies that  $\Psi \equiv 1$ for all  $z \in C$ , hence the lemma is proved.

Remark 8.1.8. For the last paragraph of the proof for Lemma 8.1.7, we can also consider the square root  $\Psi^{1/2}$ , where we take the principal branch of the square root (cut along the negative half-axis). Then  $\operatorname{Re} \Psi(z)^{1/2} \geq 0$ , so by the Casorati–Weierstrass Theorem, 0 cannot be the essential singularity for  $\Psi^{1/2}$ . Again, trivial reasoning shows that 0 cannot be a pole, so again,  $\Psi^{1/2}$  is an entire function. The condition  $\Psi^{1/2}(\infty) = 1$  then implies that  $\Psi^{1/2}(z) \equiv 1$ .

Now back to the proof of Proposition 8.1.2.

Proof of Proposition 8.1.2. At the beginning of this section, we have mentioned that  $\frac{F}{F_1}$  is a function which is analytic on  $\mathbb{C} \setminus \sigma$ , with simple poles at  $\{\lambda_k\}_{k=1}^{\infty}$ , and simple zeros at  $\{\mu_k\}_{k=1}^{\infty}$ , and equals 1 at  $\infty$ , which corresponds to property (i), (iv) in Lemma 8.1.6.

Now we can show that function  $\frac{F}{F_1}$  also satisfies the property (ii), (iii) in Lemma 8.1.6.

In fact, for property (ii), to show that  $\frac{F}{F_1}$  maps  $\mathbb{C}^+$  to  $\mathbb{C}^-$ , according to (8.1.5) it's equivalent to show that F(z) maps  $\mathbb{C}^+$  to  $\mathbb{C}^+$ . This is trivially true, because we have shown that

$$F(z) = \sum_{k \ge 1} \frac{a_k}{\lambda_k^2 - z}$$
(8.1.8)

is analytic on  $\mathbb{C}/\sigma$ , and each single term  $\frac{a_k}{\lambda_k^2-z}$  has positive imaginary part if Im(z) > 0.

As for property (iii), to show  $\frac{F}{F_1}$  is symmetric, it is equivalent to show that F(z) is symmetric, which is also trivially true from (8.1.8).

So far we have shown that  $\frac{F}{F_1}$  and  $\Phi(z)$  are two functions that satisfy all four properties (i), (ii), (iii), (iv) given in Lemma 8.1.6. Hence those two functions coincide, and we proved Proposition 8.1.2.

#### 8.1.3 Coefficients of the Scalar Spectral Measure $\rho(s)$

In this section, we calculate the coefficients for the scalar spectral measure  $\rho(s)$  mentioned in (8.1.1). We will show the following lemma.

**Lemma 8.1.9.** The function  $\Phi(z) = 1 - F(z) = \frac{F(z)}{F_1(z)}$  defined by (8.1.6) can be decomposed as

$$\Phi(z) = 1 - \sum_{n \ge 1} \frac{a_n}{\lambda_n^2 - z},$$

where

$$a_n = (\lambda_n^2 - \mu_n^2) \prod_{k \neq n} \left( \frac{\lambda_n^2 - \mu_k^2}{\lambda_n^2 - \lambda_k^2} \right).$$
(8.1.9)

*Proof.* Consider functions  $\Phi_N$  defined as

$$\Phi_N(z) = \prod_{k=1}^N \left( \frac{z - \mu_k^2}{z - \lambda_k^2} \right)$$

Trivially

$$\Phi_N(z) = 1 - \sum_{n \ge 1} \frac{a_n^N}{\lambda_n^2 - z},$$
(8.1.10)

where

$$a_n^N = (\lambda_n^2 - \mu_n^2) \prod_{\substack{k=1\\k \neq n}}^N \left( \frac{\lambda_n^2 - \mu_k^2}{\lambda_n^2 - \lambda_k^2} \right) > 0 \quad \text{if } n \le N,$$
 (8.1.11)

and  $a_n^N = 0$  if n > N. This is because the two functions on left hand side and right hand side have the same poles and the same residues, so their difference is a polynomial. Then let  $z \to \infty$ , we know that polynomial equals 0 at  $\infty$ . Hence those two functions are equal.

Next, let  $N \to \infty$ . We know, see Lemma 8.1.3 that  $\Phi_N(z)$  converges uniformly to  $\Phi(z)$  in any compact subset  $K \subset \mathbb{C} \setminus \sigma$ . Hence, to prove the lemma it remains to show that  $1 - \sum_{n \geq 1} \frac{a_n^N}{\lambda_n^2 - z}$  converges to  $1 - \sum_{n \geq 1} \frac{a_n}{\lambda_n^2 - z}$  uniformly.

Take z = 0 in (8.1.10). Then we have

$$1 - \sum_{n \ge 1} \frac{a_n^N}{\lambda_n^2} = \prod_{k=1}^N \left(\frac{\mu_k^2}{\lambda_k^2}\right) > 0,$$

so  $\sum_{n \ge 1} \frac{a_n^N}{\lambda_n^2} \le 1.$ 

Notice that for any fixed *n* the sequence  $a_n^N \nearrow a_n$  as  $N \to \infty$ , so  $\sum_{n \ge 1} \frac{a_n}{\lambda_n^2} \le 1$ . Take an arbitrary compact  $K \subset \mathbb{C} \setminus \sigma$ . Clearly for any  $z \in K$ ,

$$\left|\frac{a_n^N}{\lambda_n^2 - z}\right| \le \frac{a_n^N}{\operatorname{dist}(K, \sigma)} \le \frac{a_n}{\operatorname{dist}(K, \sigma)} \le \frac{\lambda_1^2}{\operatorname{dist}(K, \sigma)} \cdot \frac{a_n}{\lambda_n^2}$$

so the condition  $\sum_{n\geq 1} a_n/\lambda_n^2 \leq 1$  implies that  $\sum_{n\geq 1} \frac{a_n^N}{\lambda_n^2-z}$  is uniformly bounded, thus by dominated convergence theorem, we have  $\sum_{n\geq 1} \frac{a_n^N}{\lambda_n^2-z}$  converges uniformly on K, and

$$\lim_{N \to \infty} \sum_{n \ge 1} \frac{a_n^N}{\lambda_n^2 - z} = \sum_{n \ge 1} \lim_{N \to \infty} \frac{a_n^N}{\lambda_n^2 - z} = \sum_{n \ge 1} \frac{a_n}{\lambda_n^2 - z}$$

,

Remark 8.1.10. We also need to calculate the coefficients of  $\rho_1(s)$  (which is the scalar spectral measure of  $W_1$  with respect to p given in (8.1.4)) for a future use in Proposition 10.1.5. However, this is slightly different from calculating  $\rho(s)$  since Ker  $W = \{0\}$  but Ker  $W_1$  can be non-trivial. We will go back to this calculation in Proposition 8.2.9.

### 8.1.4 Existence and uniqueness of the triple $(W, W_1, p)$

*Proof.* Define a measure  $\rho(s)$  as  $\rho = \sum_{k=1}^{\infty} a_k \delta_{\lambda_k^2}$ , where  $\{a_k\}_{k=1}^{\infty}$  is defined by (8.1.9). Let W be the multiplication by independent variable in  $L^2(\rho)$ :

$$(Wf)(t) = tf(t),$$

and let  $p \equiv 1$ . Clearly W is a positive compact operator with simple eigenvalues  $\{\lambda_k^2\}_{k=1}^{\infty}$ , and its Borel transform

$$F(z) = \left\langle (W - zI)^{-1}p, p \right\rangle$$

is given by

$$F(z) = \int_{\mathbb{R}} \frac{d\rho(s)}{s-z} = \sum_{k=1}^{\infty} \frac{a_k}{\lambda_k^2 - z}.$$

Now take  $W_1 = W - pp^*$ , and let  $F_1(z) = ((W_1 - zI)^{-1}p, p)$ . Recall that by Proposition 8.1.1 we have

$$1 - F = \frac{F}{F_1}.$$
 (8.1.12)

Thus applying Lemma 8.1.9, we get

$$1 - F(z) = \prod_{k \ge 1} \left( \frac{z - \mu_k^2}{z - \lambda_k^2} \right) = \frac{F}{F_1}.$$

Together with (8.1.12), we know that  $F_1$  has simple poles exactly at points  $\{\mu_k^2\}_{k\geq 1}$ , so  $\{\mu_k^2\}_{k\geq 1}$  are the non-zero eigenvalues of  $W_1$ . So the existence of the triple  $(W, W_1, p)$  is proved.

The uniqueness follows immediately from Lemma 8.1.7.

# 8.2 Proof of Abstract Borg's Theorem: Trivial Kernel Condition

Recall that in Proposition 3.1.3 and Proposition 5.2.3, we have discussed the equivalent condition for triple  $(R, R_1, p)$  such that the constructed Hankel operator  $\Gamma$  has trivial kernel. That is,

$$||q|| = ||R^{-1}p|| = 1, \qquad q \notin \operatorname{Ran} R.$$
 (8.2.1)

Now comparing equation  $R^2 - R_1^2 = pp^*$  with  $W - W_1 = pp^*$ , we need to derive an equivalent condition for:

$$||W^{-1/2}p|| = 1, \qquad ||W^{-1}p|| = \infty.$$
 (8.2.2)

In this section, we will translate the condition (8.2.1) in terms of  $\{\lambda_n\}_{n=1}^{\infty}$  and  $\{\mu_n\}_{n=1}^{\infty}$ . Then we derive the equivalent condition for Ker  $W_1 = \{0\}$  (we will see that this condition is also closely related to the two identities in (8.2.2)). Finally, we give an expression for the coefficients of  $\rho_1(s) = \sum_{k\geq 1} b_k \delta_{\mu_k^2}(s) + b_0 \delta_0(s)$ , which is defined as the scalar spectral measure of  $W_1$  with respect to p:

$$\langle (W_1 - zI)^{-1}p, p \rangle = \int_{\mathbb{R}} \frac{d\rho_1(s)}{s-z}$$

#### 8.2.1 The Equivalence of Two Identities in Abstract Borg's theorem

**Proposition 8.2.1.** The trivial kernel condition  $||W^{-1/2}p|| = 1$  and  $||W^{-1}p|| = \infty$  is equivalent to the following

(i) 
$$\sum_{j=1}^{\infty} \left( 1 - \frac{\mu_j^2}{\lambda_j^2} \right) = \infty$$
;  
(ii)  $\sum_{j=1}^{\infty} \left( \frac{\mu_j^2}{\lambda_{j+1}^2} - 1 \right) = \infty$ 

*Proof.* We first show that

$$\|P_{Ker(W-\lambda_k^2 I)}p\| = \sqrt{a_k}.$$
(8.2.3)

In fact, since the scalar spectral measure  $\rho(s)$  satisfies that

$$\int_{\mathbb{R}} f(s)\rho(s) = \langle p, f(W)p \rangle$$

for all  $f \in C(\mathbb{R})$ , thus we take a sequence of polynomials  $\{f_n\}$  that converges to  $\mathbb{1}_{\{x=\lambda_k^2\}}$ , then we have

$$a_k = \langle p, P_{\operatorname{Ker}(W - \lambda_k^2 I)} p \rangle = \| P_{\operatorname{Ker}(W - \lambda_k^2 I)} p \|^2$$

Then from (8.2.3), condition  $||W^{-\frac{1}{2}}p|| = 1$  can be written as

$$\sum_{k=1}^{\infty} \frac{a_k}{\lambda_k^2} = 1.$$
(8.2.4)

Now recall equation (8.1.6)

$$\prod_{k=1}^{\infty} \left( \frac{z - \mu_k^2}{z - \lambda_k^2} \right) = 1 - \sum_{k=1}^{\infty} \frac{a_k}{\lambda_k^2 - z},$$
(8.2.5)

we take real  $z < 0, z \rightarrow 0-$ , then by (8.2.5) and monotone convergence theorem, we have

$$1 - \sum_{k=1}^{\infty} \frac{a_k}{\lambda_k^2} = \prod_{k=1}^{\infty} \frac{\mu_k^2}{\lambda_k^2}.$$
 (8.2.6)

Hence  $\sum_{k=1}^{\infty} \frac{a_k}{\lambda_k^2} = 1$  is equivalent to

$$\prod_{k=1}^{\infty} \frac{\mu_i^2}{\lambda_i^2} = 0, \tag{8.2.7}$$

which is also equivalent to:

$$\sum_{k=1}^{\infty} \left(\frac{\mu_k^2}{\lambda_k^2} - 1\right) = -\infty, \qquad (8.2.8)$$

gives us the condition for (9.1.2).

Now for the second condition  $||W^{-1}p|| = \infty$ , using (8.2.3) again, it's equivalent to

$$\sum_{k=1}^{\infty} \frac{a_k}{\lambda_k^4} = \infty. \tag{8.2.9}$$

We rewrite (8.1.6) as

$$\prod_{k=1}^{\infty} \left( \frac{z - \mu_k^2}{z - \lambda_k^2} \right) = 1 - \sum_{k=1}^{\infty} a_k \frac{(\lambda_k^2 - z) + z}{\lambda_k^2 (\lambda_k^2 - z)}$$
$$= 1 - \sum_{k \ge 1} \frac{a_k}{\lambda_k^2} - \sum_{k \ge 1} \frac{a_k z}{\lambda_k^2 (\lambda_k^2 - z)}$$

assuming that we already have  $\sum_{k=1}^{\infty} \frac{a_k}{\lambda_k^2} = 1$ , then we have

$$-\frac{1}{z}\prod_{k=1}^{\infty} \left(\frac{z-\mu_k^2}{z-\lambda_k^2}\right) = \sum_{k=1}^{\infty} \frac{a_k}{\lambda_k^2(\lambda_k^2-z)}.$$
 (8.2.10)

Now denoting the function of z in (8.2.10) as G(z). We apply  $z = -\lambda_N^2$  to G(z), and let  $N \to \infty$ , then RHS of (8.2.10) increases monotonically to  $\sum_{k=1}^{\infty} a_k/\lambda_k^4$ . Therefore condition (8.2.9) is equivalent to

$$\lim_{N \to \infty} G(-\lambda_N^2) = \lim_{N \to \infty} \frac{1}{\lambda_N^2} \prod_{k=1}^{\infty} \left( \frac{\lambda_N^2 + \mu_k^2}{\lambda_N^2 + \lambda_k^2} \right) = \infty.$$
(8.2.11)

Denote  $G_N(z) = \frac{1}{\lambda_N^2} \prod_{k=1}^{N-1} \left( \frac{z - \mu_k^2}{z - \lambda_k^2} \right)$ , we will prove the following lemma.

Lemma 8.2.2. The following statements are equivalent:

- (i)  $\lim_{N \to \infty} G(-\lambda_N^2) = \infty$ ;
- (ii)  $\lim_{N \to \infty} G_N(-\lambda_N^2) = \infty;$

(iii) 
$$\lim_{N \to \infty} G_N(0) = \infty$$
.

*Proof of Lemma 8.2.2.* (i)  $\iff$  (ii): To prove this equivalence, it is sufficient to show that

$$0 < C_1 \le \prod_{k=N}^{\infty} \left( \frac{\lambda_N^2 + \mu_k^2}{\lambda_N^2 + \lambda_k^2} \right) \le C_2 < \infty$$
(8.2.12)

with constants  $C_1, C_2$  independent of N. Note that for  $k \ge N$ , we trivially have

$$\frac{1}{2} \le \frac{\lambda_N^2 + \mu_k^2}{\lambda_N^2 + \lambda_k^2} \le 1.$$
(8.2.13)

Here the upper bound trivially implies the upper bound in (8.2.12) with  $C_2 = 1$ .

To get the lower bound in (8.2.12), we use the estimate (8.2.13) and the following inequality

$$\ln x \ge (\ln 2)(x-1), \quad \forall x \in [1/2, 1].$$

Thus

$$\sum_{k=N}^{\infty} \ln 2\left(\frac{\lambda_N^2 + \mu_k^2}{\lambda_N^2 + \lambda_k^2} - 1\right) = \sum_{k=N}^{\infty} -\ln 2\frac{\lambda_k^2 - \mu_k^2}{\lambda_N^2 + \lambda_k^2} \ge (-\ln 2)\sum_{k=N}^{\infty} \frac{\lambda_k^2 - \mu_k^2}{\lambda_N^2} \ge -\ln 2,$$

we see that the lower bound in (8.2.12) holds with  $C_1 = \frac{1}{2}$ .

(ii)  $\iff$  (iii): The function  $G_N$  is analytic and zero free in the disc  $D_N$  of radius  $2\lambda_N^2$  centered at  $-\lambda_N^2$ , since we have  $\lambda_N^2$  is smaller than  $\mu_{N-1}^2$  and so the function  $\ln |G_N(z)|$  is harmonic in this disc.

Now we can find a sufficiently large N, such that  $\ln |G_n(z)|$  is non-negative on the disk  $D_n$  when  $n \ge N$ . In fact, if we goes from (ii) to (iii), then we find a sufficiently

large N, such that  $|G_n(-\lambda_n^2)| > 1$  when  $n \ge N$ . Then applying the maximum principle to function  $\ln |G_n(z)|$ , then we have  $|G_n(z)| > 1$  will be hold on the whole disk Clos  $D_n$ . Similarly, we can also assume  $\ln |G_n(z)|$  is non-negative when going from (iii) to (ii).

Now applying Harnack inequality for function  $\ln |G_N(z)|$  on the disk  $D_N$  (see Remark 8.2.3)

$$\frac{1}{3}\ln|G_N(-\lambda_N^2)| \le \ln|G_N(0)| \le 3\ln|G_N(-\lambda_N^2)|, \qquad (8.2.14)$$

which proves the equivalence (ii)  $\iff$  (iii).

Now get back to the equation (8.2.11). By lemma 8.2.2, condition (8.2.9) is equivalent to  $\lim_{N\to\infty} G_N(0) = \infty$ , hence

$$\lim_{N \to \infty} \frac{1}{\lambda_1^2} \prod_{k=1}^{N-1} \frac{\mu_k^2}{\lambda_{k+1}^2} = \lim_{N \to \infty} \frac{1}{\lambda_N^2} \prod_{k=1}^{N-1} \frac{\mu_k^2}{\lambda_k^2} = \infty,$$

And condition  $\prod_{k=1}^{\infty} \frac{\mu_k^2}{\lambda_{k+1}^2} = \infty$  is equivalent to

$$\sum_{k=1}^{\infty} \left( \frac{\mu_k^2}{\lambda_{k+1}^2} - 1 \right) = \infty$$
 (8.2.15)

Thus (8.2.8) together with (8.2.15) finish our proof for Proposition 8.2.1.

*Remark* 8.2.3. Inside the proof of Lemma 8.2.2, we use Harnack inequality (see [35]) for (8.2.14), which can be stated as below:

Theorem 8.2.4 (Harnack inequality). Let f be a non-negative function defined on a closed ball  $B(x_0, R)$ . If f is continuous on the closed ball and harmonic on its interior, then for every point x with  $|x - x_0| = r < R$ , we have

$$\frac{1 - (r/R)}{[1 + (r/R)]^{n-1}} f(x_0) \le f(x) \le \frac{1 + (r/R)}{[1 - (r/R)]^{n-1}} f(x_0)$$

#### 8.2.2 Trivial Kernel Condition for $W_1$ and the Spectral Measure $\rho_1(s)$

With the constructed triple  $(W, W_1, p)$  in Theorem 8.0.1, we discuss the trivial kernel condition of  $W_1$  in this subsection, and in addition, we give an expression for the coefficients of  $\rho_1(s) = \sum_{k\geq 1} b_k \delta_{\mu_k^2}(s) + b_0 \delta_0(s)$ , which is defined as the scalar spectral measure of  $W_1$  with respect to p:

$$\langle (W_1 - zI)^{-1}p, p \rangle = \int_{\mathbb{R}} \frac{d\rho_1(s)}{s-z}.$$

**Proposition 8.2.5.** If the coefficients  $\{a_n\}_{n\geq 1}$  given in Lemma 8.1.9 satisfies  $\sum_{n\geq 1} \frac{a_n}{\lambda_n^2} < 1$ , then  $b_0 = 0$ , and the coefficients  $\{b_n\}_{n\geq 1}$  can be written as:

$$b_n = (\lambda_n^2 - \mu_n^2) \prod_{k \neq n} \left( \frac{\mu_n^2 - \lambda_k^2}{\mu_n^2 - \mu_k^2} \right)$$
(8.2.16)

*Proof.* From (8.1.5) we have

$$1 + F_1(z) = \frac{F_1}{F} = \prod_{i=1}^{\infty} \left( \frac{z - \lambda_i^2}{z - \mu_i^2} \right).$$

If  $\prod_{i=1}^{\infty} \left(\frac{\lambda_i^2}{\mu_i^2}\right) < \infty$  (thus by (8.2.6) we have  $\sum_{n \ge 1} \frac{a_n}{\lambda_n^2} < 1$ ), then follow a similar procedure as the proof in Lemma 8.1.9, we write

$$1 + \sum_{n} \frac{b_{n}^{N}}{\mu_{n}^{2} - z} = \prod_{k=1}^{N} \left( \frac{z - \lambda_{k}^{2}}{z - \mu_{k}^{2}} \right),$$

then we have

$$b_n^N = (\lambda_n^2 - \mu_n^2) \prod_{k=1, k \neq n}^N \left( \frac{\mu_n^2 - \lambda_k^2}{\mu_n^2 - \mu_k^2} \right) \text{ if } n \ge N,$$

and  $b_n^N = 0$  if n > N.

Now take a upper bound B for  $\prod_{i=1}^{\infty} \left(\frac{\lambda_i^2}{\mu_i^2}\right)$ , then we have

$$\sum_{n} \frac{b_n^N}{\mu_n^2} < B - 1$$

Since  $b_n^N > 0$  and  $b_n^N \searrow b_n$  as  $N \to \infty$ , where  $b_n$  has the formula as

$$b_n = (\lambda_n^2 - \mu_n^2) \prod_{k \neq n} \left( \frac{\mu_n^2 - \lambda_k^2}{\mu_n^2 - \mu_k^2} \right),$$

we have  $\sum_{n\geq 1} \frac{b_n}{\mu_n^2} < B-1$  and  $\sum_{n\geq 1} \frac{b_n^N}{\mu_n^2-z}$  converges uniformly to  $\sum_{n\geq 1} \frac{b_n}{\mu_n^2-z}$  on any compact  $K \subset \sigma$ .

Thus in this case we can write  $\rho_1(s) = \sum_n b_n \delta_{\mu_k^2}(z)$  where  $b_n$  has the form in (8.2.16).

**Proposition 8.2.6.** For the triple  $(W, W_1, p)$  constructed in Theorem 8.0.1, we have Ker  $W_1 \neq \{0\}$  iff the following two equations hold:

$$\sum_{n\geq 1} \frac{a_n}{\lambda_n^2} = 1, \qquad \sum_{n\geq 1} \frac{a_n}{\lambda_n^4} < \infty.$$
(8.2.17)

Hence by Proposition 8.2.1, (8.2.17) is also equivalent to:

$$\prod_{k=1}^{\infty} \left(\frac{\lambda_k^2}{\mu_k^2}\right) = \infty, \qquad \prod_{k=1}^{\infty} \left(\frac{\mu_k^2}{\lambda_{k+1}^2}\right) = \infty.$$
(8.2.18)

*Proof.* The proof is very much similar to the proof in Proposition 3.1.3.

We define two positive self-adjoint operators  $R, R_1$  as:  $R := W^{1/2}, R_1 = W_1^{1/2}$ , and also a contraction  $\mathcal{T} := R_1 R^{-1}$  implied by Douglas Lemma 3.1.1, and also a vector  $q := R^{-1}p$ . Then the two equations in (8.2.17) are equivalent to:

$$||q|| = 1, \qquad q \in \operatorname{Ran} R$$

(i) Under the condition that ||q|| < 1, we have shown in Proposition 8.2.5, that Ker  $W_1 = \{0\}$ ;

(ii) Suppose that ||q|| = 1 and  $q \notin \operatorname{Ran} R$ . Since we have

$$W_1 = R_1^2 = RT^*TR = R(I - qq^*)R$$
(8.2.19)

Since  $\operatorname{Ker}(I - qq^*) = \operatorname{Span}\{q\}$  and  $q \notin \operatorname{Ran} R$ , we can see that  $\operatorname{Ker} R(I - qq^*)R = \{0\}$ , thus  $\operatorname{Ker} W_1 = \{0\}$ .

(iii) Suppose that ||q|| = 1 and  $q \in \operatorname{Ran} R$ . Using (8.2.19) again, and apply  $u := R^{-1}q$  to (8.2.19), we have  $(I - qq^*)Ru = 0$ , thus  $W_1u = 0$ .

Now in contrast with the case when  $\sum_{k\geq 1} \frac{a_k}{\lambda_k^2} < 1$ , we consider the case when  $\sum_{k\geq 1} \frac{a_k}{\lambda_k^2} = 1$ . We first prove the following lemma:

**Lemma 8.2.7.** For all  $z \notin \{\frac{1}{\mu_j^2}\}_{j \ge 1}$ , we have the following equation

$$\prod_{j=1}^{\infty} \frac{1 - z\lambda_j^2}{1 - z\mu_j^2} = 1 - b_0 z + \sum_{k \ge 1} \frac{b_k z}{\mu_k^2 z - 1}$$

is well-defined and holds for some constant  $b_0 \in \mathbb{C}$ . Here coefficients  $\{b_k\}_{k\geq 1}$  has the expression in (8.2.16).

*Proof of Lemma 8.2.7.* The first step is similar to the proof of Proposition 8.1.9. We consider rewriting a finite product

$$\prod_{j=1}^{N} \frac{1 - z\lambda_j^2}{1 - z\mu_j^2} = 1 + \sum_{k=1}^{N} \frac{b_k^N}{\mu_k^2 - \frac{1}{z}} = 1 + \sum_{k=1}^{N} \frac{b_k^N z}{\mu_k^2 z - 1},$$
(8.2.20)

then we have the coefficients  $\boldsymbol{b}_k^N$  has the expression

$$b_k^N = (\lambda_k^2 - \mu_k^2) \prod_{n=1, n \neq k}^N \left( \frac{\mu_k^2 - \lambda_n^2}{\mu_k^2 - \mu_n^2} \right),$$

here  $b_k^N > 0$  and  $b_n^N \searrow b_n$  as  $N \to \infty$ .

Now we rewrite (8.2.20) and define function  $H_N(z)$  as:

$$H_N(z) := \frac{1}{z} \Big( \prod_{j=1}^N \frac{1 - z\lambda_j^2}{1 - z\mu_j^2} - 1 \Big) = \sum_{k=1}^N \frac{b_k^N}{\mu_k^2 z - 1}.$$
(8.2.21)

Let z = 0, and we take the expansion of infinite product at z = 0 on LHS, we have the sum

$$\sum_{k=1}^{N} b_k^N = \sum_{j=1}^{N} (\lambda_j^2 - \mu_j^2)$$

is bounded from above, thus let  $N \to \infty$ , we have  $\sum_{k=1}^{\infty} b_k < \infty$ . Now again for the function  $H_N(z)$  defined in (8.2.21), we write

$$H_N(z) = H_N(0) + \int_0^z H'_N(t)dt.$$
(8.2.22)

Here  $H'_N(t)$  has the expression

$$H'_N(t) = \sum_{j=1}^N \frac{-b_j^N \mu_j^2}{(\mu_j^2 t - 1)^2}.$$

Thus for a compact set K which doesn't intersect with any of  $\{\frac{1}{\mu_j^2}\}_{j\geq 1}$  and any  $t\in K$ , we have

$$\big|\frac{-b_j^N\mu_j^2}{(\mu_j^2t-1)^2}\big| \le C(K,\sigma)b_j^N.$$

Since  $\sum_{j\geq 1} b_j < \infty$ , we have

$$\sum_{j=1}^{N} \frac{-b_j^N \mu_j^2}{(\mu_j^2 t - 1)^2} \to \sum_{j=1}^{\infty} \frac{-b_j \mu_j^2}{(\mu_j^2 t - 1)^2}$$

uniformly on K. Hence we take  $N \to \infty$  in (8.2.22), we get

$$\frac{1}{z} \left(\prod_{j=1}^{\infty} \frac{1-z\lambda_j^2}{1-z\mu_j^2} - 1\right) = \lim_{N \to \infty} H_N(0) + \int_0^z \sum_{j=1}^{\infty} \frac{-b_j \mu_j^2}{(\mu_j^2 t - 1)^2}$$

Hence

$$\frac{1}{z} \left(\prod_{j=1}^{\infty} \frac{1 - z\lambda_j^2}{1 - z\mu_j^2} - 1\right) = \sum_{j=1}^{\infty} \frac{b_j}{\mu_k^2 z - 1} + C$$

for some constant  $C \in \mathbb{C}$ . And we finish the proof of lemma 8.2.7.

With Lemma 8.2.7, we replace z by  $\frac{1}{z}$ , we get that for  $z \notin {\{\mu_j^2\}_{j\geq 1}}$ , we have the following equation

$$\prod_{j=1}^{\infty} \frac{\lambda_j^2 - z}{\mu_j^2 - z} = 1 - b_0 \frac{1}{z} + \sum_{k \ge 1} \frac{b_k}{\mu_k^2 - z},$$

where  $\{b_k\}_{j\geq 1}$  has the form as (8.2.16). The only remaining thing to do is to find the value of  $b_0$ .

**Lemma 8.2.8.** With the assumptions that  $\sum_{k\geq 1} \frac{a_k}{\lambda_k^2} = 1$  and  $\sum_{k\geq 1} \frac{a_k}{\lambda_k^4} < \infty$ , we have

$$b_0 = \big(\sum_{k\ge 1} \frac{a_k}{\lambda_k^4}\big)^{-1}$$

*Proof.* If we denote the scalar spectral measure  $\rho_1(s) = \sum_{k \ge 1} \delta_{\mu_k^2}(s) + b_0 \delta_0(s)$ , and substitute it into

$$1 + F_1(z) = \frac{F_1(z)}{F(z)} = \prod_{k=1}^{\infty} \left(\frac{z - \lambda_k^2}{z - \mu_k^2}\right) = 1 + \sum_{k \ge 1} \frac{b_k}{\mu_k^2 - z} - \frac{b_0}{z}.$$

Hence we have

$$b_0 = \lim_{z \to 0} \left[ -z \prod_{k=1}^{\infty} \left( \frac{z - \lambda_k^2}{z - \mu_k^2} \right) \right]$$

On the other hand, from (8.2.10) we have

$$-\frac{1}{z}\prod_{k=1}^{\infty} \left(\frac{z-\mu_k^2}{z-\lambda_k^2}\right) = \sum_{k=1}^{\infty} \frac{a_k}{\lambda_k^2(\lambda_k^2-z)}.$$
 (8.2.23)

Let  $z \to 0$ , then we get  $\frac{1}{b_0} = \sum_{k \ge 1} \frac{a_k}{\lambda_k^4}$ , hence we finish the proof of Lemma 8.2.8.  $\Box$ 

Thus combining the results we get in Proposition 8.2.5, Lemma 8.2.7 and Lemma 8.2.8, we can give an expression for all coefficients of  $\rho_1(s)$ .

**Proposition 8.2.9.** Denote the scalar spectral measure of  $W_1$  with respect to p defined by (8.1.4) to be  $\rho_1(s)$ . Then  $\rho_1(s) = \sum_{k\geq 1} b_k \delta_{\mu_k^2}(s) + b_0 \delta_0(s)$  where:

$$b_n = (\lambda_n^2 - \mu_n^2) \prod_{k \neq n} \left( \frac{\mu_n^2 - \lambda_k^2}{\mu_n^2 - \mu_k^2} \right), \qquad \text{when } n \ge 1$$

and

$$b_{0} = \begin{cases} 0 & \text{if } \sum_{n} \frac{a_{n}}{\lambda_{n}^{2}} < 1, \text{ or } \sum_{n} \frac{a_{n}}{\lambda_{n}^{2}} = 1 \text{ and } \sum_{n} \frac{a_{n}}{\lambda_{n}^{4}} = \infty. \\ \left(\sum_{n} \frac{a_{n}}{\lambda_{n}^{4}}\right)^{-1} & \text{if } \sum_{n} \frac{a_{n}}{\lambda_{n}^{2}} = 1 \text{ and } \sum_{n} \frac{a_{n}}{\lambda_{n}^{4}} < \infty. \end{cases}$$
(8.2.24)

We can show that the coefficients  $\{a_k\}_{k\geq 1}$  in (8.1.9), and the coefficients  $\{b_k\}_{k\geq 0}$  in Proposition 8.2.9 satisfy the following identities.

**Proposition 8.2.10** (Modified from Corollary 1, [36]). For all  $m, p \ge 1$ , we have

$$\sum_{j\geq 1} \frac{a_j}{\lambda_j^2 - \mu_m^2} = 1, \tag{8.2.25}$$

$$\sum_{j\geq 1} \frac{b_j}{\lambda_m^2 - \mu_j^2} = 1 - \frac{b_0}{\lambda_m^2},\tag{8.2.26}$$

$$\sum_{j\geq 1} \frac{a_j}{(\lambda_j^2 - \mu_m^2)(\lambda_j^2 - \mu_p^2)} = \frac{\delta_{mp}}{b_m},$$
(8.2.27)

$$\sum_{j\geq 1} \frac{b_j}{(\mu_j^2 - \lambda_m^2)(\mu_j^2 - \lambda_p^2)} = \frac{\delta_{mp}}{a_m} - \frac{b_0}{\lambda_m^2 \lambda_p^2}.$$
 (8.2.28)

Here  $\delta_{mp} = 1$  if m = p, and equals 0 otherwise.

Proof of Proposition 8.2.10. Recall that in previous calculation, the coefficients  $\{a_k\}_{k\geq 0}$ ,  $\{b_k\}_{k\geq 0}$  is given by

$$1 - \sum_{k \ge 1} \frac{a_k}{\lambda_k^2 - z} = \prod_{k \ge 1} \left( \frac{z - \mu_k^2}{z - \lambda_k^2} \right)$$
(8.2.29)

$$1 + \sum_{k \ge 1} \frac{b_k}{\mu_k^2 - z} - \frac{b_0}{z} = \prod_{k \ge 1} \left(\frac{z - \lambda_k^2}{z - \mu_k^2}\right)$$
(8.2.30)

Then (8.2.25) can be achieved by setting  $z = \mu_m^2$  in (8.2.29), and (8.2.26) can be achieved by setting  $z = \lambda_m^2$  in (8.2.30).

For the case of  $m \neq p$  in (8.2.27), we have

$$\sum_{j\geq 1} \frac{a_j}{(\lambda_j^2 - \mu_m^2)(\lambda_j^2 - \mu_p^2)}$$
  
=  $\sum_{j\geq 1} a_j \Big[ \frac{1}{\lambda_j^2 - \mu_m^2} - \frac{1}{\lambda_j^2 - \mu_p^2} \Big] \frac{1}{\mu_m^2 - \mu_p^2}$   
=  $\frac{1}{\mu_m^2 - \mu_p^2} \Big[ \sum_{j\geq 1} \frac{a_j}{\lambda_j^2 - \mu_m^2} - \sum_{j\geq 1} \frac{a_j}{\lambda_j^2 - \mu_p^2} \Big] = 0.$ 

Here the last identity follows from (8.2.25).

For the case of m = p in (8.2.27), we take the differentiation on both sides of (8.2.29), and then we have

$$\sum_{k\geq 1} -\frac{a_k}{(\lambda_k^2 - z)^2} = \sum_{k\geq 1} \frac{\mu_k^2 - \lambda_k^2}{(z - \lambda_k^2)^2} \prod_{n\neq k} \left(\frac{z - \mu_n^2}{z - \lambda_n^2}\right).$$

By setting  $z = \mu_m^2$ , we get

$$\sum_{k\geq 1} \frac{a_k}{(\lambda_k^2 - \mu_m^2)^2} = \frac{1}{\lambda_m^2 - \mu_m^2} \prod_{n\neq m} \left(\frac{\mu_m^2 - \mu_n^2}{\mu_m^2 - \lambda_n^2}\right) = \frac{1}{b_m}.$$

For the case of  $m \neq p$  in (8.2.28), we have

$$\begin{split} &\sum_{j\geq 1} \frac{b_j}{(\mu_j^2 - \lambda_m^2)(\mu_j^2 - \lambda_p^2)} \\ &= \sum_{j\geq 1} b_j \Big[ \frac{1}{\mu_j^2 - \lambda_m^2} - \frac{1}{\mu_j^2 - \lambda_p^2} \Big] \frac{1}{\lambda_m^2 - \lambda_p^2} \\ &= \frac{1}{\lambda_m^2 - \lambda_p^2} \Big[ \sum_{j\geq 1} \frac{b_j}{\mu_j^2 - \lambda_m^2} - \sum_{j\geq 1} \frac{b_j}{\mu_j^2 - \lambda_p^2} \Big] \\ &= \frac{1}{\lambda_m^2 - \lambda_p^2} \Big[ (\frac{b_0}{\lambda_m^2} - 1) - (\frac{b_0}{\lambda_p^2} - 1) \Big] = -\frac{b_0}{\lambda_m^2 \lambda_p^2} \end{split}$$

Here the third identity follows from (8.2.26).

For the case of m = p in (8.2.28), we take the differentiation on both sides of (8.2.26), and then we get

$$\sum_{k\geq 1} \frac{b_k}{(\mu_k^2 - z)^2} + \frac{b_0}{z^2} = \sum_{k\geq 1} \frac{\lambda_k^2 - \mu_k^2}{(z - \mu_k^2)^2} \prod_{n\neq k} \left(\frac{z - \lambda_n^2}{z - \mu_n^2}\right).$$

By taking  $z = \lambda_m^2$ , we get

$$\sum_{k\geq 1} \frac{b_k}{(\mu_k^2 - \lambda_m^2)^2} = \frac{1}{\lambda_m^2 - \mu_m^2} \prod_{n\neq m} \left(\frac{\lambda_m^2 - \lambda_n^2}{\lambda_m^2 - \mu_n^2}\right) - \frac{b_0}{\lambda_m^4} = \frac{1}{a_m} - \frac{b_0}{\lambda_m^4}$$

c		

### 8.3 Abstract Borg's Theorem: Finite Rank Case

Similar to the description in Theorem 8.0.1, we can also write the finite rank version of Abstract Borg's Theorem, which can be stated as below:

**Theorem 8.3.1** (Abstract Borg's Theorem, Finite Rank Version). Given two sequences  $\{\lambda_k^2\}_{k=1}^N$ ,  $\{\mu_k^2\}_{k=1}^N$  satisfy the following intertwining relation

$$\lambda_1^2 > \mu_1^2 > \dots > \lambda_N^2 > \mu_N^2 \ge 0,$$

then there exists an unique (up to unitary equivalence) triple  $(W, W_1, p)$ , such that W, W<sub>1</sub> are positive self-adjoint compact operators with trivial kernel, and also satisfy  $W_1 = W - pp^*$ . In addition, the eigenvalues of W, W<sub>1</sub> are simple, and coincide with  $\{\lambda_k^2\}_{k=1}^N$ ,  $\{\mu_k^2\}_{k=1}^N$  respectively.

Remark 8.3.2. If  $\mu_N^2 = 0$ , then we have the constructed operator  $W_1$  has a kernel of dimension 1; if  $\mu_N^2 > 0$ , then we have the constructed operator  $W_1$  also has trivial kernel.

*Remark* 8.3.3. The proof for this Abstract Borg's theorem in finite-rank version is trivial, because the infinite product  $\prod_{k=1}^{\infty} \left(\frac{z-\mu_k^2}{z-\lambda_k^2}\right)$  now becomes a finite product  $\prod_{k=1}^{N} \left(\frac{z-\mu_k^2}{z-\lambda_k^2}\right)$ .

Under the setting in Theorem 8.3.1, if we write down the scalar spectral measure of  $W, W_1$  with respect to p:

$$F(z) = \left\langle (W - zI)^{-1}p, p \right\rangle = \int_{\sigma(W)} \frac{d\rho(s)}{s - z},$$
  
$$F_1(z) = \left\langle (W_1 - zI)^{-1}p, p \right\rangle = \int_{\sigma(W_1)} \frac{d\rho_1(s)}{s - z},$$

we still have  $1 - F = \frac{F}{F_1}$ , and we have the coefficients of  $\rho(s) = \sum_{n=1}^{N} a_n \delta_{\lambda_n^2}(s)$ ,  $\rho_1(s) = \sum_{n=1}^{N} b_n \delta_{\mu_n^2}(s)$  has the following representation:

$$a_{n} = (\lambda_{n}^{2} - \mu_{n}^{2}) \prod_{k \neq n} \left( \frac{\lambda_{n}^{2} - \mu_{k}^{2}}{\lambda_{n}^{2} - \lambda_{k}^{2}} \right),$$
$$b_{n} = (\lambda_{n}^{2} - \mu_{n}^{2}) \prod_{k \neq n} \left( \frac{\mu_{n}^{2} - \lambda_{k}^{2}}{\mu_{n}^{2} - \mu_{k}^{2}} \right).$$

Under this case, we have

$$||W^{-1/2}p|| = 1 \qquad \Leftrightarrow \qquad \sum_{k=1}^{N} \frac{a_k}{\lambda_k^2} = 1 \qquad \Leftrightarrow \qquad \mu_N = 0;$$

as for the second identity  $||W^{-1}p|| = \infty$  in the trivial kernel condition, it can't be true

because  $\sum_{k=1}^{N} \frac{a_k}{\lambda_k^4}$  is always finite.

### 8.4 Proof of Generalized Abstract Borg's Theorem

*Proof.* First by using the result of Abstract Borg Theorem 8.0.1, we can find a triple  $(W, W_1, p)$  on a Hilbert space  $\widetilde{\mathcal{H}}$ , such that

- (i)  $W, W_1$  are self-adjoint compact, positive operators. Ker  $W = \{0\}$ ;
- (ii)  $W W_1 = pp^*;$
- (iii) p is cyclic with respect to W on  $\widetilde{\mathcal{H}}$ ;
- (iv)  $\{\lambda_k^2\}_{k=1}^{\infty}, \{\mu_k^2\}_{k=1}^{\infty}$  are the simple eigevalues of  $W, W_1$  respectively.

Now define operators  $\widetilde{R}, \widetilde{R}_1$  as  $\widetilde{R} = W^{1/2}, \widetilde{R}_1 = W_1^{1/2}$ , we have  $\widetilde{R}, \widetilde{R}_1$  have simple eigenvalues  $\{\lambda_k\}_{k=1}^{\infty}, \{\mu_k\}_{k=1}^{\infty}$  respectively.

Now we define an extended Hilbert space

$$\mathcal{H} = \mathcal{H} \oplus \mathcal{H}_1 \oplus \mathcal{H}_2 \oplus ... \mathcal{H}_{2k-1} \oplus \mathcal{H}_{2k}...$$

Here  $\mathcal{H}_{2k-1}$  is a Hilbert space with dimension  $m(\lambda_k) - 1$ ,  $\mathcal{H}_{2k}$  is a Hilbert space with dimension  $m(\mu_k)$ .

And we also extend  $\widetilde{R}, \widetilde{R}_1$  to two operators  $R, R_1$  defined on  $\mathcal{H}$ . We let

- (i) For all  $k \in \mathbb{N}$ , We take  $R = R_1 = \lambda_k I$  on  $\mathcal{H}_{2k-1}$ ;
- (ii) For all  $k \in \mathbb{N}$ , we take  $R = R_1 = \mu_k I$  on  $\mathcal{H}_{2k}$ .

From (i) and (ii), we can see

$$\dim \operatorname{Ker}(R - \lambda_k I) = \dim \mathcal{H}_{2k-1} + 1 = m(\lambda_k);$$
$$\dim \operatorname{Ker}(R_1 - \lambda_k I) = \dim \mathcal{H}_{2k-1} = m(\lambda_k) - 1;$$
$$\dim \operatorname{Ker}(R - \mu_k I) = \dim \mathcal{H}_{2k} = m(\mu_k);$$
$$\dim \operatorname{Ker}(R_1 - \mu_k I) = \dim \mathcal{H}_{2k} + 1 = m(\mu_k) + 1;$$

Thus  $(R, R_1, p)$  is a satisfied triple on  $\mathcal{H}$ .

Now we move on to the uniqueness part. Assume that  $(W, W_1, p)$ ,  $(W', W'_1, p')$  are two satisfied tuples defined on spaces  $\mathcal{H}, \mathcal{H}'$  respectively. Then we denote

$$\mathcal{H}_{0} := \overline{\operatorname{Span}} \{ W^{n} p | n \geq 0 \};$$
  
$$\mathcal{E}_{\lambda_{k}} := \operatorname{Ker}(W - \lambda^{2}I), \qquad \mathcal{E}_{\mu_{k}} := \operatorname{Ker}(W - \mu_{k}^{2}I);$$
  
$$\mathcal{E}_{\lambda_{k}}^{1} := \operatorname{Ker}(W_{1} - \lambda_{k}^{2}I), \qquad \mathcal{E}_{\mu_{k}}^{1} := \operatorname{Ker}(W_{1} - \mu_{k}^{2}I);$$
  
$$\widetilde{W} = W \big|_{\mathcal{H}_{0}}, \qquad \widetilde{W}_{1} = W_{1} \big|_{\mathcal{H}_{0}}$$

And we define  $\mathcal{H}'_0, \mathfrak{E}_{\lambda_k}, \mathfrak{E}_{\mu_k}, \mathfrak{E}^1_{\lambda_k}, \mathfrak{E}^1_{\mu_k}, \widetilde{W}', \widetilde{W}'_1$  similarly.

Then followed by Proposition 6.2.10, we have  $(\widetilde{W}, \widetilde{W}_1, p)$  and  $(\widetilde{W}', \widetilde{W}'_1, p')$  are two tuples that have simple eigenvalues  $\{\lambda_k^2\}_{k=1}^{\infty}$ ,  $\{\mu_k^2\}_{k=1}^{\infty}$  respectively, which also satisfy all requirements given in Theorem 8.0.1, Hence applying Theorem 8.0.1, we have those two tuples are unitary equivalent, i.e.

$$\widetilde{W}' = \widetilde{V}\widetilde{W}\widetilde{V}^*, \qquad \widetilde{W}_1 = \widetilde{V}\widetilde{W}_1\widetilde{V}^*, \qquad p' = \widetilde{V}p$$

for an unitary  $\widetilde{V}: \mathcal{H}_0 \to \widetilde{\mathcal{H}}_0$ .

Now using Proposition 6.2.10 again, we have

$$\mathcal{H} \ominus \mathcal{H}_0 = \left( \bigoplus_{k=1}^{\infty} \mathcal{E}^1_{\lambda_k} 
ight) \bigoplus \left( \bigoplus_{k=1}^{\infty} \mathcal{E}_{\mu_k} 
ight) \ \mathcal{H}' \ominus \mathcal{H}'_0 = \left( \bigoplus_{k=1}^{\infty} \mathfrak{E}^1_{\lambda_k} 
ight) \bigoplus \left( \bigoplus_{k=1}^{\infty} \mathfrak{E}_{\mu_k} 
ight).$$

In addition we have  $W = W_1$  on  $\mathcal{H} \ominus \mathcal{H}_0$ , and  $W' = W'_1$  on  $\mathcal{H}' \ominus \mathcal{H}'_0$ ; i.e. we have

$$W = W_1 = \lambda_k I \text{ on } \mathcal{E}^1_{\lambda_k} \text{ for all } k, \qquad W = W_1 = \mu_k I \text{ on } \mathcal{E}_{\mu_k} \text{ for all } k$$
$$W' = W'_1 = \lambda_k I \text{ on } \mathfrak{E}^1_{\lambda_k} \text{ for all } k, \qquad W' = W'_1 = \mu_k I \text{ on } \mathfrak{E}_{\mu_k} \text{ for all } k$$

Since dim  $\mathcal{E}_{\lambda_k}^1 = \dim \mathfrak{E}_{\lambda_k}^1 = m(\lambda_k) - 1$ , dim  $\mathcal{E}_{\mu_k} = \dim \mathfrak{E}_{\mu_k} = m(\mu_k)$ , we can define  $\phi_k^1$  to be an arbitrary unitary operator mapping  $\mathcal{E}_{\lambda_k}^1$  to  $\mathfrak{E}_{\lambda_k}^1$ ; and  $\phi_k$  to be an arbitrary unitary operator mapping  $\mathcal{E}_{\mu_k}$ . Then we can extend  $\widetilde{V}$  to an unitary  $V : \mathcal{H} \to \mathcal{H}'$  such that

$$V = V \text{ on } \mathcal{H}_0,$$
$$V = \phi_k^1 \text{ on } \mathcal{E}_{\lambda_k}^1 \text{ for all } k, \qquad V = \phi_k \text{ on } \mathcal{E}_{\mu_k} \text{ for all } k.$$

Then we have

$$W = VWV^*, \qquad W_1 = VW_1V^*, \qquad p' = Vp.$$

Thus the uniqueness part is done.

120

### Chapter 9

# RESULT FOR COMPACT HANKEL OPERATORS WITH SIMPLE EIGENVALUES

With the Abstract Borg's Theorem 8.0.1, we are ready to solve the question that what type of spectral data can uniquely determine a Hankel operator  $\Gamma$  when  $\Gamma$  is compact with simple singular values.

In section 9.1, we will show that under the self-adjoint case, a compact Hankel operator with simple singular values will be uniquely determined by two sequences of real numbers  $\{\lambda_n\}_{n=1}^{\infty}$ ,  $\{\mu_n\}_{n=1}^{\infty}$  satisfying an intertwining relation:

$$|\lambda_1| > |\mu_1| > |\lambda_2| > |\mu_2| > \dots > |\lambda_n| > |\mu_n| > \dots \to 0$$
(9.0.1)

In section 9.2, we will show that under the non-self-adjoint case, a compact Hankel operator with simple singular values will be uniquely determined by two sequences of complex numbers  $\{\lambda_n\}_{n=1}^{\infty}$ ,  $\{\mu_n\}_{n=1}^{\infty}$ , whose modulus part satisfy an intertwining relation:

$$|\lambda_1| > |\mu_1| > |\lambda_2| > |\mu_2| > \dots > |\lambda_n| > |\mu_n| > \dots \to 0$$
(9.0.2)

In section 9.3, we generate the result for finite rank Hankel operators with simple singular values, under both self-adjoint case and non self-adjoint case.

# 9.1 Result for Self-adjoint Compact Hankel Operators with Simple Eigenvalues

**Theorem 9.1.1.** Given two sequences of real numbers  $\{\lambda_n\}_{n=1}^{\infty}$  and  $\{\mu_n\}_{n=1}^{\infty}$  satisfying intertwining relations

$$|\lambda_1| > |\mu_1| > |\lambda_2| > |\mu_2| > \dots > |\lambda_n| > |\mu_n| > \dots \to 0,$$
(9.1.1)

there exists a unique self-adjoint compact Hankel operator  $\Gamma$  such that non-zero eigenvalues of  $\Gamma$  and  $\Gamma S$  are simple, and coincide with  $\{\lambda_n\}_{n=1}^{\infty}$  and  $\{\mu_n\}_{n=1}^{\infty}$  respectively.

Moreover, Ker  $\Gamma = \{0\}$  if and only if both of the following identities hold:

$$\sum_{j=1}^{\infty} \left( 1 - \frac{\mu_j^2}{\lambda_j^2} \right) = \infty, \qquad (9.1.2)$$

$$\sum_{j=1}^{\infty} \left( \frac{\mu_j^2}{\lambda_{j+1}^2} - 1 \right) = \infty.$$
(9.1.3)

*Proof.* By Theorem 8.0.1, we can find a triple  $(W, W_1, p)$  on a Hilbert space  $\mathcal{H}$  with the relation:  $W_1 = W - pp^*$ , here W and  $W_1$  are positive, compact self-adjoint operators with simple eigenvalues  $\{\lambda_k^2\}_{k=1}^{\infty}$ ,  $\{\mu_k^2\}_{k=1}^{\infty}$  respectively. In addition, the triple is unique determined up to unitary equivalence.

Denote that the eigenvector of W corresponding to  $\lambda_k^2$  is  $u_k$ , and the eigenvector of  $W_1$  corresponding to  $\mu_k^2$  is  $v_k$ ;

$$Wu_k = \lambda_k^2 u_k, \qquad W_1 v_k = \mu_k^2 v_k.$$

Since

$$\mathcal{H} = \bigoplus_{k=1}^{\infty} \overline{\operatorname{Span}} \{ u_k \} = \operatorname{Ker} W_1 \oplus \big( \bigoplus_{k=1}^{\infty} \overline{\operatorname{Span}} \{ v_k \} \big),$$

Then we define self-adjoint  $\mathcal{R}$  and  $\mathcal{R}_1$  on  $\mathcal{H}$  by

$$\mathcal{R}u_k := \lambda_k u_k;$$
$$\mathcal{R}_1 v_k := \mu_k v_k, \qquad \mathcal{R}_1 |_{\operatorname{Ker} W_1} = 0.$$

Here we have  $\operatorname{Ker} \mathcal{R} = \{0\}$  and  $\operatorname{Ker} \mathcal{R}_1 = \operatorname{Ker} W_1$  with dimension at most 1, and we can express  $\mathcal{R}$  and  $\mathcal{R}_1$  by

$$\mathcal{R}x = \sum_{k=1}^{\infty} \lambda_k P_{u_k} x, \qquad \mathcal{R}_1 x = \sum_{k=1}^{\infty} \mu_k P_{v_k} x,$$

Hence we get the unique square roots of W and  $W_1$  according to the given signs of  $\{\lambda_k\}_{k=1}^{\infty}, \{\mu_k\}_{k=1}^{\infty}$ .

Now we have the triple  $(\mathcal{R}, \mathcal{R}_1, p)$  satisfies the relation (3.1.5), and the contraction  $\mathcal{T}$  defined by  $\mathcal{R}_1 = \mathcal{T}\mathcal{R}$  is asymptotically stable. Hence by Proposition 3.1.3, there exists an unique self-adjoint Hankel operator  $\Gamma$  such that the triple  $\Gamma|_{(\text{Ker }\Gamma)^{\perp}}, \Gamma_1|_{(\text{Ker }\Gamma)^{\perp}}, u := \Gamma e_0 = \Gamma P_{(\text{Ker }\Gamma)^{\perp}} e_0$  is unitarily equivalent to the triple  $(\mathcal{R}, \mathcal{R}_1, p)$ . This means that the non-zero eigenvalues of  $\Gamma$  and  $\Gamma_1$  are simple and coincide with  $\{\lambda_k\}_{k=1}^{\infty}$  and  $\{\mu_k\}_{k=1}^{\infty}$  respectively.

As for the uniqueness, assume that  $(\Gamma, \Gamma_1, u := \Gamma e_0)$  and  $(\Gamma', \Gamma'_1, u' := \Gamma' e_0)$  are two different set of triples satisfied the conditions. Then by Proposition 3.1.3, there are two different triples  $(\mathcal{R}, \mathcal{R}_1, p), (\mathcal{R}', \mathcal{R}'_1, p)$  defined on the same Hilbert space such that  $(\Gamma, \Gamma_1, u)$  unitary equivalent to  $(\mathcal{R}, \mathcal{R}_1, p)$ , and  $(\Gamma', \Gamma'_1, u')$  unitary equivalent to  $(\mathcal{R}', \mathcal{R}'_1, p)$ , where  $(\mathcal{R}, \mathcal{R}_1)$  and  $(\mathcal{R}', \mathcal{R}'_1)$  share the same spectral characteristics.

From the equation

$$\mathcal{R}^2 - \mathcal{R}_1^2 = \mathcal{R}'^2 - \mathcal{R}_1'^2 = pp^*,$$

and the uniqueness in the abstract Borg's theorem 8.0.1, we have  $\mathcal{R}^2 = \mathcal{R}'^2 = W, \mathcal{R}_1^2 = \mathcal{R}'^2 = W_1$ . If  $\mathcal{R} \neq \mathcal{R}'$ , then there exists a k, such that  $\operatorname{Ker}(R - \lambda I) \neq \operatorname{Ker}(R' - \lambda I)$ ,

thus dim  $\operatorname{Ker}(W - \lambda^2 I) \ge 2$ , which gives a contradiction.

Similarly we have  $R_1 = R'_1$ , which indicates the uniqueness of such Hankel operator.

As for the trivial kernel condition, we have mentioned in Proposition 3.1.3 that Ker  $\Gamma = \{0\}$  if and only if ||q|| = 1 and  $q \notin \operatorname{Ran}\{\mathcal{R}\}$ , where  $q := \mathcal{R}^{-1}p$ .

Here the first condition ||q|| = 1 can be rewritten as  $||W^{-1/2}p|| = 1$ , and we have shown in Theorem 8.0.1 that this identity is equivalent to

$$\sum_{k=1}^{\infty} \left( \frac{\mu_k^2}{\lambda_k^2} - 1 \right) = -\infty \qquad \Leftrightarrow \qquad \prod_{k=1}^{\infty} \frac{\mu_k^2}{\lambda_k^2} = 0$$

And the second condition  $q \notin \operatorname{Ran} \mathcal{R}$  can be rewritten as  $||W^{-1}p|| = \infty$ , again implied by Theorem 8.0.1 that this identity is equivalent to

$$\sum_{k=1}^{\infty} \left( \frac{\mu_k^2}{\lambda_{k+1}^2} - 1 \right) = \infty \qquad \Leftrightarrow \qquad \prod_{k=1}^{\infty} \frac{\mu_k^2}{\lambda_{k+1}^2} = \infty.$$

Remark 9.1.2. With the discussion in subsection 8.2.2, we can also get an equivalent condition for Ker  $\Gamma_1/$  Ker  $\Gamma = \{0\}$  for the constructed Hankel operator, which is:

$$\prod_{k=1}^{\infty} \frac{\mu_k^2}{\lambda_k^2} = 0, \qquad \prod_{k=1}^{\infty} \frac{\mu_k^2}{\lambda_{k+1}^2} < \infty$$

## 9.2 Result for Compact Hankel Operators with Simple Singular Values as Complex Symmetric Operators

**Theorem 9.2.1.** Given complex sequences  $\{\lambda_n\}_{n=1}^{\infty}$  and  $\{\mu_n\}_{n=1}^{\infty}$ , where  $\lambda_k = s_k e^{i\theta_k}$ ,  $\mu_k = t_k e^{i\theta'_k}$ , and the modulus part satisfies the following intertwining relation

$$s_1 > t_1 > s_2 > t_2 > \dots \to 0, \tag{9.2.1}$$

then there exists a unique compact Hankel operator  $\Gamma$ , such that we can find a conjugation  $\mathfrak{J}_u$  commutes with  $|\Gamma|$  and preserves  $u := \Gamma^* e_0$  (implied by Lemma 4.0.1) with  $\widetilde{\phi}, \widetilde{\phi}_1$  defined in (5.1.5), satisfying that non-zero eigenvalues of  $|\widetilde{\Gamma}|\widetilde{\phi}, |\widetilde{\Gamma}_1|\widetilde{\phi}_1$  are simple, and coincide with  $\{\lambda_n\}_{n=1}^{\infty}$  and  $\{\mu_n\}_{n=1}^{\infty}$  respectively.

Moreover, Ker  $\Gamma = \{0\}$  if and only if both of the following identities hold:

$$\sum_{j=1}^{\infty} \left( 1 - \frac{t_j^2}{s_j^2} \right) = \infty,$$
$$\sum_{j=1}^{\infty} \left( \frac{s_j^2}{t_{j+1}^2} - 1 \right) = \infty.$$

*Proof.* By Theorem 8.0.1, we can find a triple  $(W, W_1, p)$  on a Hilbert space  $\mathcal{H}$  (unique up to equivalence) satisfying

- (i)  $W_1 = W pp^*$ , where  $W, W_1$  are positive, self-adjoint operators;
- (ii) W has trivial kernels, p is a cyclic vector for W;
- (iii)  $W, W_1$  have simple nonzero eigenvalues  $\{s_k^2\}_{k=1}^{\infty}, \{t_k^2\}_{k=1}^{\infty}$  respectively.

Now denote the eigenvectors of W corresponding to  $s_k^2$  is  $u_k$ , and eigenvectors of  $W_1$  corresponding to  $t_k^2$  is  $v_k$ . Since

$$\mathcal{H} = \bigoplus_{k=1}^{\infty} \overline{\operatorname{Span}} \{ u_k \} = \operatorname{Ker} W_1 \oplus \big( \bigoplus_{k=1}^{\infty} \overline{\operatorname{Span}} \{ v_k \} \big),$$

We define positive self-adjoint operators  $R, R_1$  on  $\mathcal{H}$  as

$$Rx := \sum_{k=1}^{\infty} s_k P_{u_k} x, \qquad R_1 x := \sum_{k=1}^{\infty} t_k P_{v_k} x$$
(9.2.2)

Now we also define a unitary  $\varphi$  and a partial isometry  $\varphi_1$  as

$$\varphi x := \sum_{k=1}^{\infty} e^{i\theta_k} P_{u_k} x, \qquad \varphi_1 x := \sum_{k=1}^{\infty} e^{i\theta'_k} P_{v_k} x \tag{9.2.3}$$

Easy to see that  $\varphi$ ,  $\varphi_1$  commutes with R,  $R_1$  respectively, and share the same corresponding one-dimensional eigenspaces. Here  $R\varphi$ ,  $R_1\varphi_1$  are compact operators with simple eigenvalues  $\{\lambda_n\}_{n=1}^{\infty}$ ,  $\{\mu_n\}_{n=1}^{\infty}$  respectively.

Now we define a conjugation  $\mathfrak{J}_p$  on  $\mathcal{H}$ . Since  $\mathcal{H} = \overline{\operatorname{Span}}\{R^n p | n \ge 0\}$ , we define  $\mathfrak{J}_p$  by

$$\mathfrak{J}_p R^n p = R^n p$$
 for all  $n$ .

This is the same as defining  $\mathfrak{J}_p$  by

$$\mathfrak{J}_p p_k = p_k \qquad \text{for all } k,$$

where

$$p_k = P\big|_{\operatorname{Ker}(W - \lambda_k^2 I)} p.$$

In addition, we can check that  $\varphi, \varphi_1$  are  $\mathfrak{J}_p$ -symmetric. In fact, for all k we have

$$\mathfrak{J}_p\varphi p_k = \mathfrak{J}_p e^{i\theta_k} p_k = e^{-i\theta_k} \mathfrak{J}_p p_k = \varphi^* \mathfrak{J}_p p_k.$$

Thus  $\mathfrak{J}_p \varphi = \varphi^* \mathfrak{J}_p$  on  $\mathcal{H}$ , and a similar result holds for  $\varphi_1$ .

Now all the assumptions in Proposition 5.2.4 are satisfied, and we apply the tuple  $(R, R_1, p, \varphi, \varphi_1, \mathfrak{J}_p)$  to Proposition 5.2.4. Thus there exists a unique Hankel  $\Gamma$ , such that we can find a conjugation  $\mathfrak{J}_u$  commuting with  $|\widetilde{\Gamma}|, |\widetilde{\Gamma}_1|$  and preserves  $u := \Gamma^* e_0$ , satisfying that the tuple  $(|\widetilde{\Gamma}|, |\widetilde{\Gamma}_1|, \widetilde{\mathfrak{J}}_u, \widetilde{\phi}, \widetilde{\phi}_1, u)$  and  $(R, R_1, \mathfrak{J}_p, \varphi, \varphi_1, p)$  are unitary equivalent. This implies that  $|\widetilde{\Gamma}| \widetilde{\phi}, |\widetilde{\Gamma}_1| \widetilde{\phi}_1$  are compact, and has simple non-zero eigenvalues as  $\{\lambda_n\}_{n=1}^{\infty}$  and  $\{\mu_n\}_{n=1}^{\infty}$  respectively. And we finish the proof of existence part.

Now we move on to uniqueness. Suppose that there are two different Hankel  $\Gamma, \Gamma'$  that satisfies all requirements in Proposition 9.2.1. We will show that  $\Gamma, \Gamma'$  have the

same Hankel coefficients  $\{\gamma_k\}_{k=1}^{\infty}$ , hence they are the same.

Denote their corresponding tuples to be  $(|\widetilde{\Gamma}|, |\widetilde{\Gamma}_1|, u, \widetilde{\phi}, \widetilde{\phi}_1)$  and  $(|\widetilde{\Gamma}'|, |\widetilde{\Gamma}'_1|, u', \widetilde{\phi}', \widetilde{\phi}'_1)$ . We have

$$|\widetilde{\Gamma}|^2 - |\widetilde{\Gamma}_1|^2 = uu^*, \qquad |\widetilde{\Gamma}'|^2 - |\widetilde{\Gamma}'_1|^2 = u'(u')^*.$$

Here  $|\widetilde{\Gamma}|, |\widetilde{\Gamma}'|$  have simple eigenvalues  $\{s_k\}_{k=1}^{\infty}$ , and  $|\widetilde{\Gamma}_1|, |\widetilde{\Gamma}'_1|$  have simple eigenvalues  $\{t_k\}_{k=1}^{\infty}$ . Then by the uniqueness in Theorem 8.0.1, those two triples  $(|\widetilde{\Gamma}|, |\widetilde{\Gamma}_1|, u)$  and  $(|\widetilde{\Gamma}'|, |\widetilde{\Gamma}'_1|, u')$  are unitary equivalent. That is, there exists a unitary  $\widetilde{\mathcal{V}} : \ell^2 \to \ell^2$  such that

$$|\widetilde{\Gamma}| = \widetilde{\mathcal{V}}|\widetilde{\Gamma}'|\widetilde{\mathcal{V}}^*, \qquad |\widetilde{\Gamma}_1| = \widetilde{\mathcal{V}}|\widetilde{\Gamma}_1'|\widetilde{\mathcal{V}}^*, \qquad u = \widetilde{\mathcal{V}}u' \tag{9.2.4}$$

We will show that

$$\widetilde{\phi} = \widetilde{\mathcal{V}}\widetilde{\phi}'\mathcal{V}^*, \qquad \widetilde{\phi}_1 = \widetilde{\mathcal{V}}\widetilde{\phi}_1'\widetilde{\mathcal{V}}^*.$$
(9.2.5)

In fact, if we denote the eigenvectors of  $|\tilde{\Gamma}|$  corresponding to  $\{s_k\}_{k=1}^{\infty}$  as  $\{u_k\}_{k=1}^{\infty}$ , i.e.

$$|\widetilde{\Gamma}|u_k = s_k u_k$$
 holds for all  $k$ .

Then since  $|\widetilde{\Gamma}|$  commutes with  $\phi$ , we have  $\{u_k\}_{k=1}^{\infty}$  are also eigenvectors for  $\phi$ :

$$\widetilde{\phi}u_k = e^{i\theta_k}u_k$$
 holds for all  $k$ .

Now from  $|\widetilde{\Gamma}| = \widetilde{\mathcal{V}}|\widetilde{\Gamma}'|\widetilde{\mathcal{V}}^*$ , we have  $\widetilde{\mathcal{V}}^*u_k$  is an eigenvector for  $|\widetilde{\Gamma}'|$ . In fact,

$$|\widetilde{\Gamma}'|\widetilde{\mathcal{V}}^* u_k = \widetilde{\mathcal{V}}^* |\widetilde{\Gamma}| u_k = s_k \widetilde{\mathcal{V}}^* u_k$$

Since  $|\widetilde{\Gamma}'|$  commutes with  $\phi'$ . we have  $\{\widetilde{\mathcal{V}}^* u_k\}_{k=1}^{\infty}$  are eigenvectors for  $\phi'$ :

$$\widetilde{\phi}'\widetilde{\mathcal{V}}^*u_k = e^{i\theta_k}\mathcal{V}^*u_k$$
 holds for all  $k$ .

Thus for all k we have

$$\widetilde{\mathcal{V}}\widetilde{\phi}'\widetilde{\mathcal{V}}^*u_k = e^{i\theta_k}\widetilde{\mathcal{V}}\widetilde{\mathcal{V}}^*u_k = e^{i\theta_k}u_k = \widetilde{\phi}u_k.$$

This proves that  $\widetilde{\mathcal{V}}\widetilde{\phi}'\widetilde{\mathcal{V}}^* = \widetilde{\phi}$ . For a similar reason, we have  $\widetilde{\mathcal{V}}\widetilde{\phi}'_1\widetilde{\mathcal{V}}^* = \widetilde{\phi}_1$ . Thus tuple  $(|\widetilde{\Gamma}|, |\widetilde{\Gamma}_1|, u, \widetilde{\phi}, \widetilde{\phi}_1)$  and  $(|\widetilde{\Gamma}'|, |\widetilde{\Gamma}'_1|, u', \widetilde{\phi}', \widetilde{\phi}'_1)$  are unitary equivalent.

Now we apply Proposition 5.1.4, we have the Hankel coefficients of  $\Gamma$  can be represented by:

$$\gamma_k = \langle (\mathfrak{S}^*)^k u, v \rangle, \qquad (9.2.6)$$

where  $\mathfrak{S}^* = |\widetilde{\Gamma}_1|\widetilde{\phi}_1^*\widetilde{\phi}|\widetilde{\Gamma}|^{-1}$ , and  $v = |\widetilde{\Gamma}|^{-1}\widetilde{\phi}^*u$  (see equation 5.1.8).

Since the two tuples for  $\Gamma$ ,  $\Gamma_1$ ,  $(|\widetilde{\Gamma}|, |\widetilde{\Gamma}_1|, u, \widetilde{\phi}, \widetilde{\phi}_1)$  and  $(|\widetilde{\Gamma}'|, |\widetilde{\Gamma}'_1|, u', \widetilde{\phi}', \widetilde{\phi}'_1)$  are unitary equivalent. Thus by (9.2.6),  $\Gamma$ ,  $\Gamma'$  have the same Hankel coefficients, thus they are the same.

As for the trivial kernel condition, we have Ker  $\Gamma = \{0\}$  is equivalent to ||q|| = 1and  $q \notin \operatorname{Ran} R$ . Again, with the construction of  $(W, W_1, p, \varphi, \varphi_1)$  in the existence part, ||q|| = 1 and  $q \notin \operatorname{Ran} R$  are equivalent to:

$$\|q\| = 1 \Leftrightarrow \|\varphi^* R^{-1} p\| = 1 \Leftrightarrow \|W^{-1/2} p\| = 1$$
$$q \notin \operatorname{Ran} R \Leftrightarrow \|R^{-1} q\| = \infty \Leftrightarrow \|R^{-2} p\| = \|W^{-1} p\| = \infty$$

Thus by Theorem 8.0.1,  $\operatorname{Ker} \Gamma = \{0\}$  if and only if

$$\sum_{j=1}^{\infty} \left( 1 - \frac{t_j^2}{s_j^2} \right) = \infty, \qquad \sum_{j=1}^{\infty} \left( \frac{s_j^2}{t_{j+1}^2} - 1 \right) = \infty.$$

Remark 9.2.2. Note that we use notation  $(\mathcal{R}, \mathcal{R}_1)$  in section 8.1, meaning that  $\mathcal{R}, \mathcal{R}_1$  are self-adjoint. While in section 8.2 we use notation  $(R, R_1)$  implying that  $R, R_1$  are also positive. In other words, in the self-adjoint case if we write the polar decomposition form of  $\mathcal{R}, \mathcal{R}_1$ :

$$\mathcal{R} = R\varphi, \qquad \mathcal{R}_1 = R_1\varphi_1,$$

then  $\sigma(\varphi) = \sigma_p(\varphi) \subseteq \{\pm 1\}, \ \sigma(\varphi_1) = \sigma_p(\varphi_1) \subseteq \{0, \pm 1\}$  and also dim Ker  $\varphi_1 \leq 1$ .

Remark 9.2.3. In the definition of unitary  $\varphi$  and partial isometry  $\varphi_1$  (see (9.2.3)), we can also write  $\varphi = f(R), \varphi_1 = f_1(R_1)$  for some unimodular measurable functions  $f, f_1$ . Here  $f, f_1$  has the following expression:

$$f(x) = \sum_{k=1}^{\infty} \mathbb{1}_{\{x=s_k\}} e^{i\theta_k} + \mathbb{1}_{\mathbb{R}/\{s_k\}_{k=1}^{\infty}},$$
(9.2.7)

$$f_1(x) = \sum_{k=1}^{\infty} \mathbb{1}_{\{x=t_k\}} e^{i\theta'_k} + \mathbb{1}_{\mathbb{R}/\{t_k\}_{k=1}^{\infty}}.$$
(9.2.8)

Remark 9.2.4. With the discussion in subsection 8.2.2, we can also get an equivalent condition for Ker  $\Gamma_1$ /Ker  $\Gamma = \{0\}$  for the constructed Hankel operator, which is:

$$\prod_{k=1}^{\infty} \frac{t_k^2}{s_k^2} = 0, \qquad \prod_{k=1}^{\infty} \frac{t_k^2}{s_{k+1}^2} < \infty.$$

## 9.3 Result for Finite Rank Hankel Operators with Simple Singular Values

Using the finite rank version of Abstract Borg's Theorem 8.3.1 given in section 8.3, we can also find out the spectral data that can uniquely determine a finite rank Hankel operator  $\Gamma$  with simple singular values. The following result is for the self-adjoint case.

**Theorem 9.3.1.** Given two sequences of real numbers  $\{\lambda_k\}_{k=1}^N$ ,  $\{\mu_k\}_{k=1}^N$ , satisfying the following intertwining relation

$$|\lambda_1| > |\mu_1| > \dots > |\lambda_N| > |\mu_N| \ge 0,$$

there exists a unique self-adjoint finite-rank Hankel operator  $\Gamma$  with simple singular values, such that the non-zero eigenvalues of  $\Gamma|_{(\text{Ker }\Gamma)^{\perp}}$  and  $\Gamma_1|_{(\text{Ker }\Gamma)^{\perp}}$  are simple, and coincide with  $\{\lambda_k\}_{k=1}^N$ ,  $\{\mu_k\}_{k=1}^N$  respectively.

In addition, for the finite-rank case, we have that the constructed  $\Gamma$  must have a infinite dimensional kernel. And Ker  $\Gamma_1$ /Ker  $\Gamma = \{0\}$  if and only if  $\mu_N > 0$ .

The following result is for the non self-adjoint case.

**Theorem 9.3.2.** Given complex sequences  $\{\lambda_n\}_{n=1}^N$  and  $\{\mu_n\}_{n=1}^N$ , where the modulus part  $s_k := |\lambda_k|, t_k := |\mu_k|$  satisfy the intertwining relation

$$s_1 > t_1 > s_2 > t_2 > \dots > s_N > t_N \ge 0,$$

then there exists a unique compact Hankel operator  $\Gamma$  with simple singular values, such that we can find a conjugation  $\mathfrak{J}_u$  commutes with  $|\Gamma|$  and preserves  $u := \Gamma^* e_0$  (implied by Lemma 4.0.1) with  $\tilde{\phi}, \tilde{\phi}_1$  defined in (5.1.5), satisfying that non-zero eigenvalues of  $|\tilde{\Gamma}|\tilde{\phi}, |\tilde{\Gamma}_1|\tilde{\phi}_1$  are simple, and coincide with  $\{\lambda_n\}_{n=1}^{\infty}$  and  $\{\mu_n\}_{n=1}^{\infty}$  respectively.

Moreover, the constructed Hankel operator  $\Gamma$  must have an infinite dimensional kernel  $\Gamma$ , and Ker  $\Gamma_1$ /Ker  $\Gamma = \{0\}$  if and only if  $\mu_N = 0$ .

The proof to these two theorems are very much similar to the ones for Theorem 9.1.1 and Theorem 9.2.1, so we omit them.

### 9.4 Conclusion

In this chapter, we further translate the spectral data of two operators that can uniquely determine a compact Hankel operator with simple singular values.

(i) If the Hankel operator  $\Gamma$  is self-adjoint, Theorem 9.1.1 indicates that  $\Gamma$  will be uniquely determined by two sequences of real numbers  $\{\lambda_n\}_{n=1}^{\infty}$ ,  $\{\mu_n\}_{n=1}^{\infty}$ , satisfying an intertwining relations

$$|\lambda_1| > |\mu_1| > |\lambda_2| > |\mu_2| > \dots > |\lambda_n| > |\mu_n| > \dots \to 0.$$
(9.4.1)

If we write the Borel transform of  $\Gamma^2$  with respect to u:

$$\left((\Gamma^2 - zI)^{-1}u, u\right) = \int \frac{d\rho(s)}{s-z},$$

then the coefficients of the scalar spectral measure  $\rho(s) = \sum_{k\geq 1} a_k \delta_{\lambda_k^2}(s)$  is uniquely determined with the expression in (8.1.9).

(ii) For the case when Hankel operator is not self-adjoint, Theorem 9.2.1 implies that  $\Gamma$  can be uniquely determined by two sequences of complex numbers  $\{\lambda_n\}_{n=1}^{\infty}, \{\mu_n\}_{n=1}^{\infty}$ , whose modulus part satisfy an intertwining relation (9.2.1).

$$|\lambda_1| > |\mu_1| > |\lambda_2| > \dots \to 0.$$

In other words, with the constructed conjugation  $\tilde{\mathfrak{J}}_{\Gamma}$  which commutes with  $|\tilde{\Gamma}|, |\tilde{\Gamma}_1|$ and preserves  $u := \Gamma^* e_0$ , and the induced unitary  $\tilde{\phi}$  and partial isometry  $\tilde{\phi}_1$  given in (5.1.5). We have the non-zero eigenvalues of the two operators  $|\tilde{\Gamma}|\tilde{\phi}, |\tilde{\Gamma}_1|\tilde{\phi}_1$  are simple, and coincide with  $\{\lambda_n\}_{n=1}^{\infty}$  and  $\{\mu_n\}_{n=1}^{\infty}$  respectively.

For this second case, we can also say that the Hankel operator  $\Gamma$  can be uniquely determined by three factors:

(a) A discrete measure  $\rho(s)$ , which is the scalar spectral measure of  $|\Gamma|^2$  with respect

to  $u = \Gamma^* e_0$ :

$$\left((|\Gamma|^2 - zI)^{-1}u, u\right) = \int \frac{d\rho(s)}{s-z}.$$

The coefficients of this scalar measure is uniquely determined by the modulus part of sequences  $\{\lambda_n\}_{n=1}^{\infty}$  and  $\{\mu_n\}_{n=1}^{\infty}$ .

- (b) A unimodular measurable function  $f : \mathbb{R} \to \mathbb{C}$  (with the expression in (9.2.7)) , which is determined by the phase part of complex sequences  $\{\lambda_n\}_{n=1}^{\infty}$ . This function induces an unitary operator  $f(|\widetilde{\Gamma}|)$  commutes with  $|\Gamma|$  and share the same eigenvectors with  $|\widetilde{\Gamma}|$ .
- (c) A unimodular measurable function  $f_1 : \mathbb{R} \to \mathbb{C}$  (with the expression in (9.2.8)), which is determined by the phase part of complex sequences  $\{\mu_n\}_{n=1}^{\infty}$ . This function induces an operator  $f_1(|\tilde{\Gamma}_1|)$  commutes with  $|\tilde{\Gamma}_1|$  and share the same eigenvectors with  $|\tilde{\Gamma}_1|$ .

Notice that for both situations, we require the cyclicity of vector p to guarantee the asymptotic stability of a contraction  $\mathcal{T} := R_1 \varphi_1 \varphi^* R$  in the description of Proposition 3.1.3 and Proposition 5.2.3. Thus the geometric multiplicities of singular values of constructed  $\Gamma$  must be no more than 1.

### Chapter 10

# RESULT FOR COMPACT HANKEL OPERATORS WITH NON-SIMPLE EIGENVALUES

In this chapter, we will further discuss the spectral data we need that can uniquely determine a compact Hankel operator with non-simple singular values. The main result in this chapter is given in Theorem 10.1.5, saying that a compact Hankel operator with non-simple singular values can be uniquely determined by two sequences of positive real numbers  $\{\lambda_k\}_{k=1}^{\infty}$ ,  $\{\mu_k\}_{k=1}^{\infty}$ , and two sequences of positive discrete probability measure  $\{\tilde{\rho}_k\}_{k=1}^{\infty}$ ,  $\{\tilde{\rho}_k^1\}_{k=1}^{\infty}$ . Here the two sequences of real numbers satisfy the following intertwining relation:

$$\lambda_1 > \mu_1 > \lambda_2 > \mu_2 > \dots \to 0,$$

In section 10.1, we recall the setting in previous chapter and do some preparation work. And in section 10.2, we prove the main result (Theorem 10.1.5) in this chapter.

### **10.1** Preparation Work and Main Result

#### 10.1.1 The setting of Tuple from Previous Chapters

We first recall the setting of tuple  $(R, R_1, p, \varphi, \varphi_1, \mathfrak{J}_p)$  defined in chapter 5:
- (i)  $R, R_1$  are two positive, self-adjoint compact operator defined on a Hilbert space  $\mathcal{H}$ . In addition we have Ker  $R = \{0\}$ ;
- (ii)  $R^2 R_1^2 = pp^*$  for a vector p with  $||R^{-1}p|| \le 1$ ;
- (iii)  $\mathfrak{J}_p$  is a conjugation commutes with  $R, R_1$  and preserves p, implied by Lemma 4.0.1 and Lemma 5.1.1;
- (iv)  $\varphi$  is a  $\mathfrak{J}_p$ -symmetric unitary operator, which commutes with R;
- (v)  $\varphi_1$  is a  $\mathfrak{J}_p$ -symmetric partial isometry with Ker  $\varphi_1 = \text{Ker } R_1$ , which commutes with  $R_1$ . In addition, we have  $\varphi_1|_{(\text{Ker } R_1)^{\perp}}$  is unitary (See Remark 5.2.1);
- (vi) The contraction  $\mathcal{T}$  defined as  $\mathcal{T} := \varphi_1 R_1 R^{-1} \varphi^*$  is asymptotically stable.

We again use the definition  $\mathcal{H}_0 := \overline{\text{Span}}\{R^n p | n \ge 0\}$ . Then in Proposition 6.2.2, we have shown that there exists an intertwining real sequence

$$\lambda_1 > \mu_1 > \lambda_2 > \mu_2 > \dots \to 0,$$
 (10.1.1)

such that  $R|_{\mathcal{H}_0}, R_1|_{\mathcal{H}_0}$  have  $\{\lambda_k\}_{k=1}^{\infty}$ .  $\{\mu_k\}_{k=1}^{\infty}$  as their non-zero eigenvalues respectively.

Afterwards, Proposition 6.2.10 gives the complete structure of the eigenspaces of  $R, R_1$ . From Lemma 6.2.8 we know that  $R, R_1$  doesn't have any eigenvalues other than  $\{\lambda_k\}_{k=1}^{\infty}, \{\mu_k\}_{k=1}^{\infty}$  and 0. Using the same notation  $E_{\lambda_k}, E_{\lambda_k}^1, E_{\mu_k}, E_{\mu_k}^1$  defined in (6.2.7), then we have

(i)

$$\mathcal{H}_0^{\perp} = \left(\bigoplus_{k=1}^{\infty} E_{\lambda_k}^1\right) \bigoplus \left(\bigoplus_{k=1}^{\infty} E_{\mu_k}\right)$$

(ii)

$$E_{\lambda_k} = E_{\lambda_k}^1 \oplus \operatorname{Span}\{p_k\}, \qquad E_{\mu_k}^1 = E_{\mu_k} \oplus \operatorname{Span}\{p_k^1\},$$
134

where  $p_k, p_k^1$  are defined in (6.2.9).

Furthermore, if we take the canonical choice of  $\varphi, \varphi_1$  given in Lemma 6.3.2:

$$\left.\varphi_1\right|_{E^1_{\lambda_k}} = I, \qquad \left.\varphi\right|_{E_{\mu_k}} = I,$$

then by Proposition 7.6.3, the asymptotic stability of  $\mathcal{T}$  implies that:

- (i)  $p_k$  is \*-cyclic for  $\varphi$  restricted in  $E_{\lambda_k}$ ;
- (ii)  $p_k^1$  is \*-cyclic for  $\varphi_1$  restricted in  $E_{\mu_k}^1$ .

*Remark* 10.1.1. (i) We should notice that all  $E_{\lambda_k}$ ,  $E_{\lambda_k}^1$ ,  $E_{\mu_k}$ ,  $E_{\mu_k}^1$  are eigenspaces of compact operators, so they must have finite dimensions.

(ii) From the asymptotic stability assumption, we have for all k

$$E_{\lambda_k} = \overline{\operatorname{Span}} \big\{ \varphi^n p_k | n \in \mathbb{Z} \big\}.$$

Here we denote  $\varphi^n = (\varphi^*)^{-n}$  for a negative n.

Actually, we can show that  $E_{\lambda_k} = \overline{\text{Span}} \{ \varphi^n p_k | |n| \leq N_k, n \in \mathbb{Z} \}$  holds for a finite  $N_k$ .

In fact, if we define  $E_i = \overline{\text{Span}} \{ \varphi^n p_k | |n| \le i, n \in \mathbb{Z} \}$ , then  $E_1 \subseteq E_2 \subseteq ...$  is an increasing sequence. Since dim  $E_{\lambda_k} < \infty$ , thus the sequence must stay at a constant subspace from a certain  $E_i$ .

## 10.1.2 Behavior of $\varphi, \varphi_1$ on the Eigenspaces

Now for the given  $p_k \in E_{\lambda_k}$  and the unitary operator  $\varphi|_{E_{\lambda_k}}$ , we consider all functions f(z) defined on  $\mathbb{T}$  with the following form:

$$f(z) = \sum_{n=-N}^{N} a_n z^n$$
 N is finite  $, a_n \in \mathbb{C}.$ 

Denote the set that consists of all such f to be S. For all  $f \in S$ , we have

$$f \to \langle p_k, f(\varphi) p_k \rangle$$
 (10.1.2)

is a linear functional on S. According to Stone Weierstrass Theorem,  $P(\mathbb{T})$  (the set of polynomials on  $\mathbb{T}$  with complex coefficients) is dense in  $C(\mathbb{T})$  (the set of continuous functions on  $\mathbb{T}$ ), thus S is also dense in  $C(\mathbb{T})$ . And (10.1.2) can be extended to all  $f \in C(\mathbb{T}).$ 

Now applying Riesz Representation Theorem, we can find a complex Borel measure on  $\mathbb{T}$  (We have  $\sigma(\varphi) \subseteq \mathbb{T}$  since  $\varphi$  is unitary), denoted as  $\rho_k$ , such that

$$\int_{\mathbb{T}} f(s) d\rho_k(s) = \langle p_k, f(\varphi) p_k \rangle.$$
(10.1.3)

This measure  $\rho_k$  is also known as the *scalar spectral measure* of  $\varphi|_{E_{\lambda_k}}$  with respect to  $p_k$ .

In addition, the *Borel transform* of  $\varphi|_{E_{\lambda_k}}$  with respect to  $p_k$  is given by

$$F_k(z) = \langle (\varphi - zI)^{-1} p_k, p_k \rangle = \int_{\mathbb{T}} \frac{d\rho_k(s)}{s - z}.$$

**Proposition 10.1.2.** The unitary operator  $\varphi|_{E_{\lambda_k}}$  and the scalar spectral measure  $\rho_k(s)$ in (10.1.3) satisfies the following properties:

(i)  $\rho_k(s) = \sum_{i=1}^N a_i \delta_{\beta_i}(s)$  for some  $a_i > 0, |\beta_i| = 1;$ \_\_\_\_ N

(ii) 
$$\varphi|_{E_{\lambda_k}} = \sum_{i=1}^{k} \beta_i P_{\operatorname{Ker}(\varphi|_{E_{\lambda_k}} - \beta_i I)}$$

- (iii) dim Ker $(\varphi|_{E_{\lambda_k}} \beta_i I) = 1$  holds for all *i*. Hence  $N = \dim E_{\lambda_k}$ ;
- (iv)  $\sum_{i=1}^{N} a_i = \rho_k(\mathbb{T}) = ||p_k||^2$ . For each *i*, we have

$$a_i = \|P_{\operatorname{Ker}(\varphi|_{E_{\lambda_k}} - \beta_i I)} p_k\|^2$$
136

Proof of Proposition 10.1.2. First notice that  $\varphi|_{E_{\lambda_k}}$  is unitary and finite rank, thus by the spectral theorem for compact normal operators (See [37]), we can write

$$\varphi\big|_{E_{\lambda_k}} = \sum_{i=1}^N \beta_i P_i. \tag{10.1.4}$$

Here  $\{\beta_i\}_{i=1}^N$  are all eigenvalues of  $\varphi|_{E_{\lambda_k}}$ . and  $P_i$  is the projection operator onto  $\operatorname{Ker}(\varphi|_{E_{\lambda_k}} - \beta_i I)$ . Since  $\varphi|_{E_{\lambda_k}}$  is unitary, we have all  $\beta_i$  has module 1.

Then the scalar spectral measure  $\rho_k(s)$  can be written as

$$\rho_k(s) = \sum_{i=1}^N a_i \delta_{\beta_i}(s)$$

for some  $a_i \in \mathbb{C}$ .

Now we write  $p_k = \sum_{i=1}^{N} p_{ki}$ , where  $p_{ki} \in \text{Ker}(\varphi|_{E_{\lambda_k}} - \beta_i I)$ . We will show that

$$p_{ki} \neq 0, \qquad \dim \operatorname{Ker}(\varphi \big|_{E_{\lambda_k}} - \beta_i I) = 1.$$

In fact, from (10.1.4), we can write  $\varphi^*|_{E_{\lambda_k}}$  as

$$\varphi^*\big|_{E_{\lambda_k}} = \sum_{i=1}^N \overline{\beta_i} P_i. \tag{10.1.5}$$

If  $p_{ki} = 0$ , then we have

$$\overline{\operatorname{Span}}\big\{\varphi^n p_k, (\varphi^*)^n p_k\big\} \perp \operatorname{Ker}(\varphi\big|_{E_{\lambda_k}} - \beta_i I),$$

which contradicts to the \*-cyclicity of  $p_k$ .

On the other hand, we know from the representation of  $\varphi, \varphi^*$  in (10.1.4) and

(10.1.5), we know that the intersection

$$\overline{\operatorname{Span}}\big\{\varphi^n p_k, (\varphi^*)^n p_k\big\} \cap \operatorname{Ker}(\varphi\big|_{E_{\lambda_k}} - \beta_i I)$$

is at most a one-dimensional space. Thus if dim  $\operatorname{Ker}(\varphi|_{E_{\lambda_k}} - \beta_i I) \geq 2$  for some *i*, it will also contradict to the \*-cyclicity of  $p_k$ . Hence we finish the proof of (iii).

Now for (iv), we apply a sequence of polynomials  $\{f_n\}$  that converges to  $\mathbb{1}_{\{x=\beta_i\}}$  to (10.1.3), then we get

$$a_i = \langle p_k, P_{\operatorname{Ker}(\varphi|_{E_{\lambda_k}} - \beta_i I)} p_k \rangle = \| p_{ki} \|^2 > 0.$$

Thus we have

$$\rho_k(\mathbb{T}) = \sum_{i=1}^N a_i = \sum_{i=1}^N ||p_{ki}||^2 = ||p_k||^2.$$

We can also get this identity by simply setting f = 1 into (10.1.3)

Remark 10.1.3. Note that equation (10.1.4) implies that  $\varphi|_{E_{\lambda_k}}$  is diagonalizable. Thus for each eigenvalue  $\beta_i$ , the algebraic multiplicity of  $\beta_i$  equals to the geometric multiplicity of  $\beta_i$ . This is not necessarily true for a general compact operator, where algebraic multiplicity is usually bigger than the geometric multiplicity.

Remark 10.1.4. Denoting  $N = \dim E_{\lambda_k}$ , we can show that

$$\overline{\operatorname{Span}}\big\{\varphi^n p_k\big| 0 \le n \le N - 1\big\} = E_{\lambda_k},$$

which is stronger than the condition in Proposition 7.6.3.

In fact, this can immediately be resulted from the fact that the projection of  $p_k$  on

each eigenspace  $\operatorname{Ker}(\varphi|_{E_{\lambda_k}} - \beta_i I)$  is non-zero, and the Vandermonde matrix

$$M_{ij} = \begin{pmatrix} 1 & 1 & \dots & 1 \\ \beta_1 & \beta_2 & \dots & \beta_N \\ \dots & & & & \\ \beta_1^{N-1} & \beta_2^{N-1} & \dots & \beta_N^{N-1} \end{pmatrix}$$

has non-zero determinant

$$|(M_{ij})_{1 \le i,j \le N}| = \prod_{i < j} (\beta_j - \beta_i).$$

Now similarly for  $p_k^1$  and  $\varphi_1|_{E_{\mu_k}^1}$ , we can define the scalar spectrum measure  $\rho_k^1$  on  $\mathbb{T}$  by:

$$\int_{\mathbb{T}} f(s) d\rho_k^1(s) = \langle p_k^1, f(\varphi_1) p_k^1 \rangle,$$

and the Borel transform of  $\varphi_1|_{E^1_{\mu_k}}$  with respect to  $p^1_k$ 

$$F_k^1(z) = \langle (\varphi_1 - zI)^{-1} p_k^1, p_k^1 \rangle = \int_{\mathbb{T}} \frac{d\rho_k^1(s)}{s - z}.$$

Similar to the proof in Proposition 10.1.2,  $\rho_k^1(s)$  is also a positive discrete measure, which concentrate on finitely many points on  $\mathbb{T}$ . And we also have

$$\rho_k^1(\mathbb{T}) = \|p_k^1\|^2.$$

We can see that the total measure of  $\rho_k(s)$ ,  $\rho_k^1(s)$  equal to  $||p_k||^2$ ,  $||p_k^1||^2$ , which is not independent on the intertwining sequence  $\{\lambda_k\}_{k=1}^{\infty}$ ,  $\{\mu_k\}_{k=1}^{\infty}$ . Hence if we normalize  $\rho_k$ ,  $\rho_k^1$  by changing  $p_k$ ,  $p_k^1$  to  $\frac{p_k}{||p_k||}$ ,  $\frac{p_k^1}{p_k^1}$  respectively, that is, we denote  $\tilde{\rho}_k$ ,  $\tilde{\rho}_k^1$  to be the scalar spectral measure of  $\varphi|_{E_{\lambda_k}}, \varphi_1|_{E_{\mu_k}^1}$  with respect to  $\frac{p_k}{\|p_k\|}, \frac{p_k^1}{\|p_k^1\|}$  respectively:

$$\int_{\mathbb{T}} f(s)d\widetilde{\rho}_k(s) = \langle \frac{p_k}{\|p_k\|}, f(\varphi) \frac{p_k}{\|p_k\|} \rangle, \qquad \int_{\mathbb{T}} f(s)d\widetilde{\rho}_k^1(s) = \langle \frac{p_k^1}{\|p_k^1\|}, f(\varphi_1) \frac{p_k^1}{\|p_k^1\|} \rangle,$$

then we have  $\widetilde{\rho}_k(\mathbb{T}) = \widetilde{\rho}_k^1(\mathbb{T}) = 1.$ 

In this section, we will show that a Hankel operator can be uniquely determined by the choice of  $\{\lambda_k\}_{k=1}^{\infty}$ ,  $\{\mu_k\}_{k=1}^{\infty}$ ,  $\{\widetilde{\rho}_k\}_{k=1}^{\infty}$ ,  $\{\widetilde{\rho}_k\}_{k=1}^{\infty}$ ; i.e. we can take  $\{\lambda_k\}_{k=1}^{\infty}$ ,  $\{\mu_k\}_{k=1}^{\infty}$ ,  $\{\widetilde{\rho}_k\}_{k=1}^{\infty}$ ,  $\{\widetilde{\rho}_k^1\}_{k=1}^{\infty}$  as the spectral data of a compact  $\Gamma$ .

**Theorem 10.1.5.** Given an intertwining sequence

$$\lambda_1 > \mu_1 > \lambda_2 > \mu_2 > \dots \to 0,$$

and two sequences of normalized measure on  $\mathbb{T}$ , denoted as  $\{\widetilde{\rho}_k\}_{k=1}^{\infty}$ ,  $\{\widetilde{\rho}_k^1\}_{k=1}^{\infty}$ , satisfy

(i) 
$$\widetilde{\rho}_k(\mathbb{T}) = 1, \widetilde{\rho}_k^1(\mathbb{T}) = 1;$$

(ii)  $\rho_k, \rho_k^1$  are positive discrete measure concentrated on finitely many points on  $\mathbb{T}$ .

then there exists a unique Hankel operator  $\Gamma$ , such that

(a)  $|\widetilde{\Gamma}|, |\widetilde{\Gamma}_1|$  are compact, and have no other non-zero eigenvalues than  $\{\lambda_k\}_{k=1}^{\infty}, \{\mu_k\}_{k=1}^{\infty}$ . Furthermore, if denoting

$$\begin{split} \mathfrak{E}^{1}_{\lambda_{k}} &:= \operatorname{Ker}(|\widetilde{\Gamma}_{1}| - \lambda_{k}I), \qquad \mathfrak{E}_{\lambda_{k}} &:= \operatorname{Ker}(|\widetilde{\Gamma}| - \lambda_{k}I); \\ \mathfrak{E}^{1}_{\mu_{k}} &:= \operatorname{Ker}(|\widetilde{\Gamma}_{1}| - \mu_{k}I), \qquad \mathfrak{E}_{\mu_{k}} &:= \operatorname{Ker}(|\widetilde{\Gamma}| - \mu I), \end{split}$$

then we have

$$\mathfrak{E}^{1}_{\lambda_{k}} \oplus \operatorname{span}\{v_{1}^{k}\} = \mathfrak{E}_{\lambda_{k}}, \qquad \mathfrak{E}_{\mu_{k}} \oplus \operatorname{span}\{v_{2}^{k}\} \subseteq \mathfrak{E}^{1}_{\mu_{k}}$$

for some vectors  $v_1^k, v_2^k$ .

(b) Under the canonical choice of tuple  $\langle \tilde{\mathfrak{J}}_{u}, \tilde{\phi}, \tilde{\phi}_{1} \rangle$  (See (5.1.5) and Lemma 6.3.2), i.e.

$$\widetilde{\phi}|_{\mathfrak{E}_{\mu_k}} = I, \qquad \widetilde{\phi}_1|_{\mathfrak{E}^1_{\lambda_k}} = I,$$

we have the spectral measure of  $\widetilde{\phi}|_{E_{\lambda_k}}$  with respect to  $v_1^k$ , and the spectral measure of  $\widetilde{\phi}_1|_{E_{\mu_k}^1}$  with respect to  $v_2^k$ , are just scalar multiplication of  $\widetilde{\rho}_k$ ,  $\widetilde{\rho}_k^1$  respectively.

We will prove this proposition in next section.

## **10.2 Proof of Theorem 10.1.5**

#### 10.2.1 Proof of Existence

*Proof.* According to Proposition 5.2.4, to prove the existence of such Hankel  $\Gamma$ , it's sufficient to construct a tuple  $(R, R_1, p, \varphi, \varphi_1)$  that satisfies the same property stated in (a),(b).

First by applying The abstract Borg's theorem 8.0.1, there exists an unique (up to unitary equivalence) triple  $(W, W_1, p)$  defined on a Hilbert space  $\mathcal{H}_0$ , such that  $W, W_1$ are positive compact operators, Ker  $W = \{0\}$  satisfying

- (i)  $W = W_1 + pp^*;$
- (ii)  $W, W_1$  have simple non-zero eigenvalues as  $\{\lambda_k^2\}_{k=1}^{\infty}$ ,  $\{\mu_k^2\}_{k=1}^{\infty}$  respectively.

Thus if we denoting the eigenvectors of  $W, W_1$  as

$$Wx_k = \lambda_k^2 x_k, \qquad W_1 y_k = \mu_k^2 y_k;$$

Then we can construct two operators  $\widetilde{R}, \widetilde{R}_1$  on  $\mathcal{H}_0$  defined as

$$\widetilde{R}x_k := \lambda_k x_k, \qquad \widetilde{R}_1 y_k = \mu_k y_k.$$

Here we can express the norm of  $p_k, p_k^1$  in terms of  $\{\lambda_k^2\}_{k=1}^{\infty}$ ,  $\{\mu_k^2\}_{k=1}^{\infty}$ , which are defined as

$$p_k := P_{\operatorname{Ker}(R-\lambda_k I)} p, \qquad p_k^1 := P_{\operatorname{Ker}(R_1-\mu_k)} p.$$

In fact, if we denote  $\mu, \mu_1$  as the scalar spectral measure of  $R^2, R_1^2$  with respect to p, and  $g(z), g_1(z)$  to be the Cauchy transform of  $R^2, R_1^2$  with respect to p. That is,

$$\int_{\mathbb{R}} f(s)d\mu(s) = \langle p, f(R^2)p \rangle \text{ holds for all } f \in C(\mathbb{R})$$
(10.2.1)

$$g(z) = \langle (R^2 - zI)^{-1}p, p \rangle = \int \frac{d\mu(s)}{s-z}$$
(10.2.2)

$$\int_{\mathbb{R}} f(s) d\mu_1(s) = \langle p, f(R_1^2)p \rangle \text{ holds for all } f \in C(\mathbb{R})$$
(10.2.3)

$$g_1(z) = \langle (R_1^2 - zI)^{-1}p, p \rangle = \int \frac{d\mu_1(s)}{s-z}$$
(10.2.4)

Then we have  $1 - g = \frac{g}{g_1}$  from Proposition 8.1.1, and we can write

$$\mu(z) = \sum_{k} a_k \delta_{\lambda_k^2}(z), \qquad \mu_1(z) = \sum_{k} b_k \delta_{\mu_k^2}(z) + b_0 \delta_0(z).$$

According to Lemma 8.1.9 and Proposition 8.2.9, the coefficients  $\{a_k\}_{k=1}^{\infty}, \{b_k\}_{k=1}^{\infty}$ has the following representation:

$$a_{k} = (\lambda_{k}^{2} - \mu_{k}^{2}) \prod_{i \neq k} \left( \frac{\lambda_{k}^{2} - \mu_{i}^{2}}{\lambda_{k}^{2} - \lambda_{i}^{2}} \right);$$
(10.2.5)

$$b_n = (\lambda_n^2 - \mu_n^2) \prod_{k \neq n} \left( \frac{\mu_n^2 - \lambda_k^2}{\mu_n^2 - \mu_k^2} \right)$$
(10.2.6)

$$b_0 = \begin{cases} 0 & \text{if } \sum_n \frac{a_n}{\lambda_n^2} < 1, \text{ or } \sum_n \frac{a_n}{\lambda_n^2} = 1 \text{ and } \sum_n \frac{a_n}{\lambda_n^4} = \infty. \\ \left(\sum_n \frac{a_n}{\lambda_n^4}\right)^{-1} & \text{if } \sum_n \frac{a_n}{\lambda_n^2} = 1 \text{ and } \sum_n \frac{a_n}{\lambda_n^4} < \infty. \end{cases}$$
(10.2.7)

Here according to the proof in section 8.2,

$$\sum_{n} \frac{a_n}{\lambda_n^2} = 1, \qquad \Leftrightarrow \|W^{-1/2}p\| = 1, \qquad \Leftrightarrow \sum_{j=1}^{\infty} \left(1 - \frac{\mu_j^2}{\lambda_j^2}\right) = \infty,$$
$$\sum_{n} \frac{a_n}{\lambda_n^4} = \infty, \qquad \Leftrightarrow \|W^{-1}p\| = \infty, \qquad \Leftrightarrow \sum_{j=1}^{\infty} \left(\frac{\mu_j^2}{\lambda_{j+1}^2} - 1\right) = \infty$$

Now we apply a sequence of polynomials  $\{f_n\}$  which converges to  $\mathbb{1}_{\{x=\lambda_k^2\}}$  into (10.2.1), then we will have

$$a_k = \langle p, P_{\operatorname{Ker}(R^2 - \lambda_k^2 I)} p \rangle = \|p_k\|^2,$$

thus

$$||p_k|| = \sqrt{a_k} = \sqrt{(\lambda_k^2 - \mu_k^2) \prod_{i \neq k} \left(\frac{\lambda_k^2 - \mu_i^2}{\lambda_k^2 - \lambda_i^2}\right)}.$$
 (10.2.8)

Similarly we have

$$\|p_k^1\| = \sqrt{b_k} = \sqrt{(\lambda_k^2 - \mu_k^2) \prod_{i \neq k} \left(\frac{\mu_k^2 - \lambda_i^2}{\mu_k^2 - \mu_i^2}\right)}.$$
 (10.2.9)

Now denoting  $\widetilde{p}_k = \frac{p_k}{\|p_k\|}, \widetilde{p}_k^1 = \frac{p_k^1}{\|p_k^1\|}$ , we want to construct a sequence of orthogonal Hilbert space  $\mathcal{H}_1, \mathcal{H}_2, \dots, \mathcal{H}_{2k}$  which are all orthogonal to  $\mathcal{H}_0$ , and two sequences of unitary operators  $\{\varphi_k\}_{k=1}^{\infty}, \{\varphi_k^1\}_{k=1}^{\infty}$  such that

(i)  $\varphi_k : \mathcal{H}_{2k-1} \oplus \operatorname{span}\{p_k\} \to \mathcal{H}_{2k-1} \oplus \operatorname{span}\{p_k\}$  has  $\widetilde{\rho}_k$  as its spectral measure with respect to  $\widetilde{p}_k$ ;

(ii)  $\varphi_k^1 : \mathcal{H}_{2k} \oplus \operatorname{span}\{p_k^1\} \to \mathcal{H}_{2k} \oplus \operatorname{span}\{p_k^1\}$  has  $\tilde{\rho}_k^1$  as its spectral measure with respect to  $\tilde{p}_k^1$ ;

We first need to determine the dimension of each  $\mathcal{H}_k$ . This can be easily seen from Proposition 10.1.2. If  $\tilde{\rho}_k$  concentrates on m + 1 discrete points on  $\mathbb{T}$ , then we have  $\dim \mathcal{H}_{2k-1} = m$ . And a similar result holds for  $\tilde{\rho}_k^1$  and  $\mathcal{H}_{2k}$ .

Then we choose an orthogonal basis for  $\mathcal{H}_{2k-1} \oplus \operatorname{span}\{p_k\}$ , denoted as  $\widetilde{p}_k, e_1, ..., e_m$ . We construct the unitary  $\varphi_k$  such that

$$\overline{\operatorname{Span}}\{\widetilde{p}_k, \varphi_k^j \widetilde{p}_k \big| 1 \le j \le i\} = \overline{\operatorname{Span}}\{\widetilde{p}_k, e_1, \dots, e_i\} \qquad \text{holds for all } 1 \le i \le m$$

Notice that the inner product between any two of  $\tilde{p}_k, \varphi_k \tilde{p}_k, \dots, \varphi_k^m \tilde{p}_k$  can be uniquely determined by  $\tilde{\rho}_k(s)$ . In fact,

$$\langle (\varphi_k)^i \widetilde{p}_k, (\varphi_k)^j \widetilde{p}_k \rangle = \langle \widetilde{p}_k, (\varphi_k)^{j-i} \widetilde{p}_k \rangle = \int_{\mathbb{T}} s^{j-i} d\widetilde{\rho}_k(s) \text{ for all integers } i, j.$$
(10.2.10)

Thus if we write  $\varphi_k \tilde{p}_k = \beta_{10} \tilde{p}_k + \beta_{11} e_1$ , we have  $\beta_{10} = \langle \varphi_k \tilde{p}_k, \tilde{p}_k \rangle$ , and we take  $\beta_{11} = \sqrt{1 - \beta_{10}^2}$ . Since in Remark 10.1.4, we have stated that  $\tilde{p}_k, \varphi_k \tilde{p}_k, \dots, \varphi_k^m \tilde{p}_k$  are linearly independent, thus  $\beta_{11} \neq 0$ .

Similarly, if we write  $\varphi_k^i \widetilde{p}_k = \beta_{i0} \widetilde{p}_k + \ldots + \beta_{ii} e_i$ . Then the coefficients  $\beta_{i0}, \ldots, \beta_{i(i-1)}$ will be uniquely determined by the inner product between  $\varphi_k^i \widetilde{p}_k$  and  $\widetilde{p}_k, \varphi_k \widetilde{p}_k, \ldots, \varphi_k^{i-1} \widetilde{p}_k$ . And then we take  $\beta_{ii} = \sqrt{1 - \sum_{j=0}^{i-1} \beta_{ij}^2}$ . Proposition 10.1.2 guarantees that  $\beta_{ii} = 0$ .

So now we have defined  $\varphi^i \widetilde{p}_k$   $(1 \leq i \leq m)$  in terms of basis  $\widetilde{p}_k, e_1, ..., e_m$ . Since  $\overline{\text{Span}}\{\widetilde{p}_k, \varphi^j_k \widetilde{p}_k | 1 \leq j \leq m\} = \overline{\text{Span}}\{\widetilde{p}_k, e_1, ..., e_m\}$ , we have  $\varphi_k$  well-defined on  $\mathcal{H}_{2k-1} \oplus$  $\text{span}\{p_k\}$ .

Similarly, we find an orthogonal basis on  $\mathcal{H}_{2k}$ , and then calculate the coordinates of  $(\varphi_k^1)^i \widetilde{p}_k^1$  in terms of this set of basis and  $\widetilde{p}_k^1$ , where  $1 \leq i \leq \dim \mathcal{H}_{2k}$ .

Now we denote

$$\mathcal{H} := \mathcal{H}_0 \oplus \mathcal{H}_1 \oplus \mathcal{H}_2 ...,$$

We extend  $\widetilde{R}, \widetilde{R}_1$  to  $R, R_1$  defined from  $\mathcal{H}$  to  $\mathcal{H}$ :

$$R|_{\mathcal{H}_0} = \widetilde{R}, \qquad R|_{\mathcal{H}_{2k-1}} = \lambda_k I, \qquad R|_{\mathcal{H}_{2k}} = \mu_k I;$$
$$R_1|_{\mathcal{H}_0} = \widetilde{R}_1, \qquad R_1|_{\mathcal{H}_{2k-1}} = \lambda_k I, \qquad R_1|_{\mathcal{H}_{2k}} = \mu_k I.$$

Now we define an unitary  $\varphi : \mathcal{H} \to \mathcal{H}$  as:

$$\left. \varphi \right|_{\mathcal{H}_{2k-1} \oplus \operatorname{span}\{p_k\}} = \varphi_k, \qquad \left. \varphi \right|_{\mathcal{H}_{2k}} = I \text{ for all } k;$$

and a partial isometry  $\varphi_1 : \mathcal{H} \to \mathcal{H}$  as:

$$\varphi_1|_{\mathcal{H}_{2k}\oplus \operatorname{span}\{p_k^1\}} = \varphi_k^1, \qquad \varphi_1|_{\mathcal{H}_{2k-1}} = I \text{ for all } k; \qquad \varphi_1|_{\mathcal{H}_0 \cap \operatorname{Ker} R_1} = 0.$$

Thus we have constructed a tuple  $(R, R_1, p, \varphi, \varphi_1)$  on  $\mathcal{H}$ . To determine the conjugation  $\mathfrak{J}_p$  on  $\mathcal{H}$ , we can define  $\mathfrak{J}_p$  on  $\mathcal{H}_0$  by:

$$\mathfrak{J}_p p = p, \qquad \mathfrak{J}_p(R^n p) = R^n p \text{ for all } n.$$

The value of  $\mathfrak{J}_p|_{\mathcal{H}_{2k-1}\oplus \mathrm{span}\{p_k\}}$  can be defined by

$$\mathfrak{J}_p p_k = p_k, \qquad \mathfrak{J}_p(\varphi^n p_k) = (\varphi^*)^n p_k, \qquad \mathfrak{J}_p((\varphi^*)^n p_k) = \varphi^n p_k \text{ for all } n.$$

Similarly the value of  $\mathfrak{J}_p|_{\mathcal{H}_{2k}\oplus \operatorname{span}\{p_k^1\}}$  can be defined by

$$\mathfrak{J}_p p_k^1 = p_k^1, \qquad \mathfrak{J}_p(\varphi_1^n p_k^1) = (\varphi_1^*)^n p_k^1, \qquad \mathfrak{J}_p((\varphi_1^*)^n p_k^1) = \varphi_1^n p_k^1 \text{ for all } n$$

So now we have defined a tuple  $(R, R_1, p, \mathfrak{J}_p, \varphi, \varphi_1)$ , such that the contraction  $\mathcal{T} := R_1 \varphi_1 \varphi^* R^{-1}$  is asymptotically stable implied by Proposition 7.6.3. Thus by Proposition 5.2.4, we can find a Hankel operator  $\Gamma$ , such that there exists a conjugation  $\mathfrak{J}_u$ commuting with  $|\Gamma|, |\Gamma_1|$  and preserves  $\Gamma^* e_0$ , and the tuple  $(|\widetilde{\Gamma}|, |\widetilde{\Gamma}_1|, u, \widetilde{\phi}, \widetilde{\phi}_1, \widetilde{\mathfrak{J}}_u)$  given in (5.1.5) is unitarily equivalent to  $(R, R_1, p, \mathfrak{J}_p, \varphi, \varphi_1, \mathfrak{J}_p)$ , and  $\Gamma$  satisfies properties (a),(b) given in Proposition 10.1.5.

#### 10.2.2 Proof of Uniqueness

*Proof.* According to Proposition 5.2.4, we only need to show the uniqueness of tuple  $(R, R_1, p, \varphi, \varphi_1)$  (the unique choice of  $\mathfrak{J}_p$  is given in Remark 7.6.6).

Suppose that there are two tuples  $(R, R_1, p, \varphi, \varphi_1)$  and  $(R', R'_1, p', \varphi', \varphi'_1)$  that both satisfy the condition (a),(b) in Proposition 10.1.5 (Here  $\varphi, \varphi_1, \varphi', \varphi'_1$  are all the canonical choices for their equivalent classes). We will show that they are unitary equivalent.

We also reuse the definition inside the proof of Theorem 8.0.4. That is,

$$\mathcal{H}_{0} := \overline{\operatorname{Span}} \{ R^{n} p | n \ge 0 \};$$
  
$$\mathcal{E}_{\lambda_{k}} := \operatorname{Ker}(R - \lambda_{k}I), \qquad \mathcal{E}_{\mu_{k}} := \operatorname{Ker}(R - \mu_{k}I);$$
  
$$\mathcal{E}_{\lambda_{k}}^{1} := \operatorname{Ker}(R_{1} - \lambda_{k}I), \qquad \mathcal{E}_{\mu_{k}}^{1} := \operatorname{Ker}(R - \mu_{k}I);$$
  
$$\widetilde{R} = R |_{\mathcal{H}_{0}}, \qquad \widetilde{R}_{1} = R_{1} |_{\mathcal{H}_{0}}$$

$$\mathcal{H}'_{0} := \overline{\operatorname{Span}} \{ (R')^{n} p' | n \ge 0 \};$$
  

$$\mathfrak{E}_{\lambda_{k}} := \operatorname{Ker}(R' - \lambda_{k}I), \qquad \mathfrak{E}_{\mu_{k}} := \operatorname{Ker}(R' - \mu_{k}I);$$
  

$$\mathfrak{E}^{1}_{\lambda_{k}} := \operatorname{Ker}(R'_{1} - \lambda_{k}I), \qquad \mathfrak{E}^{1}_{\mu_{k}} := \operatorname{Ker}(R'_{1} - \mu_{k}I);$$
  

$$\widetilde{R}' = R' |_{\mathcal{H}'_{0}}, \qquad \widetilde{R}'_{1} = R'_{1} |_{\mathcal{H}'_{0}}$$

And we also define the projection vector  $p_k, p_k^1$  as

$$p_k := P_{\mathcal{H}_0 \cap \operatorname{Ker}(R - \lambda_k I)} p, \qquad p_k^1 := P_{\mathcal{H}_0 \cap \operatorname{Ker}(R_1 - \mu_k I)} p,$$

 $p'_k, (p^1_k)'$  are defined similarly.

Now by Theorem 8.0.1, there exists a unitary equivalence between the tuple  $(R|_{\mathcal{H}_0}, R_1|_{\mathcal{H}_0}, p)$ and  $(R'|_{\mathcal{H}'_0}, R'_1|_{\mathcal{H}_0}, p')$ :

$$\widetilde{R}' = \widetilde{V}\widetilde{R}\widetilde{V}^*, \qquad \widetilde{R}'_1 = \widetilde{V}\widetilde{R}_1\widetilde{V}^*, \qquad p' = \widetilde{V}p$$

We can easily see that

$$p'_k = \widetilde{V}p_k, \qquad (p_k^1)' = \widetilde{V}(p_k^1),$$

holds for all k.

Now also recall that in the proof of Generalized Borg's Theorem 8.0.4, that we extend  $\tilde{V}$  to a unitary  $V : \mathcal{H} \to \mathcal{H}'$  by defining an arbitrary unitary  $\phi_k$  on  $\mathcal{E}_{\mu_k}$  and an arbitrary unitary  $\phi_k^1$  on  $\mathcal{E}_{\lambda_k}^1$ . and then construct the unitary equivalence between  $(R, R_1, p)$  and  $(R', R'_1, p')$ . Now for the proof of this theorem, we build up the unitary equivalence between  $(\varphi, \varphi_1)$  and  $(\varphi', \varphi'_1)$  by carefully choosing  $\phi_k$  and  $\phi_k^1$ .

We know that  $p_k$  is a \*-cyclic vector for  $\varphi|_{\mathcal{E}_{\lambda_k}}$  on  $\mathcal{E}_{\lambda_k}$ . Denote  $N = \dim \mathcal{E}_{\lambda_k}$ . In Proposition 10.1.2, we have shown that  $\varphi|_{\mathcal{E}_{\lambda_k}}$  have N different eigenvalues on T, and the projection of  $p_k$  on each eigenspace is non-zero. Denoting that the eigenvalues to be  $\beta_1, ..., \beta_N$ , and the projection of  $p_k$  onto the corresponding eigenspace to be  $p_{k1}, ..., p_{kN}$ .

Now we consider  $p'_k$  and the unitary  $\varphi'|_{\mathfrak{E}_{\lambda_k}}$ . Since the scalar spectral measure for two different tuples are the same, we have  $\varphi'|_{\mathfrak{E}_{\lambda_k}}$  also have simple eigenvalues as  $\beta_1, ..., \beta_N$ . In addition, the projection of  $p'_k$  onto the corresponding eigenspaces, defined as  $p'_{k1}, ..., p'_{kN}$ , have the same norm as  $p_{k1}, ..., p_{kN}$  respectively. Thus we define a unitary  $\phi_k : \mathcal{E}_{\lambda_k} \to \mathfrak{E}_{\lambda_k}$  by:

$$\phi_k p_{k1} = p'_{k1}, \qquad \dots, \qquad \phi_k p_{kN} = p'_{kN}. \tag{10.2.11}$$

We can show that  $\phi_k(\varphi^n p_k) = (\varphi')^n p'_k$  for all n. In fact,

$$\phi_k(\varphi^n p_k) = \phi_k \left( \beta_1^n p_{k1} + \dots + \beta_N^n p_{kN} \right)$$
  
=  $\beta_1^n p'_{k1} + \dots \beta_N^n p'_{kN} = (\varphi')^n p'_k$  (10.2.12)

Since in Remark 10.1.4, we have shown that  $\overline{\text{Span}}\{\varphi^n p_k | 0 \leq n \leq N-1\} = \mathcal{E}_{\lambda_k}$ . We can also use (10.2.12) for the definition of  $\phi_k$ . Easy to see that for this  $\phi_k$ , we have  $\phi_k p_k = p'_k$  and  $\phi_k$  maps  $\mathcal{E}_{\lambda_k}$  to  $\mathfrak{E}_{\lambda_k}$ .

Similarly we define  $\phi_k^1$  be a unitary operator mapping  $\mathcal{E}_{\mu_k}^1$  to  $\mathfrak{E}_{\mu_k}^1$ , such that

$$\phi_k^1(\varphi_1^n p_k^1) = (\varphi_1')^n (p_k^1)'$$

holds for all  $n \in \mathbb{N}$ .

Now we can define a unitary  $V : \mathcal{H} \to \mathcal{H}'$  by:

$$V = V \text{ on } \mathcal{H}_0,$$
$$V = \phi_k \text{ on } \mathcal{E}^1_{\lambda_k}, \qquad V = \phi^1_k \text{ on } \mathcal{E}_{\mu_k}.$$

We show that  $\varphi' = V \varphi V^*$  (the other equality  $\varphi'_1 = V \varphi_1 V^*$  can be shown similarly). For  $\forall x \in \mathcal{E}_{\mu_k}$ , we have

$$V\varphi x = Vx = \phi_k^1 x, \qquad \varphi' Vx = \varphi' \phi_k^1 x = \phi_k^1 x.$$

Here  $\varphi x = x$  and  $\varphi'(\phi_k^1 x) = \phi_k^1 x$  follows from  $\varphi|_{\mathcal{E}_{\mu_k}} = I$  and  $\varphi'|_{\mathfrak{E}_{\mu_k}} = I$ . Thus  $V\varphi = \varphi' V$  on  $\mathcal{E}_{\mu_k}$ .

Now we show  $V\varphi = \varphi' V$  is also true on  $\mathcal{E}_{\lambda_k}$ . For vector  $(\varphi^n p_k)$ , using (10.2.12), we

have

$$\varphi' V(\varphi^n p_k) = \varphi'(\varphi')^n p_k^1 = (\varphi')^{n+1} p_k^1$$
$$V\varphi(\varphi^n p_k) = V(\varphi^{n+1} p_k) = (\varphi')^{n+1} p_k^1$$

Since  $\mathcal{E}_{\lambda_k} = \overline{\text{Span}} \{ \varphi^n p_k | n \ge 0 \}$  given by Remark 10.1.4, we have  $V \varphi = \varphi' V$  is true on  $\mathcal{E}_{\lambda_k}$ , hence holds on the whole space  $\mathcal{H}$ .

Similarly, we can show that  $\varphi'_1 = V \varphi_1 V^*$  also holds on  $\mathcal{H}$ . Thus we build up the unitary equivalence between  $(R, R_1, p, \varphi, \varphi_1)$  and  $(R', R'_1, p', \varphi', \varphi'_1)$ . And the proof of uniqueness is done.

# 10.3 Result for Finite Rank Hankel Operators with Multiple Singular Values

In this section, we will discuss the spectral data that can uniquely determine a finite rank Hankel operators with multiple singular values.

Recall that Proposition 6.2.11, given finite rank operators  $R, R_1$ , we have shown that on a subspace  $\mathcal{H}_0 := \overline{\text{Span}} \{ R^n p | n \ge 0 \}$ , there exists an intertwining sequence

$$\lambda_1 > \mu_1 > \dots > \lambda_N > \mu_N \ge 0,$$

such that  $R|_{\mathcal{H}_0}, R_1|_{\mathcal{H}_0}$  has simple eigenvalues as  $\{\lambda_k\}_{k=1}^N, \{\mu_k\}_{k=1}^N$  respectively.

**Condition 1:**  $\mu_N > 0$ , then from a similar proof to Proposition 7.6.3, we can show that for  $1 \le k \le N$ ,  $p_k$  is a \*-cyclic vector for  $E_{\lambda_k}$ , and  $p_k^1$  is a \*-cyclic vector for  $E_{\mu_k}^1$ .

Hence for  $1 \le k \le N$ , we can define the scalar spectral measure  $\rho_k, \rho_k^1$  on  $\mathbb{T}$  by

$$\langle (\varphi - zI)^{-1}p_k, p_k \rangle = \int_{\mathbb{T}} \frac{d\rho_k(s)}{s-z}, \qquad \langle (\varphi_1 - zI)^{-1}p_k^1, p_k^1 \rangle = \int_{\mathbb{T}} \frac{d\rho_k^1(s)}{s-z}.$$

By normalizing  $\{\rho_k\}_{k=1}^N, \{\rho_k^1\}_{k=1}^N$  as probability measure  $\{\widetilde{\rho}_k\}_{k=1}^N, \{\widetilde{\rho}_k^1\}_{k=1}^N$ , we can also

prove that  $\{\widetilde{\rho}_k\}_{k=1}^N, \{\widetilde{\rho}_k^1\}_{k=1}^N$  are discrete measure, and concentrated on finitely many points on  $\mathbb{T}$ . We can show that the four sequences  $\{\lambda_k\}_{k=1}^N, \{\mu_k\}_{k=1}^N, \{\widetilde{\rho}_k\}_{k=1}^N, \{\widetilde{\rho}_k^1\}_{k=1}^N$ will uniquely determine a finite rank Hankel operator.

**Condition 2:**  $\mu_N = 0$ , then we have dim  $E_{\mu_N}^1 = 1$  and  $E_{\mu_N}$  is trivial. Under this case, we have  $\varphi_1|_{E_{\mu_N}^1} = 0$ , and

- (i) For  $1 \le k \le N$ ,  $p_k$  is a \*-cyclic vector for  $E_{\lambda_k}$ ;
- (ii) For  $1 \le k \le N 1$ ,  $p_k^1$  is a \*-cyclic vector for  $E_{\mu_k}^1$ .

We can show that the four sequences  $\{\lambda_k\}_{k=1}^N$ ,  $\{\mu_k\}_{k=1}^N$ ,  $\{\widetilde{\rho}_k\}_{k=1}^N$ ,  $\{\widetilde{\rho}_k^1\}_{k=1}^{N-1}$  will uniquely determine a finite rank Hankel operator.

The proof to both conditions are typically similar to the proof for theorem 10.1.5, so we omit it.

# BIBLIOGRAPHY

- N.K.Nikolski, Operators, Functions, and Systems: An easy reading Volume I: Hardy, Hankel, and Toeplitz, translated by Andreas Hartmann, 2000 Mathematics Subject Classification, Primary 47-02, 30-02, 93-02, 30D55, 47B35, 47A45, 93B05, 93C05
- [2] Z. Nehari, On bounded bilinear forms, Acta Math. 65,153-162 (1957)
- [3] W. Magnus, On the spectrum of Hilbert matrix, Amer. J. Math. 72(1950), 699-704
- [4] P. Hartman, On completely continuous Hankel matrices, Proc. Amer. Math. Spc.9 (1958), 862-866
- [5] L. Kronecker, Zur Theorie der Elimination einer Variablen aus zwei algebraischen Gleichungen, Monatsber. Konigl. Preussischen Akad. Woss. (Berlin), 1881, 535-600
- [6] S. V. Khrushchev, S. V. Peller, Moduli of Hankel operators, past and future, in linear and complex analysis problem book, lecture notes in Math. 1043, pp.92-97, Springer-Verlag, Berlin-New York, 1984
- [7] N. K. Nikolskii, Treatise on the shift operator, spectral function theory, Springer-Verlag, Berlin-Heidelberg, 1986
- [8] S. R.Treil, moduli of Hankel operators and a problem of Peller-Khrushchev, Dokl.Akad. Nauk SSSR, 283 (1995), 1095-1099; English translation in Soviet Math. Dokl,. 32(1985), 293-297

- S. R. Treil, V. I. Vasyunin, The inverse spectral problem for the modulus of a Hankel operator, Algebra i Analiz, 1:4 (1989), 54-67; English translation in Leningrad Math. J., 1 (1990), 859-870
- [10] S. R. Treil, An inverse spectral problem for the modulus of the Hankel operators, and balanced realizations, Algebra i Analiz, 1990, Volume 2, Issue 2, 158-182
- [11] M. S. Birman, M. Z. Solomjak, Spectral Theory of Self-Adjoint Operators in Hilbert space, translated from Russian by S. Khrushchev and V. Peller, Springer-Verlag, Dordrecht Netherlands, 1987
- [12] A. V. Megretskiĭ, V. V. Peller, and S. R. Treil, The inverse spectral problem for self-adjoint Hankel operators, Acta Math. 174 (1995), no. 2, 241–309.
- [13] S. C. Power, Quasinilpotent Hankel operators, in Linear and Complex Analysis Probleme Book, Lect. Notes Math., vol. 1043, Springer-Verlag, 1984, pp. 259-261
- [14] A. V. Megretskii, A quasinilpotent Hankel operator, Leningrad Math. J. 2(1991), 879-889
- [15] R. A. Martinez-Avendano, S. R. Treil, An inerse spectral problem for Hankel operators, J. Operator Theory 48 (2002), no. 1, 83-93. 47B35 (47A10 47B15)
- [16] E. V. Abakumov, The inverse spectral problem for finite-rank Hankel operators, Journal of Mathematical Sciences, Vol. 85, No. 2, 1997
- [17] P. Gerard, S. Grellier, Invariant Tori for the cubic Szegö equation, arXiv: 1011.5479
- [18] P. Gerard, S. Grellier, The cubic Szegö equation, Ann. Scient. ć. Norm. Sup. 43 (2010), 761-810
- [19] P. Gerard, S. Grellier, Multiple singular values of Hankel operators, arXiv: 1402.1716

- [20] R. G. Douglas, On majorization, factorization and range inclusion of operators on Hilbert space, Proc. Amer. Math. Soc., 17 (1966), 413-415
- [21] S. Garcia, M. Putinar, Complex Symmetric Operators And Applications II, transactions of the American mathematical society, Volume 359, Number 8, August 2007, Pages 3913-3931, S 0002-9947(07)64213-4
- [22] S. Garcia, R. Stephan, E. Prodan, M. Putinar, Mathematical and physical aspects of complex symmetric operators, J. Phys. A 47(2014), no. 35, 353001, 54 pp.47A05 (47-02 81Q10 81Q12)
- [23] S. Garcia, Conjugation and Clark operators, Recent advances in operator-related function theory, 67-111, Contemp. Math., 393, Amer. Math. Soc., Providence, RI, 2006. 47B35 (30D55 46E22 47A15 47B32 47B38)
- [24] M. C. Câmara, K. K. Garlicka, B. Lanucha, M. Ptak, Conjugations in L<sup>2</sup> and their invariants, Analysis and Mathematical Physics (2020)
- [25] H. Radjavi, P. Rosenthal, *Invariant Subspaces*, Springer, New York (1973)
- [26] B. S. Nagy, C. Foias, H. Bercivici, and L. Kérchy, Harmonic Analysis of Operators on Hilbert Space Second Edition, Springer-Verlag, New York-Dordrecht-Heidelberg-London, 2010
- [27] S. R. Foguel, Powers of a contraction in Hilbert space, Pacific J.Math.13 (1963), 551-562
- [28] M. C. Pereyra, S. Marcantognini, A. M. Stokolos, W. Urbina, Harmonic Analysis, Partial Differential Equations, Banach Spaces, and Operator Theory (Volume 2), celebrating Cora Sadosky's Life, Association for Women in Mathematics Series, 5. Springer, Cham, 2017. xx+460 pp. ISBN:978-3-319-51593-9; 978-3-319-51591-5
- [29] J. B. Conway, A Course In Functional Analysis second edition, Springer Science Business Media New York 2007

- [30] J. Bram, Subnormal operators, Duke Math. **22**(1955), 75-94
- [31] P. R. Halmos, A Hilbert Space Problem Book, Van Nostrand, New York, 1967, 2nd edition, Springer, New York, 1982
- [32] G. Borg, Eine Umkehrung der Sturm-Liouvilleschen Eigenwertaufgabe : Bestimmung der Differentialgleichung durch die Eigenwerte, Acta Math. 78(1946) 1-96
- [33] N. Levinson, The inverse Sturm-Liouville problem, Matematisk Tidsskrift. B(1949), pp.25-30
- [34] B. Simon, Operator Theory, A Comprehensive course in Analysis, Part 4, AMS Providence, United States, 2015
- [35] A. Harnack, Die Grundlagen der Theorie Des logarithmischen Potentiales und der eindeutigen Potentialfunktion in der Ebene, Leipzig, 1887
- [36] P. Gerard, S. Grellier, Inverse spectral problems for compact Hankel operators, J. Inst. Math. Jussieu 13(2014), no.2 2, 273-301.47B35 (37K15 47A75)
- [37] http://www1.maths.leeds.ac.uk/~kisilv/courses/math3263m011.html
- [38] S. R. Treil, Z. Liang, A dynamical system approach to the inverse spectral problem for Hankel operators: the case of compact operators with simple singular values, preprint
- [39] S. R. Treil, Z. Liang, An inverse spectral problem for Hankel operators: the case of compact operators with non-simple singular values, preprint